First print



New South Wales

Planning Bill 2013

Explanatory note

This explanatory note relates to this Bill as introduced into Parliament. The *Planning Administration Bill 2013* is cognate with this Bill.

Overview of Bill

The object of this Bill is to introduce a new planning system for New South Wales as outlined in the Government's White Paper released in April 2013. Following public consultation a number of changes have been made to the reforms outlined in the White Paper, in particular to increase community participation and local community powers in relation to planning matters and to provide greater scrutiny of planning decisions.

Outline of provisions

Part 1 General

Division 1.1 contains preliminary provisions, including the name of the proposed Act, the commencement of the proposed Act on a date to be proclaimed and the following objects of the proposed Act:

- (a) to promote the growth of the State's economy and increased productivity,
- (b) to promote sustainable development,
- (c) to provide opportunities for early and on-going community participation in strategic planning and to promote transparent decision-making,
- (d) to facilitate and manage growth by the co-ordination, planning, delivery and integration of infrastructure and services in strategic planning,

- (e) to promote the timely delivery of business, employment and housing opportunities (including for housing choice and affordable housing),
- (f) to promote the protection of the environment and heritage, including by:
 - (i) the conservation of biodiversity, and
 - (ii) the conservation and sustainable management of built and cultural heritage (including Aboriginal cultural heritage),
- (g) to enable the effective management of natural hazards and natural resources, including agricultural land, water and minerals,
- (h) to promote health and safety in the design, construction and performance of buildings,
- (i) to promote health, amenity and quality in the design and planning of the built environment,
- (j) to promote efficient and timely development assessment that is proportionate to the likely impacts of proposed development,
- (k) to share responsibility between all levels of government for planning and the management of growth.

Sustainable development is defined as development that meets the needs of the present generation without compromising the ability of future generations to meet their own needs. Sustainable development is achieved by the integration of relevant economic, environmental and social considerations in decision-making about planning and development.

Division 1.2 deals with the interpretation of words and expressions used in the proposed Act. The Division contains the definitions of key words and expressions—Schedule 1 contains a dictionary of all relevant words and expressions. The key words and expressions include the following:

- (a) development,
- (b) strategic plans (namely, NSW planning policies, regional growth plans, subregional delivery plans and local plans),
- (c) planning approvals (namely, development consents, State infrastructure approvals and construction and other certificates under Part 8),
- (d) infrastructure plans and contributions (including local infrastructure plans and growth infrastructure plans),
- (e) development likely to significantly affect threatened species and biodiversity offsets.

Division 1.3 contains general provisions that list the categories of development and that provide for criminal and civil enforcement for carrying out prohibited development or development without or contrary to planning approval or building and related work activity without a certificate under Part 8.

Part 2 Community participation

The Part sets out the Community Participation Charter which contains principles to engage the community (including industry, businesses, residents, interest groups and organisations) in strategic planning, planning decisions and other planning matters. The Part requires planning authorities (including the Minister and councils) to exercise their planning functions consistently with the Charter. For that purpose, planning authorities are to have community participation plans about how they will undertake community participation.

Certain community participation requirements will be mandatory (see Part 1 of Schedule 2 to the proposed Act).

The NSW Planning Director-General is also required to provide online delivery of planning information and services (including on the NSW planning portal).

Part 3 Strategic planning

Division 3.1 contains introductory provisions, including identifying the relevant planning authority for preparing strategic plans, being the NSW Planning Director-General for NSW

planning policies or regional growth plans, the relevant subregional planning board for subregional delivery plans and the council (or other designated authority) for local plans.

Division 3.2 deals with the preparation and making of strategic plans (other than local plans). The Division sets out what each such plan is to contain and establishes a hierarchy of plans that give effect to each other (namely NSW planning policies, then regional growth plans, then subregional delivery plans and finally local plans).

Division 3.3 makes general provisions in relation to local plans. The Division establishes a local plan for each council area (and for any other areas not within a council area). The local plan is to comprise provisions as follows:

- (a) strategic context provisions,
- (b) planning control provisions,
- (c) development code or guide provisions,
- (d) contribution provisions.

The Division declares that the Minister may make the provisions of local plans, but authorises the council to make planning control provisions authorised by a gateway determination, local infrastructure contributions provisions and other provisions.

Division 3.4 deals with the planning control provisions of local plans. The provisions are those that are currently contained in environmental planning instruments, including land use zoning and the categories of development. The Division continues provisions relating to the standard instrument program for the planning control provisions of a local plan. The Division also continues as part of the process for making planning control provisions:

- (a) the requirement for a planning proposal to be prepared, and
- (b) a gateway process where the Minister determines whether the proposal is to proceed and a number of related matters (such as the period of public exhibition and whether the making of the plan is devolved to the council), and
- (c) the ability to suspend covenants and agreements.

Division 3.5 contains miscellaneous provisions relating to strategic plans, including provisions relating to the public exhibition of proposed plans and the circumstances in which plans can be made without compliance with the conditions precedent for their making.

Part 4 Development (other than infrastructure) assessment and consent

Division 4.1 contains introductory provisions including:

- (a) declaring that the Part applies to development that requires development consent, and
- (b) identifying the provisions that apply to development consent in the form of a complying development certificate, and
- (c) providing that development applications may be assessed as complying development, code assessed or merit assessed.

Division 4.2 declares who is the consent authority for development, namely:

- (a) a certifier—for complying development the subject of an application for a complying development certificate, and
- (b) the Minister—for State significant development, and
- (c) a regional planning panel—for regionally significant development, and
- (d) the council or other designated public authority—for other development.

Division 4.3 deals with complying development. A building or subdivision certifier is required to issue a certificate if the development complies with the relevant standards in the development code or guide provisions of the local plan.

Division 4.4 deals with code and merit assessment. Code assessment is a new assessment process under which consent may be granted for development that is identified in the planning control provisions of the local plan and that meets the development standards in the relevant development assessment code for precinct development (and which adopts acceptable or alternative solutions to meet the relevant performance criteria in the local plan). Code assessment will not apply to certain development (such as EIS assessed development, development that requires heritage approval or that is subject to concurrence or consultation because of the effect on threatened species).

If development is not code assessed it is to be merit assessed according to similar considerations that currently apply.

The Division continues provisions relating to the referral of proposals to the Planning Assessment Commission (*PAC*), staged development applications and other matters.

Division 4.5 relates to the imposition of conditions of development consent in line with existing provisions.

Division 4.6 makes additional provisions for State significant development. The Minister is authorised to "call in" development as State significant development.

Division 4.7 introduces a new procedure to enable the carrying out of development otherwise prohibited by the planning control provisions of a local plan. Proposed development may be permitted if the regional planning panel (or, in limited circumstances, the NSW Planning Director-General) issues a strategic compatibility certificate on the grounds that the planning control provisions have not yet been amended to give effect to a new regional growth plan or subregional delivery plan and that the proposed development will not have any significant adverse impact on the likely future uses of surrounding land.

Division 4.8 deals with the modification of development consents.

Division 4.9 contains miscellaneous provisions (including with respect to Crown development applications and the lapsing of development consent).

Part 5 Infrastructure and environmental impact assessment

Division 5.1 continues arrangements under Part 5 of the existing Act for environmental impact assessment of development that is not subject to Part 4 assessment and consent. The assessment generally applies when a public authority proponent carries out development or a determining authority grants approval for the development under other legislation. The Division continues the general duty to consider the environmental impact of relevant development and to obtain and consider an environmental impact statement if the development is likely to have a significant effect on the environment (including if the development is likely to significantly affect threatened species).

Division 5.2 provides for the declaration of State infrastructure development that requires the approval of the Minister but does not require development consent under Part 4. The Division continues existing provisions relating to the environmental assessment and approval of such development.

Division 5.3 deals with a new category of development to be called public priority infrastructure. Once particular development is declared by the Minister as public priority infrastructure, it can be carried out without further planning approval. In order to be declared, the development needs to be identified as priority infrastructure in a strategic or infrastructure plan (other than a local plan) or, in the opinion of the Minister, be essential for the State. Before public priority infrastructure is carried out, the proponent is required to prepare and publicly consult on a project definition report. The other provisions of the proposed Act do not generally apply to public priority infrastructure.

Part 6 Concurrences, consultation and other legislative approvals

Division 6.1 sets out the approvals under other legislation that do not apply to public priority infrastructure, State infrastructure development or State significant development (including

approvals under fisheries management legislation, heritage approvals, native vegetation clearing approvals and water management approvals).

The Division also sets out the approvals under other legislation that must be issued consistently for public priority infrastructure, State infrastructure development or State significant development (including environment protection licences and mining leases).

Division 6.2 sets out requirements in relation to development (with certain exceptions) for consultation or concurrence with relevant Ministers or agencies if the development is likely to significantly affect threatened species, if required under planning control provisions of local plans or if carried out on certain bush fire prone land.

The Division continues and expands existing provisions that enable a local plan to be amended to remove consultation or concurrence requirements or other statutory provisions so as to facilitate the carrying out of development.

Division 6.3 introduces a new one stop referrals and decisions process for other legislative approvals (and for concurrences and consultation) to replace existing provisions that apply to integrated development. Under the new procedure, a single set of general terms of approval for development are to be issued by the NSW Planning Director-General in accordance with State assessment requirements rather than being issued by the approval bodies. If a development consent is granted for development consistently with the general terms of approval, the approval body is to grant any related legislative approval in a form that is substantially consistent with the general terms of approval.

Part 7 Infrastructure and other contributions

Division 7.1 contains introductory provisions, including:

- (a) a definition of *local infrastructure* to mean local roads, local drainage works, open space and community facilities, and
- (b) a definition of *regional infrastructure* to mean regional or State roads, land for drainage, transport infrastructure, regional open space and educational establishments, and
- (c) the application of the Part to development that requires consent under Part 4, or any State infrastructure development that is not carried out by or on behalf of a public authority.

Division 7.2 empowers a consent authority, by condition of development consent, to impose a local infrastructure contribution in accordance with the contribution provisions of the local plan. A local infrastructure contribution may be a direct contribution (a reasonable contribution having a nexus with the proposed development) or an indirect contribution (a contribution based on a percentage of capital investment value or based on the area of the proposed development). The Division contains provisions for local infrastructure plans which identify the local infrastructure to be funded by local infrastructure contributions.

Division 7.3 empowers a consent authority, by condition of development consent, to impose a regional infrastructure contribution in accordance with the contribution provisions of the local plan. A regional infrastructure contribution is to be an indirect contribution (a contribution based on a percentage of capital investment value or calculated by reference to the area of the proposed development or other authorised method). The Division contains provisions for growth infrastructure plans which identify the regional infrastructure to be funded by regional infrastructure contributions.

Division 7.4 authorises a consent authority, by condition of development consent, to impose a biodiversity offset contribution for the conservation and enhancement of the natural environment of the State in accordance with the contribution provisions of the local plan. Biodiversity offset contributions are to be paid into specially created funds in the threatened species legislation.

Division 7.5 relates to planning agreements between public authorities and the developers of land for the payment of money, the provision of land or the carrying out of works by the developer for infrastructure, affordable housing or the conservation or enhancement of the natural environment. Planning agreements can exclude the payment of local, regional or biodiversity offset contributions. A planning agreement can be made in contemplation of a development application

or proposed change to a local plan, but cannot require a public authority to grant consent or exercise planning functions to change a local plan.

Division 7.6 establishes a Regional Contributions Fund for regional infrastructure contributions to be administered by the Secretary of the Treasury.

Part 8 Building and subdivision

Division 8.1 contains introductory provisions, including the list of certificates that may be issued under the Part, namely:

- (a) construction certificates, and
- (b) subdivision works certificates, and
- (c) occupation certificates, and
- (d) subdivision certificates, and
- (e) compliance certificates.

Division 8.2 sets out the requirements in order to commence building work, the requirement to obtain a construction certificate from a certifier (unless a complying development certificate has been issued as development consent) and the requirement to obtain an occupation certificate from a certifier once the work is completed.

Division 8.3 sets out the requirements in order to commence subdivision work, the requirement to obtain a subdivision works certificate from a certifier and the requirement to obtain a subdivision certificate to register the subdivision once the work is completed and other obligations have been complied with.

Division 8.4 deals with compliance certificates that may be obtained from a certifier (including a completion of work compliance certificate that can be an authorised alternative to an occupation certificate in certain cases).

Division 8.5 carries forward existing provisions that limit liability for defective building or subdivision work.

Division 8.6 contains miscellaneous provisions, including the new requirement for an owner's building manual to be provided with the issue of an occupation certificate for a building.

Part 9 Reviews and appeals

Division 9.1 contains introductory provisions, including a definition of *Court* to mean the Land and Environment Court.

Division 9.2 enables an applicant for development consent to request, in certain cases, a review of the determination of the application. Generally, reviews are to be conducted by a person who is not subordinate to the person who determined the application.

Division 9.3 continues rights of appeal to the Court under the existing Act in relation to development consent, including rights of appeal by the applicant for development consent or by the objector in the case of EIS assessed development. There is no right of appeal if the decision is made after a public hearing by the PAC or if the decision related to a complying development certificate.

Division 9.4 provides rights of appeal to the Court in relation to a failure or refusal to issue a certificate under Part 8.

Division 9.5 provides rights of appeal to the Court in relation to the issue of development control orders under Part 10.

Division 9.6 makes miscellaneous provisions relating to appeals.

Part 10 Enforcement

Division 10.1 confers Ministerial enforcement powers, including directions to planning authorities for the timely exercise of their planning functions.

Planning Bill 2013 [NSW] Explanatory note

Division 10.2 makes provision for development control orders to enforce obligations under the proposed Act, comprising general orders, fire safety orders and brothel closure orders. The Division sets out the persons or bodies who may give orders, including councils and the Minister or NSW Planning Director-General in relation to State significant and other specified development.

Detailed provisions relating to orders are contained in Schedule 10.

Division 10.3 continues the civil enforcement jurisdiction of the Land and Environment Court in relation to remedying or restraining breaches of the proposed Act.

The Division also continues provisions:

- (a) that limit the period for court challenges to the validity of plans and planning approvals to the period of 3 months after the plan or approval is made or given, and
- (b) that limit the requirements of the proposed Act that are mandatory for the purposes of determining the validity of plans and approvals to the mandatory community participation requirements, and
- (c) that make special provision where development consents are tainted by corruption.

Division 10.4 deals with criminal offences and proceedings. The Division sets out the maximum penalties for offences against the proposed Act, with different penalties for Tier 1, Tier 2 and Tier 3 offences.

The Division also makes other provisions relating to the time within which proceedings may be commenced, penalty notices and ancillary offences.

Part 11 Miscellaneous

Division 11.1 empowers the making of regulations for the purposes of the proposed Act, including regulations that insert or amend provisions in the ancillary Schedules of the proposed Act and savings and transitional regulations.

Division 11.2 provides for fees and charges under the planning legislation.

Division 11.3 provides for the issue of planning and building information certificates to prospective purchasers of property and other persons.

Division 11.4 continues provisions of the existing Act relating to the protection of existing uses.

Division 11.5 continues provisions of the existing Act relating to contaminated land liability.

Division 11.6 makes provisions for bush fire prone land, including the preparation of maps of any such land.

Division 11.7 contains general miscellaneous provisions, including the continuation of existing provisions relating to the disclosure of political donations in connection with planning matters.

Schedule 1 General—ancillary provisions

The Schedule contains the dictionary of words and expressions used in the proposed Act. Key terms are defined in Part 1 of the proposed Act.

The Schedule also sets out the test for determining whether proposed development is likely to significantly affect threatened species (which continues the existing test but provides that proposed development is not likely to do so if biodiversity assessment procedures adopted by the regulations so determine).

Schedule 2 Community participation—ancillary provisions

Part 1 sets out the mandatory community participation requirements for public exhibition or notification. Part 2 contains general provisions relating to community participation. Part 3 deals with the delivery of e-planning services and information, including the NSW planning portal.

Schedule 3 Strategic planning—ancillary provisions

The Schedule contains regulation-making powers.

Schedule 4 Development (other than infrastructure) assessment and consent—ancillary provisions

The Schedule deals with the lapsing of consent, biobanking, orders for conditional validity of consents and other miscellaneous matters.

Schedule 5 Infrastructure and environmental impact assessment—ancillary provisions

The Schedule continues existing provisions relating to environmental impact assessments and State infrastructure development, and regulation-making powers for that development or public priority infrastructure.

Schedule 6 Concurrences, consultation and other legislative approvals—ancillary provisions

The Schedule contains a provision relating to heritage and bush fire approvals. In particular, the Schedule requires the NSW Planning Director-General to act (except in limited circumstances) in accordance with advice of the Heritage Council and the Rural Fires Services Commissioner when determining one stop referrals.

Schedule 7 Infrastructure and other contributions—ancillary provisions

The Schedule contains regulation-making powers (including for a system of statutory charges to secure payment of contributions under Part 7). The Schedule also continues the funds under the existing Act administered by the Ministerial Corporation (re-named the Planning Growth Funds and Trust Funds).

Schedule 8 Building and subdivision—ancillary provisions

The Schedule contains regulation-making powers.

Schedule 9 Reviews and appeals—ancillary provisions

The Schedule contains regulation-making powers.

Schedule 10 Enforcement—ancillary provisions

Part 1 of the Schedule continues Ministerial powers to appoint regional planning panels or administrators to exercise the planning functions of councils in the place of the council in certain specified circumstances.

Part 2 of the Schedule contains the detailed provisions relating to the issue of development control orders.

Part 3 of the Schedule contains ancillary evidentiary provisions for civil enforcement.

Schedule 11 Miscellaneous—ancillary provisions

The Schedule contains miscellaneous regulation-making powers and continues special provisions relating to paper subdivisions.

Schedule 12 Savings, transitional and other provisions

The Schedule contains savings and transitional provisions consequent on the repeal of the existing Act and the enactment of the proposed Act.

First print



New South Wales

Planning Bill 2013

Contents

				Page
Part 1	Gen	eral		
	Divis	sion 1.1	Preliminary	
	1.1	Name of	Act	2
	1.2	Commer	ncement of Act	2
	1.3	Objects	of Act	2
	Divis	sion 1.2	Interpretation	
	1.4	Definitio	ns—general	2
	1.5	Develop	ment	3
	1.6	Strategic	plans and planning control provisions	3
	1.7	Planning	approvals	3
	1.8	Infrastru	cture plans and contributions	4
	1.9	Building	work and subdivision work	4
	1.10	Planning	bodies	4
	1.11	Develop	ment likely to significantly affect threatened species	4
	1.12	Biodivers	sity offsets	4
	Divis	sion 1.3	Development—general provisions	
	1.13	Categori	es of development	5
		•	development	5
	1.15		ment that requires development consent under Part 4	6

				Page
	1.16		ment that requires State infrastructure approval under Part 5	6
	1.17		g out development without or contrary to planning approval nan Part 8 certificate)	6
	1.18	•	work etc without or contrary to Part 8 certificate	6
	1.19		g out development that is prohibited	6
Part 2	Con	nmunity	v participation	
	2.1	The Cor	nmunity Participation Charter	8
	2.2		g authorities and functions to which Charter applies	8
	2.3	Applying	the Charter in the exercise of planning functions	9
	2.4	Commu	nity participation plans	9
	2.5	Mandato	ory community participation requirements	10
	2.6	Public n	otice of community participation in proposed strategic plans	10
	2.7	Online p	planning services and information	10
	2.8	Commu	nity Participation Advisory Panel	10
Part 3	Stra	ategic pl	anning	
	Divis	sion 3.1	Introductory	
	3.1	Definitio	n of "relevant planning authority"	11
	3.2	Preparin	ng or making strategic plans	11
	Divis	sion 3.2	NSW planning policies, regional growth plans and subregional delivery plans	
	3.3	Prepara	tion and content of NSW planning policies	11
	3.4	Prepara	tion and content of regional growth plans	11
	3.5	Prepara	tion and content of subregional delivery plans	12
	3.6	-	of NSW planning policies, regional growth plans and	
			onal delivery plans	12
	3.7	Impleme subregio	entation of NSW planning policies, regional growth plans and onal delivery plans	13
	Divis	sion 3.3	Local plans—general	
	3.8	Local pla	ans for local government and other areas	13
	3.9		sition of local plans	13
	3.10		t planning authority for provisions of local plan	14
	3.11	Making	provisions of local plans	14
		sion 3.4	Local plan—planning control provisions	. –
	3.12		ion of Division	15
	3.13		es for which planning control provisions may be made	15
	3.14 3.15		nitiated acquisition of land zoned for public purposes disation of planning control or other provisions of local plans	15 16
	3.15		it planning authority to prepare planning proposal for proposed	
		planning	control provisions	17
	3.17		y determination	17
	3.18		It planning authority may vary proposals or not proceed	18
	3.19		anning proposal	18
	3.20 3.21		of planning control provisions	19
	3.21	consent	ent local plan amendments and applications for development	19
		CONSCIL		15

			Page
	3.22	Suspension of covenants, agreements and instruments by planning control provisions) 19
	3.23	Miscellaneous provisions relating to planning control provisions	20
	Divis	sion 3.5 Strategic plans—miscellaneous provisions	
	3.24 3.25 3.26 3.27 3.28	General considerations for preparation of proposed strategic plans Public exhibition of proposed strategic plans Publication of strategic plans Dispensing with conditions precedent to making strategic plans Periodic review and staged repeal of strategic plans	20 20 21 21 22
Part 4		velopment (other than infrastructure) assessment and nsent	
	Divis	ision 4.1 Introductory	
	4.1 4.2 4.3	Application: Part 4 Interpretation: Part 4 Development assessment tracks: Part 4	23 23 23
	Divis	sion 4.2 Consent authorities	
	4.4 4.5	Consent authorities Delegation of consent authority functions	23 24
	Divis	sion 4.3 Complying development	
	4.6 4.7 4.8 4.9	Application for complying development certificate Variations from complying development standards and requirement Determination of applications for complying development certificate Building Code of Australia requirements to apply to complying development	
	4.10 4.11 4.12 4.13	Form of complying development certificate Conditions of a complying development certificate Duration of complying development certificate Modification of complying development	25 25 25 25 26
	Divis	sion 4.4 Code and merit assessment	
	4.14 4.15 4.16 4.17 4.18 4.19	Application for development consent Determination of applications for development consent Development assessment codes Code assessment Merit assessment EIS assessed development that is not State significant development—additional requirements	26 26 27 27 28 29
	4.20 4.21 4.22	Determination must await any PAC public hearing or advice Staged development applications and consents Date from which development consent has effect	29 29 30
		ision 4.5 Conditions of development consent	
	4.23 4.24 4.25	Imposition of conditions—general Conditions that identify outcomes "Deferred commencement" conditions	30 31 31

	4.26	Reviewable conditions of development consent	31
	4.27	Conditions and arrangements relating to securities for development requirements or damage	32
	Divis	ion 4.6 State significant development—additional provisions	
	4.28	"Call in" of development as State significant development	32
	4.29	Environmental impact assessment for, and consent to, State significant development etc	33
	4.30	Staged development applications for State significant development	33
	4.31	PAC to determine certain applications	34
	Divis	ion 4.7 Strategic compatibility certificates—development not permissible under local plan	
	4.32	Prohibited development permissible with consent if strategic	
		compatibility certificate issued	34
	4.33	Who may issue certificate	35
	4.34	Grounds on which certificate may be issued	35
	4.35	Requirements before certificate is issued	35
	4.36	Conditions and duration of certificate	36
	4.37	Development consents to be granted consistently with certificate	36
		ion 4.8 Modification of development consents	
	4.38	Who may make application for modification of development consent	36
	4.39	What is a modification of development consent	36
	4.40	Who may modify development consent	36
	4.41 4.42	Modifications that may be made to development consents Provisions relating to Land and Environment Court	36 37
	Divis	ion 4.9 Miscellaneous	
	4.43	Long service leave levy	37
	4.44	Determination of Crown development applications	37
	4.45	When species impact statement required	38
	4.46	Lapsing of development consent	38
Part 5	Infra	astructure and environmental impact assessment	
	Divis	ion 5.1 Environmental impact assessment (except for State or public priority infrastructure)	
	5.1	Definitions: Division 5.1	39
	5.2	Relevant development to which this Division applies	39
	5.3	General duty to consider environmental impact of relevant development	39
	5.4	Determining authority to obtain and consider EIS for relevant	
	-	development likely to significantly affect the environment	40
	5.5	Significant effect on environment requiring EIS includes significant	
		effect on threatened species	40
	5.6	Decision of determining authority where EIS obtained for relevant	
		development	41
	5.7	Public hearing by Planning Assessment Commission	41
	5.8	Examination of EIS by NSW Planning Director-General	42
	5.9	Exemptions	42

Page

			Page
	Divis	sion 5.2 State infrastructure development	
	5.10	Declaration of State infrastructure development	42
	5.11	State infrastructure development not subject to Part 4 or to	72
	5.11	prohibitions or restrictions in planning control provisions	43
1	5.12	Application for approval of State infrastructure development	43
	5.13	Environmental assessment requirements for approval	43
	5.14	Environmental impact assessment and public consultation	43
	5.15	NSW Planning Director-General's environmental impact assessment	
		report	44
Į	5.16	Giving of approval by Minister to carry out State infrastructure	
		development	44
!	5.17	Staged State infrastructure development applications	45
	5.18	Status of staged State infrastructure development applications and	
	5110	approvals	45
1	5.19	Determinations with respect to development subject to an approved	10
	5.10	staged State infrastructure development application	45
1	5.20	Modification of Minister's approval	46
	5.21	Validity of action under this Division	47
	5.22	Miscellaneous provisions relating to approvals under this Division	47
		wide of a new provision of relating to approvale and of the Division	
	Divis	sion 5.3 Public priority infrastructure	
ł	5.23	Declaration of public priority infrastructure	48
	5.24	Amendment or revocation of declaration	48
	5.25	Declared public priority infrastructure may be carried out	49
	5.26	Project definition report by proponent of public priority infrastructure	49
	5.27	Application of provisions of this Act to public priority infrastructure	49
	Cor	an and other legislative environments	
	Con	ncurrences, consultation and other legislative approvals	
	Divis	sion 6.1 Approvals under other legislation—public priority infrastructure, State infrastructure development and State significant development	
	6.1	Application and interpretation: Division 6.1	51
	6.2	Approvals that do not apply to public priority infrastructure, State	
		infrastructure development or State significant development	51
	6.3	Approvals that must be issued consistently for public priority	
		infrastructure, State infrastructure development or State significant	
		development	52
	6.4	Statutory directions, orders etc that cannot be made or given	53
	Divis	sion 6.2 Concurrences, consultation and other legislative	
		approvals—general provisions	
	6.5	Consultation with Minister administering threatened species legislation	
		if a Minister is consent authority or determining authority	53
	6.6	Concurrence of or consultation with responsible agency administering	
		threatened species legislation if a Minister is not consent authority or	
		determining authority	53
	6.7	Other consultation and concurrence requirements under planning	_
		control provisions	54
	6.8	Consultation and development consent—certain bush fire prone land	55

Part 6

				Page			
	6.9		may amend local plan for purpose of removing consultation, ence and other legislative provisions	56			
	Divis	sion 6.3	One stop referrals and decisions for other legislative approvals and for concurrences and consultation				
	6.10 6.11 6.12 6.13 6.14	Referral One sto One sto	ion and interpretation: Division 6.3 to and consultation by NSW Planning Director-General p referrals and decisions for other legislative approvals p referrals and decisions for consultation and concurrences heritage provision	57 57 58 59 59			
Part 7	Infra	astructu	are and other contributions				
	Divis	sion 7.1	Introductory				
	7.1 7.2	Definitio Applicat	ns ion of Part	61 61			
	Divis	sion 7.2	Local infrastructure contributions				
	7.3 7.4		frastructure contributions for proposed development utions to be in accordance with contribution provisions of an	61 62			
	7.5 7.6	Kinds of Nexus fo	f local infrastructure contributions—direct and indirect or contributions	62 63			
	7.7 7.8 7.9	Use of lo Complyi	nt of local infrastructure contributions ocal infrastructure contributions ing development	63 63 64			
	7.10 7.11 7.12	Cross-b	frastructure plans oundary issues frastructure may be provided outside NSW	64 64 65			
	7.13	Appeals		65			
	Division 7.3 Regional infrastructure contributions						
	7.14 7.15 7.16 7.17	How reg Paymen Complyi	on of regional infrastructure contributions gional infrastructure contributions are imposed at of regional infrastructure contributions ing development	65 66 66 66			
	7.18 7.19 7.20	arrange Growth	g control provisions of local plan for satisfactory ments for regional infrastructure infrastructure plans ions on appeals and changes to conditions	66 67 67			
		sion 7.4	Biodiversity offset contributions				
	7.21 7.22 7.23 7.24 7.25	Impositi How bio Paymen Complyi	on of biodiversity offset contributions odiversity offset contributions are imposed at of biodiversity offset contributions ing development ions on appeals and changes to conditions	67 68 68 68 69			
	Divis	sion 7.5	Planning agreements				
	7.26 7.27		pers can enter into planning agreements g agreements can limit contribution requirements	69 69			

			Page
	7.28	Use of planning agreement contributions	70
	7.29	Parties to planning agreements	70
	7.30	Limitations on planning agreements	70
	7.31	Contents of planning agreements	70
	7.32 7.33	Registered planning agreements to run with land Circumstances in which planning agreements can or cannot be	71
	1.55	required to be made	71
	7.34	No appeals	72
	7.35	Procedures and other requirements relating to planning agreements	72
	Divis	sion 7.6 Regional Contributions Fund	
	7.36	Definition	72
	7.37		72
	7.38	Payments into Fund	73
	7.39	Payments out of Fund	73
Part 8	Bui	Iding and subdivision	
	Divis	sion 8.1 Introductory	
	8.1	Interpretation: Part 8	74
	8.2	Kinds of certificates under Part 8	74
	8.3	Functions of building certifiers and subdivision certifiers	75
		sion 8.2 Building work and certificates relating to building	
	8.4	Requirements before building work commences	75
	8.5	Requirement for construction certificate	76
	8.6 8.7	Restriction on issue of construction certificate	77 77
	8.8	Requirement for occupation certificate Restrictions on issue of occupation certificates	77
	8.9	Effect of occupation certificate on earlier occupation certificates	78
		sion 8.3 Subdivision work and certificates relating to subdivisior	
	8.10	Requirements before subdivision work commences	78
	8.10	Requirement for subdivision works certificate	79
	8.12	Restriction on issue of subdivision works certificate	79
	8.13	Restrictions on issue of subdivision certificates	79
	Divis	sion 8.4 Compliance certificates	
	8.14	Requirement for compliance certificate	80
	8.15	Persons who may issue compliance certificates	80
	8.16	Restriction on issue of compliance certificates	80
	Divis	sion 8.5 Liability for defective building or subdivision work	
	8.17	Definitions	81
	8.18	Limitation on time when action for defective building or subdivision	~ .
	0.40	work may be brought	81
	8.19	Division not to affect rights to recover damages for death or personal injury	81
		пјату	01

				Page
	Divi	sion 8.6	Miscellaneous	
	8.20			04
	8.20 8.21		building manual building, demolition and incidental work	81 82
	8.21 8.22		s may be satisfied as to certain matters	82
	8.23		tion as to compliance with conditions precedent to the issue	02
	0.25	of certifi		83
	8.24		ns by certifiers	83
Part 9	Rev	views ar	nd appeals	
	Divis	sion 9.1	Introductory	
	9.1	Interpre	tation: Part 9	84
	Divis	sion 9.2	Reviews	
	9.2		nations and decisions subject to review	84
	9.3		tion for and conduct of review	84
	9.4		e of review	85
	9.5		aneous provisions relating to reviews	85
		sion 9.3	Appeals—development consents	
	9.6		ns subject to appeal to Court under this Division	86
	9.7		by applicant—applications for development consent	86
	9.8 9.9		by an objector—EIS assessed development applications by applicant—modifications of development consent	86 87
	9.9 9.10		thin which appeals may be made	87
	9.11		stances in which consent taken to have been refused for	07
	0.11		es of appeal rights	87
	9.12	· ·	of appeals to be given and right to be heard	87
	9.13		f appeals on operation of consents	88
	9.14		of Court on appeals	88
	9.15		neous provisions relating to appeals under this Division	89
	Divis	sion 9.4	Appeals—Part 8 certificates	
	9.16	Appeals	against failure or refusal to issue Part 8 certificates	89
	9.17	Deemeo	d refusal for purposes of appeal	90
	Divis	sion 9.5	Appeals—development control orders	
	9.18	Appeals	s concerning orders	90
	9.19		ng of compensation concerning orders	90
	9.20	Effect of	f appeal on order	91
	Divis	sion 9.6	Appeals—miscellaneous	
	9.21		concerning decisions on security for development	01
	9.22		nents or damage s against refusal to extend consent lapsing period	91 91
	9.22 9.23		s concerning compliance cost notices	91 91
	9.23 9.24		with respect to building information certificates	91
	0.27	, appears		52

Page	Ś
· ugu	•

Part 10	Enfo	prcement			
	Division 10.1 Ministerial enforcement powers				
	10.1	Directions by the Minister	93		
	10.2	Local government investigation into council's performance	93 93		
	10.3 Settlement of planning disputes				
	Divis	ion 10.2 Development control orders			
	10.4	Orders that may be given	94		
	10.5	Relevant enforcement authorities who may give orders	94 05		
	10.6 10.7	Provisions relating to orders Failure to comply with order—offence	95 95		
			00		
		ion 10.3 Civil enforcement proceedings	05		
	10.8 10.9	Definitions Restraint etc of breaches of this Act	95 96		
	10.10	Orders of the Court	96		
	10.11	Time within which validity of planning control provisions,			
		planning approvals etc may be challenged	97		
	10.12	Exclusion of legal proceedings	97		
	10.13	Special provision where development consent tainted by corruption	98		
		ion 10.4 Criminal offences and proceedings			
	10.14	Maximum penalty—Tier 1	99		
	10.15 10.16	Maximum penalty—Tier 2 Maximum penalty—Tier 3	100 100		
	10.10	Proceedings for offences	100		
	10.18	Criminal and civil enforcement proceedings in respect of same matter	101		
	10.19	Time within which proceedings may be commenced	101		
	10.20	Penalty notices for certain offences	101		
	10.21	Ancillary offences	102		
	10.22 10.23	Offence—false or misleading information Continuing obligations under orders etc	102 103		
	10.23	Miscellaneous provisions	103		
Dout 44					
Part 11		cellaneous			
	Divis	ion 11.1 Regulations			
	11.1	Regulations generally	105		
	11.2 11.3	Specific regulation-making powers Savings and transitional regulation-making powers	105 105		
			105		
		ion 11.2 Fees and charges	400		
	11.4 11.5	Fees and charges to which this Division applies Prescription of fees and charges and their payment	106 106		
		ion 11.3 Planning and building information certificates			
	11.6	Information certificates to which this Division applies	106		
	11.7	Who may apply for information certificates	106		
	11.8	Making of applications for information certificates	106		

		Page
11.9	Issue of information certificates	107
11.10	Nature and effect of planning information certificates	107
11.11	Issue, nature and effect of building information certificate	107
11.12	Miscellaneous provisions relating to building information certificates	108
	sion 11.4 Existing uses	
11.13	Definition of "existing use" etc	109
11.14 11.15	Continuance of and limitations on existing use	109
11.15	Regulations respecting existing use Continuance of and limitations on other lawful uses	109 110
11.17	Uses unlawfully commenced	110
11.18	Saving of effect of existing development consents	110
Divis	sion 11.5 Contaminated land liability	
11.19	Definitions	111
11.20	Exemption from liability—contaminated land	111
11.21	Contaminated land planning guidelines	111
	sion 11.6 Bush fire prone land	
11.22	Bush fire prone land	112
Divis	sion 11.7 General miscellaneous	
11.23	Act to bind Crown	112
11.24	Disclosure of political donations	112
11.25	Proof of ownership of land	116
11.26 11.27	Right to be heard—legal representation	116 116
11.27	Service or giving of documents Review of legislation	117
		117
Schedule 1	General—ancillary provisions	118
Schedule 2	Community participation—ancillary provisions	126
Schedule 3	Strategic planning—ancillary provisions	132
Schedule 4	Development (other than infrastructure) assessment	
	and consent—ancillary provisions	133
Schedule 5	Infrastructure and environmental impact	400
Schedule 6	assessment—ancillary provisions Concurrences, consultation and other legislative	138
Ochedule 0	approvals—ancillary provisions	142
Schedule 7	Infrastructure and other contributions—ancillary	
• • • • •	provisions	143
Schedule 8	Building and subdivision—ancillary provisions	147
Schedule 9	Reviews and appeals—ancillary provisions	149
Schedule 10	Enforcement—ancillary provisions	150
Schedule 11 Schedule 12	Miscellaneous—ancillary provisions	172
Schedule 12	Savings, transitional and other provisions Repeal of Acts and instruments	180 185
	Nopour of Aoto and moti amenito	100



New South Wales

Planning Bill 2013

No , 2013

A Bill for

An Act relating to planning and sustainable growth in New South Wales.

See also the Planning Administration Bill 2013.

The	Legisl	ature	of New	v South Wales enacts:	1
Par	't 1	Gei	neral		2
Divi	sion	1.1	Prel	iminary	3
1.1	Nam	e of A	ct		4
		This	Act is	the Planning Act 2013.	5
1.2	Com	mence	ement	of Act	6
		This	Act co	mmences on a day or days to be appointed by proclamation.	7
1.3	Obje	ects of	Act		8
	(1)	The	objects	of this Act are as follows:	9
		(a)	to pro	omote the growth of the State's economy and increased productivity,	10
		(b)	to pro	omote sustainable development (as set out in subsection (2)),	11
		(c)		ovide opportunities for early and on-going community participation in egic planning and to promote transparent decision-making,	12 13
		(d)		cilitate and manage growth by the co-ordination, planning, delivery and ration of infrastructure and services in strategic planning,	14 15
		(e)		romote the timely delivery of business, employment and housing rtunities (including for housing choice and affordable housing),	16 17
		(f)	to pro	omote the protection of the environment and heritage, including by:	18
			(i)	the conservation of biodiversity, and	19
			(ii)	the conservation and sustainable management of built and cultural heritage (including Aboriginal cultural heritage),	20 21
		(g)		able the effective management of natural hazards and natural resources, ding agricultural land, water and minerals,	22 23
		(h)	to pro build	omote health and safety in the design, construction and performance of ings,	24 25
		(i)		omote health, amenity and quality in the design and planning of the built conment,	26 27
		(j)		omote efficient and timely development assessment that is proportionate e likely impacts of proposed development,	28 29
		(k)		are responsibility between all levels of government for planning and the agement of growth.	30 31
	generation v needs. Susta environmen developmen			planning policies may inform any such decision-making about planning and	32 33 34 35 36 37
ייים	eion		opment		38
	sion			rpretation	39
1.4			s—gen		40
	(1)	Кеу	express	sions used in this Act are defined in this Division.	41

	(2)	inter Note	dule 1 contains a Dictionary of expressions used in this Act and other pretative provisions. The defined expressions apply generally to the planning legislation, which comprises act, the <i>Planning Administration Act 2013</i> and the instruments under those Acts.	1 2 3 4			
1.5	Deve	elopme	ent	5			
	(1)	For t	he purposes of this Act, <i>development</i> is any of the following:	6			
		(a)	the use of land,	7			
		(b)	the subdivision of land,	8			
		(c)	the erection of a building,	9			
		(d)	the carrying out of a work,	10			
		(e)	the demolition of a building or work,	11			
		(f)	any other act, matter or thing that may be controlled by the planning control provisions of a local plan.	12 13			
	(2)	regul	ever, development does not include any act, matter or thing excluded by the lations (either generally for the purposes of this Act or only for the purposes of ified provisions of this Act).	14 15 16			
	(3)		he purposes of this Act, the <i>carrying out of development</i> is the doing of the acts, ers or things referred to in subsection (1).	17 18			
1.6	Strat	Strategic plans and planning control provisions					
	(1)	For t	he purposes of this Act, a strategic plan is any of the following:	20			
		(a)	a NSW planning policy,	21			
		(b)	a regional growth plan,	22			
		(c)	a subregional delivery plan,	23			
		(d)	a local plan.	24			
		Note. of ma	. See section 3.1 for the definition of the <i>relevant planning authority</i> for the purposes aking strategic plans.	25 26			
	(2)		the purposes of this Act, the <i>planning control provisions</i> of a local plan prise:	27 28			
		(a)	the planning control provisions of the local plan (containing land use zoning, categories of development and other land use provisions) and the instruments that make, amend or replace those provisions, and	29 30 31			
		(b)	the provisions of any transitional planning instrument, being environmental planning instruments under the former Act that are taken by this Act to be a part of the planning control provisions of a local plan.	32 33 34			
1.7	Plan	ning a	ipprovals	35			
		For t	he purposes of this Act, a <i>planning approval</i> is any of the following:	36			
		(a)	a development consent (being a consent under Part 4 to carry out development, and includes a complying development certificate),	37 38			
		(b)	a State infrastructure approval (being an approval under Division 5.2 of Part 5 to carry out State infrastructure development),	39 40			
		(c)	a certificate under Part 8 other than a compliance certificate (being a construction certificate, an occupation certificate, a subdivision works certificate or a subdivision certificate).	41 42 43			

1.8	Infra	struct	ure plans and contributions	1
	(1)		he purposes of this Act, an <i>infrastructure plan</i> is a local infrastructure plan or a <i>v</i> th infrastructure plan.	2 3
	(2)	contr	he purposes of this Act, an <i>infrastructure contribution</i> is a local infrastructure ribution, or a regional infrastructure contribution, set out in the contribution isions of a local plan.	4 5 6
	(3)		he purposes of this Act, a <i>biodiversity offset contribution</i> is a biodiversity offset ribution set out in the contribution provisions of a local plan.	7 8
1.9	Build	ding w	ork and subdivision work	9
	(1)		he purposes of this Act, <i>building work</i> means any physical activity involved in rection of a building.	10 11
	(2)	autho a de subso regul appro	the purposes of this Act, <i>subdivision work</i> means any physical activity prised to be carried out in connection with a subdivision under the conditions of evelopment consent for the subdivision of land. For the purposes of this ection, a development consent includes a State infrastructure approval if the lations under Part 5 apply Part 8 to subdivision work under a State infrastructure oval. . See Schedule 1 for the meaning of <i>subdivision of land</i> .	12 13 14 15 16 17 18
1.10	Plan	ning b	oodies	19
		For t	he purposes of this Act, a <i>planning body</i> is any of the following:	20
		(a)	the Planning Ministerial Corporation,	21
		(b)	the Planning Assessment Commission,	22
		(c)	a regional planning panel,	23
		(d)	a subregional planning board,	24
		(e)	a planning committee or panel established by the Minister or the NSW Planning Director-General,	25 26
		(f)	an independent hearing and assessment panel of a council.	27
1.11	Deve	elopm	ent likely to significantly affect threatened species	28
			he purposes of this Act, development is <i>likely to significantly affect threatened ies</i> if:	29 30
		(a)	it is carried out in critical habitat, or	31
		(b)	it is likely to significantly affect threatened species, populations or ecological communities, or their habitats.	32 33
		is like and/c	See clause 1.5 of Schedule 1 for the determination of whether proposed development by to have such a significant effect. Schedule 1 defines threatened species, populations or ecological communities by reference to the <i>Threatened Species Conservation Act 1995</i> the case of fish and marine vegetation, Part 7A of the <i>Fisheries Management Act 1994</i> .	34 35 36 37
1.12	Biod	liversi	ty offsets	38
	(1)	bene for th	he purposes of this Act, <i>biodiversity offsets</i> for development are measures that fit biodiversity in or outside the site of the development to offset or compensate ne likely impacts of the development on biodiversity after all practical steps have taken to avoid or minimise those impacts.	39 40 41 42
	(2)	biodi	measures to benefit biodiversity are generally measures to benefit the iversity values impacted in the site or in the same locality as the site (or similar gher biodiversity values).	43 44 45

	(3)	The	measu	res to benefit biodiversity include (without limitation) the following:	1
		(a)	reser	vation of, or covenants on, land for the purposes of conservation,	2
		(b)	retiri	ng biodiversity credits under the threatened species legislation,	3
		(c)	eradi	cation or suppression of pests, weeds or predators of impacted species,	4
		(d)	resea	rch, surveys or educational programs with respect to biodiversity,	5
		(e)		ns under strategic government programs for the conservation or ncement of the natural environment of the State.	6 7
	(4)	offse	ets (or o	tions may specify measures that are included or excluded as biodiversity circumstances in which they are included or excluded).	8 9
		Note provi	. See se sions re	ections 6.6 and 6.9, Division 7.4 of Part 7 and clause 1.5 of Schedule 1 for relevant elating to biodiversity offsets.	10 11
Divi	sion	1.3	Dev	elopment—general provisions	12
1.13 Categories of development			velopment	13	
		For t	he pur	poses of this Act, there are the following categories of development:	14
		(a)		npt development (development that is exempt from the assessment and oval requirements of this Act),	15 16
		(b)	deve	lopment requiring development consent under Part 4, including:	17
			(i)	complying development (development that requires development consent by a consent authority or certifier),	18 19
			(ii)	code assessed development (development that is assessed in accordance with development assessment codes),	20 21
			(iii)	regionally significant development (development that requires development consent by a regional planning panel),	22 23
			(iv)	State significant development (development that requires development consent by the Minister),	24 25
			(v)	EIS assessed development (development that requires an environmental impact statement before consent can be granted and that gives objectors appeal rights),	26 27 28
				Note. Some EIS assessed development may also be regionally or State significant development.	29 30
		(c)		5 environmental impact assessment development (relevant development is subject to environmental impact assessment under Division 5.1 of 5),	31 32 33
		(d)		infrastructure development (development that requires the approval of Ainister under Division 5.2 of Part 5),	34 35
		(e)		ic priority infrastructure (development declared under Division 5.3 of 5 that does not require further planning approval).	36 37
1.14	Exen	npt de	evelop	ment	38
	(1)	The	carryin	g out of exempt development does not require:	39
		(a)		lopment consent, or	40
		(b)		5 environmental impact assessment, or	41
		(c)		infrastructure approval, or	42
		(d)	a cer	tificate under Part 8.	43
	(2)			velopment is development that is declared to be exempt development by a because of its minor impact.	44 45

1.15	Deve	lopm	ent that requires development consent under Part 4	1
		The	carrying out of development requires development consent under Part 4 if:	2
		(a)	a local plan provides that the development may be carried out with development consent or declares that it is complying development, and	3 4
		(b)	the development is not exempt development, and	5
		(c)	the development is not State infrastructure development or public priority infrastructure.	6 7
			Note. See also section 5.10 (5).	8
1.16	Deve	lopm	ent that requires State infrastructure approval under Part 5	9
		The	carrying out of development requires a State infrastructure approval if:	10
		(a)	the development is declared under Division 5.2 of Part 5 to be State infrastructure development, and	11 12
		(b)	the development is not public priority infrastructure, and	13
		(c)	the development is not exempt development (unless the application for State infrastructure approval specifically includes the development in the application for which approval is sought).	14 15 16
1.17	Carry certif	ying o ficate)	ut development without or contrary to planning approval (other than Part 8)	17 18
	(1)		erson must not carry out development without a planning approval that is ired under this Act for the carrying out of that development.	19 20
	(2)		rson must not, in carrying out development, contravene a planning approval that ies to the carrying out of that development.	21 22
	(3)		is section, <i>planning approval</i> does not include a certificate under Part 8.	23
			imum penalty: Tier 1.	24
		Note	. For civil enforcement—see Division 10.3 of Part 10.	25
1.18	Build	ling w	ork etc without or contrary to Part 8 certificate	26
	(1)		is section:	27
			ling work or related activity means:	28
		(a)	building work, or	29
		(b)	subdivision work, or	30
		(c)	the occupation or use of a building (including a change of use), or	31
		(d)	the subdivision of land, or	32
		(e)	any other activity to which Part 8 applies.	33
		certij	ficate under Part 8 does not include a compliance certificate.	34
	(2)		rson must not carry out any building work or related activity without a certificate or Part 8 that is required by this Act for that work or activity.	35 36
	(3)	a cer Max	rson must not, in carrying out any building work or related activity, contravene tificate under Part 8 that applies to the carrying out of that work or activity. imum penalty: Tier 1. . For civil enforcement—see Division 10.3 of Part 10.	37 38 39 40
1.19	Carry		out development that is prohibited	41
1.13	(1)	A pe	erson must not carry out any development that is prohibited by the planning rol provisions of a local plan unless the carrying out of the development is	41 42 43

(2)

permitted by a strategic compatibility certificate, by a State infrastructure approval, as public priority infrastructure or under any provision of this Act. Note. Section 4.29 (4) enables consent for State significant development that is partly prohibited.	1 2 3 4
For the purposes of this section, development that the planning control provisions of a local plan declare cannot be carried out with or without development consent is taken to be development prohibited by those provisions.	5 6 7
Maximum penalty: Tier 1.	8

Maximum penalty: Tier 1.

Note. For civil enforcement—see Division 10.3 of Part 10.

Part	t 2	Со	mmunity participation	1
2.1	The Community Participation Charter		nunity Participation Charter	2
	(1)	The	Community Participation Charter comprises the following:	3
		(a)	The community has a right to be informed about planning matters that affect the community.	4 5
		(b)	Planning authorities should encourage the establishment of effective and on-going partnerships with the community in order to identify meaningful opportunities for the community to participate in planning.	6 7 8
		(c)	Planning information should be in plain language, readily accessible and in a form that facilitates community participation in planning.	9 10
		(d)	The community should be given opportunities to participate in strategic planning as early as possible to enable community views to be genuinely considered.	11 12 13
		(e)	Community participation methods should be appropriate having regard to the significance and likely impact of the proposed development.	14 15
		(f)	Community participation should be inclusive and planning authorities should actively seek views that are representative of the community.	16 17
		(g)	Planning decisions should be made in an open and transparent way and the community should be provided with reasons for planning decisions (including how community views have been taken into account).	18 19 20
	(2)	legis other	the purposes of the Community Participation Charter and the planning slation, <i>community participation</i> in strategic planning, planning decisions and r planning matters is the process of engaging the community (including industry, nesses, residents, interest groups and organisations) in those planning matters.	21 22 23 24
2.2	Planr	ning a	authorities and functions to which Charter applies	25
	(1)	-	Community Participation Charter applies to the following planning authorities:	26
		(a)	the Minister,	27
		(b)	the NSW Planning Director-General,	28
		(c)	the Planning Assessment Commission,	29
		(d)	a regional planning panel,	30
		(e)	a subregional planning board,	31
		(f)	a council,	32
		(g)	an independent hearing and assessment panel of a council,	33
		(h)	a determining authority under Division 5.1 of Part 5,	34
		(i)	a public authority prescribed by the regulations.	35
	(2)	The	Community Participation Charter applies to the exercise of the following ning functions:	36 37
		(a)	functions under section 2.4 (Community participation plans),	38
		(b)	strategic planning functions under Part 3,	39
		(c)	development consent functions under Part 4,	40
		(d)	environmental impact assessment functions under Division 5.1 of Part 5 if an environmental or species impact statement is required,	41 42
		(e)	State infrastructure approval functions under Division 5.2 of Part 5,	43

(f) infrastructure plan functions under Part 7.

2.3	Appl	Applying the Charter in the exercise of planning functions				
	(1)	he Community Participation Charter when exercising a planning function to which	2 3 4			
	(2)		5 6			
			7 8			
		community participation under the planning legislation and that community 1	9 10 11			
2.4	Com	unity participation plans	2			
	(1)	equired to prepare a community participation plan about how it will undertake community participation when exercising planning functions to which the Charter applies (subject to subsection (2)).	3 4 5 6			
	(2)		8			
		(a) a community participation plan prepared by the NSW Planning 1 Director-General applies to the exercise of planning functions by the Minister, 2	19 20 21			
		Director-General applies to the exercise of planning functions by determining authorities under Division 5.1 of Part 5 (other than councils or prescribed 2	22 23 24 25			
		planning authority applies to the exercise of planning functions by another planning authority and that the other planning authority is not required to 2	26 27 28 29			
	(3)	A community participation plan: 3	30			
			31 32			
		discretionary community participation when exercising planning functions, 3	33 34 35			
		requirements with respect to the plan that are published by the NSW Planning 3	36 37 38			
	(4)	he matters required under this section in the community participation plan of the 4	39 10 11			
	(5)	Community participation plans are to be published on the NSW planning portal. 4	12			
	(6)	Community participation plans are to be reviewed periodically. 4	13			
	(7)	procedures for making and publishing community participation plans (or any 4	14 15 16			

	(8)	The regulations (or the requirements published by the NSW Planning Director-General) may provide for the inclusion in community participation plans of guidance relating to the responsibilities of the community with respect to community participation.	1 2 3 4
2.5	Mand	latory community participation requirements	5
		Part 1 of Schedule 2 sets out the mandatory requirements for community participation by planning authorities with respect to the exercise of planning functions to which the Community Participation Charter applies. Note. The mandatory requirements include public exhibition for a minimum period, public notification requirements and the giving of reasons for decisions.	6 7 8 9 10
2.6	Publi	c notice of community participation in proposed strategic plans	11
	(1)	Before a proposed strategic plan is prepared for public exhibition, the planning authority is to give public notice of the ways in which the community can participate in the preparation of the proposed plan.	12 13 14
	(2)	The regulations may prescribe the manner in which public notice is to be given.	15
2.7	Onlin	e planning services and information	16
2.7	Onlin (1)	The NSW Planning Director-General is to establish and facilitate the online delivery of planning services and information (including the NSW planning portal).	16 17 18
2.7		The NSW Planning Director-General is to establish and facilitate the online delivery	17
2.7 2.8	(1) (2)	The NSW Planning Director-General is to establish and facilitate the online delivery of planning services and information (including the NSW planning portal).	17 18
	(1) (2)	The NSW Planning Director-General is to establish and facilitate the online delivery of planning services and information (including the NSW planning portal). Part 3 of Schedule 2 contains provisions with respect to those matters.	17 18 19
	(1) (2) Com	The NSW Planning Director-General is to establish and facilitate the online delivery of planning services and information (including the NSW planning portal). Part 3 of Schedule 2 contains provisions with respect to those matters. munity Participation Advisory Panel The Minister is required to establish a Community Participation Advisory Panel	17 18 19 20 21
	 (1) (2) Common (1) 	The NSW Planning Director-General is to establish and facilitate the online delivery of planning services and information (including the NSW planning portal). Part 3 of Schedule 2 contains provisions with respect to those matters. munity Participation Advisory Panel The Minister is required to establish a Community Participation Advisory Panel under section 2.3 of the <i>Planning Administration Act 2013</i> .	17 18 19 20 21 22
	 (1) (2) Common (1) 	 The NSW Planning Director-General is to establish and facilitate the online delivery of planning services and information (including the NSW planning portal). Part 3 of Schedule 2 contains provisions with respect to those matters. munity Participation Advisory Panel The Minister is required to establish a Community Participation Advisory Panel under section 2.3 of the <i>Planning Administration Act 2013</i>. The Panel has the following functions: (a) to provide advice on the preparation and operation of community participation 	17 18 19 20 21 22 23 24

Par	t 3	Stra	ategic planning	1	
Divi	sion:	3.1	Introductory	2	
3.1	Defin	finition of "relevant planning authority"			
		In thi	s Part:	4	
		releva	ant planning authority for a strategic plan means:	5	
		(a)	in the case of a NSW planning policy or regional growth plan—the NSW Planning Director-General, or	6 7	
		(b)	in the case of a subregional delivery plan—the relevant subregional planning board, or	8 9	
		(c)	in the case of a local plan—the relevant planning authority designated under Division 3.3.	10 11	
		Note. plan, a	This Act generally defines a <i>strategic plan</i> as a NSW planning policy, a regional growth a subregional delivery plan or a local plan.	12 13	
3.2	Prepa	aring o	or making strategic plans	14	
			he purposes of this Part, preparing or making a strategic plan (however ibed) includes:	15 16	
		(a)	preparing or making a strategic plan to amend, replace or repeal a strategic plan, and	17 18	
		(b)	in the case of a local plan—preparing or making provisions of the local plan (or amending, replacing or repealing those provisions).	19 20	
Divi	sion:	3.2	NSW planning policies, regional growth plans and subregional delivery plans	21 22	
3.3	Prepa	aratior	n and content of NSW planning policies	23	
	(1)	The N	NSW Planning Director-General may prepare draft NSW planning policies.	24	
	(2)	are to perform	ft NSW planning policy is to identify the policies and principles of the State that o be implemented by strategic plans and infrastructure plans (including rmance requirements for planning authorities concerning the achievement of policies and principles).	25 26 27 28	
3.4	Prepa	aratior	n and content of regional growth plans	29	
	(1)		NSW Planning Director-General may prepare a draft regional growth plan for egion in the State.	30 31	
	(2)	A dra	ft regional growth plan is to identify the following:	32	
		(a)	the basis for strategic planning in the region,	33	
		(b)	existing and proposed transport and other infrastructure for the region (including any priority infrastructure),	34 35	
		(c)	regionally significant areas,	36	
		(d)	targets for achieving the planning outcomes for the region (including housing, employment and environmental targets),	37 38	
		(e)	actions required to be undertaken by planning authorities to achieve those targets,	39 40	
		(f)	the basis on which planning authorities are to monitor and report on performance against those targets,	41 42	

		(g)	any other matters the NSW Planning Director-General considers relevant to planning for the region.	1 2
	(3)	grow	re is no subregional delivery plan for any part of the region, the draft regional th plan may identify for that part of the region matters that may be identified in	3 4
			regional delivery plan (until there is a subregional delivery plan). Section 7.19 relates to the preparation of growth infrastructure plans for regions in the	5 6 7
3.5	Prepa	aratior	n and content of subregional delivery plans	8
	(1)		pregional planning board may prepare a draft subregional delivery plan for the gion for which it has been established.	9 10
	(2)	A dra	ft subregional delivery plan is to identify the following:	11
		(a)	existing and proposed transport and other infrastructure for the subregion (including any priority infrastructure),	12 13
		(b)	significant areas in the subregion,	14
		(c)	how the housing, employment and environmental targets in the relevant regional growth plan are to be achieved in the subregion,	15 16
		(d)	proposed growth areas in the subregion and the proposed planning controls that should apply in those growth areas or the strategic planning process that should be undertaken by planning authorities to establish those planning controls,	17 18 19 20
		(e)	proposed exempt or complying development or development proposed for code assessment in the subregion,	21 22
		(f)	any other matters the subregional planning board considers relevant to planning for the subregion.	23 24
	(3)	delive identi	re is no regional growth plan for any part of the subregion, the draft subregional ery plan may identify for that part of the subregion matters that may be ified in a regional growth plan (until there is a regional growth plan). Section 7.19 relates to the preparation of growth infrastructure plans for subregions in ate.	25 26 27 28 29
3.6	Maki plans		NSW planning policies, regional growth plans and subregional delivery	30 31
	(1)	grow	elevant planning authority may submit a draft NSW planning policy, regional th plan or subregional delivery plan it has prepared to the Minister. Schedule 2 requires a draft plan to be publicly exhibited for at least 28 days.	32 33 34
	(2)	The N delive	Ainister may make a NSW planning policy, regional growth plan or subregional ery plan in the form in which it was submitted or with such modifications as the ster considers appropriate. The Minister may decide not to make the draft policy	35 36 37 38
	(3)		n submitting a draft policy or plan to the Minister, the relevant planning rity is to provide the Minister with the following:	39 40
		(a)	any public submissions (or summary of submissions) that have been made about the draft policy or plan,	41 42
		(b)	any submissions that have been made by a public authority about the draft policy or plan,	43 44
		(c)	if the Planning Assessment Commission has held a public hearing into the draft plan—the findings and recommendations of the Commission about the draft plan.	45 46 47

3.7		ementa ery pla	ation of NSW planning policies, regional growth plans and subregional ans	1 2
		A rel	levant planning authority is to give effect to:	3
		(a)	NSW planning policies when preparing other draft strategic plans, and	4
		(b)	regional growth plans when preparing draft subregional delivery plans and local plans, and	5 6
		(c) Note. plann	subregional delivery plans when preparing draft local plans. . Part 7 requires persons preparing draft infrastructure plans to have regard to NSW ing policies.	7 8 9
Divi	sion	3.3	Local plans—general	10
3.8	Loca	l plan	s for local government and other areas	11
	(1)		I plans are established for development in the local government area of each cil. Any such local plan is called the "[name of council] Local Plan".	12 13
	(2)	(inclu local	cal plan is also established for development in any other area of the State uding the coastal waters of the State) that is declared by the regulations to be a plan area. Any such local plan is called the "[<i>name assigned by the regulations</i>] Il Plan".	15
	(3)	of a s that a	regulations may assign an area that is outside a local government area to be part specified adjoining local government area for the purposes of the local plan of adjoining area. For the purposes of the planning legislation, the assigned area is a to be a part of the local government area concerned.	19
	(4)		regulations may make transitional provisions relating to local plans when there change in the local government area of a council.	22 23
3.9	Com	positio	on of local plans	24
		A loc	cal plan established under this Division comprises the following:	25
		Part	1—Strategic context:	26
			An explanation of how NSW planning policies, regional growth plans, subregional delivery plans and infrastructure plans are given effect to in the area concerned (having regard to any applicable community strategic plan under section 402 of the <i>Local Government Act 1993</i>).	28
		Part	2—Planning controls:	31
			Containing spatial and other provisions for the area concerned relating to land use zoning, the categories of development and other matters for which planning control provisions may be made under this or any other Act.	
		Part	3—Development codes and guides:	35
			Containing:	36
			(a) development assessment codes and codes identifying exempt or complying development, and	37 38
			(b) development guides for development in the area to give effect to the aims of planning control provisions and the objectives of land use zones, and to facilitate permissible development.	
		Part	4—Contributions:	42
			Containing provisions relating to the amount of local and regional infrastructure contributions, or biodiversity offset contributions, payable in relation to particular kinds of development in the area concerned.	

3.10	Rele	vant p	planning authority for provisions of local plan	1
	(1)		the purposes of this Division, the <i>relevant planning authority</i> for the provisions local plan is as follows:	2 3
		(a)	the relevant council, subject to paragraphs (b) and (c),	4
		(b)	for the regional infrastructure or biodiversity offset contribution provisions— the NSW Planning Director-General,	5 6
		(c)	if the Minister so directs under subsection (2) for proposed provisions—the NSW Planning Director-General, a regional planning panel or any other person or body prescribed by the regulations.	7 8 9
	(2)	body	Minister may direct that the NSW Planning Director-General (or other person or referred to in subsection (1) (c)) is the relevant planning authority for proposed isions of a local plan in any of the following cases:	10 11 12
		(a)	the proposed provisions relate to a matter that, in the opinion of the Minister, is of State, regional or subregional planning significance,	13 14
		(b)	the Planning Assessment Commission or a regional planning panel has recommended to the Minister that the proposed provisions should be submitted to the Minister for consideration or that the proposed provisions should be made,	15 16 17 18
		(c)	the proposed provisions are to apply to an area that is not within the area of a council.	19 20
	(3)	its fu requi prop	levant planning authority that is requested by the owner of any land to exercise inctions under this Part in relation to the land may, as a condition of doing so, ire the owner to carry out studies or provide other information concerning the osed provisions or to pay the costs of the authority in accordance with the lations.	21 22 23 24 25
3.11	Mak	ing pro	ovisions of local plans	26
	(1)	The	Minister may, in accordance with this Part, make any provisions of a local plan.	27
	(2)	The oplan:	council of the area of a local plan may make the following provisions of the local	28 29
		(a)	any provisions that the council is authorised to make by the gateway determination of the proposal for those provisions,	30 31
		(b)	the local infrastructure contribution provisions.	32
	(3)	the s those maki gatev	the case of the planning control provisions of a local plan, Division 3.4 deals with submission to the Minister by the relevant planning authority of proposals for e provisions of the local plan, the gateway determination of the proposals and the ing of those provisions. The regulations may apply the planning proposal, way determination and other provisions of Division 3.4 (with or without ification) to the making of development code or guide provisions of a local plan.	33 34 35 36 37 38
	(4)	subn may	the case of other provisions of a local plan, the relevant planning authority may nit to the Minister draft provisions of a local plan it has prepared. The Minister make those provisions of a local plan in the form in which they were submitted ith such modifications as the Minister considers appropriate (or decide not to do	39 40 41 42 43

	(5)	of a l Note. makin Note. (such	Minister may delegate, under the planning legislation, to the council of the area ocal plan or to another public authority the making of provisions of a local plan. Part 1 of Schedule 2 specifies the mandatory public exhibition requirements for the ng of provisions of a local plan. Under proposed transitional arrangements, former environmental planning instruments as SEPPs and LEPs under the former Act), and former development control plans and butions plans, will continue in force as part of the relevant provisions of a local plan.	1 2 3 4 5 6 7		
Divi	sion	3.4	Local plan—planning control provisions	8		
3.12	Appl	icatio	n of Division	9		
			Division applies to the planning control provisions of a local plan and to the ng of those provisions.	10 11		
3.13	Purp	oses f	or which planning control provisions may be made	12		
		achie	ning control provisions of a local plan may be made for the purposes of eving any of the objects of this Act and, in particular, for or with respect to any e following:	13 14 15		
		(a)	protecting, improving or utilising, to the best advantage, the environment,	16		
		(b)	controlling development (including by land use zoning or by prescribing development standards),	17 18		
		(c)	identifying different categories of development for the purposes of this Act (including State infrastructure development, State significant development, regionally significant development, EIS assessed development and code assessed development),	19 20 21 22		
		(d)	identifying the consent authority for development for the purposes of Part 4 (if not otherwise identified by this Act),	23 24		
		(e)	requiring the consent authority functions of a council to be exercised by (or an application for development consent to be assessed by) an independent hearing and assessment panel of the council,	25 26 27		
		(f)	identifying whether development does or does not require development consent to be carried out, or is prohibited,	28 29		
		(g)	requiring (before development consent is granted or before an application for development consent is made) the concurrence of or consultation with a Minister or other public authority,	30 31 32		
		(h)	protecting trees,	33		
		(i)	protecting native animals and plants, including threatened species, populations and ecological communities, and their habitats,	34 35		
		(j)	identifying and protecting State or local heritage items and conservation areas,	36		
		(k)	encouraging the provision of affordable housing and housing choice,	37		
		(1)	such other matters as are authorised or required to be included in the planning control provisions of a local plan by or under this or any other Act.	38 39		
			See, for example, sections 7.18 and 8.3.	40		
3.14			iated acquisition of land zoned for public purposes	41		
	(1)	This section applies in relation to the planning control provisions of a local plan that zone land for use exclusively for any of the following public purposes (as determined in accordance with section 21 of the <i>Land Acquisition (Just Terms Compensation) Act 1991</i>):				
		(a)	open space,	46		

		(b)	public place or public reserve within the meaning of the <i>Local Government Act</i> 1993,	1 2			
		(c)	national park or other land reserved under the National Parks and Wildlife Act 1974,	3 4			
		(d)	public transport corridor,	5			
		(e)	public cemetery,	6			
		(f)	public hospital,	7			
		(g)	public school,	8			
		(h)	a purpose declared by the regulations to be a public purpose.	9			
	(2)	Planning control provisions to which this section applies must specify an authority of the State that will be the relevant authority to acquire the land if the land is required to be acquired under Division 3 of Part 2 of the <i>Land Acquisition (Just Terms Compensation) Act 1991</i> . An authority of the State cannot be so specified unless it has notified the relevant planning authority that it concurs in those planning control provisions.					
	(3)	requi	anning control provision (whenever made and despite any of its terms) cannot ire an authority of the State to acquire land, except as required by Division 3 of 2 of the <i>Land Acquisition (Just Terms Compensation) Act 1991</i> .	16 17 18			
3.15	Stan	Standardisation of planning control or other provisions of local plans					
	(1)	The Governor may, by order published on the NSW legislation website, prescribe the standard form and content of the planning control provisions of a local plan or any part of those provisions (a <i>standard instrument</i>).					
	(2)	The p	planning control provisions of a local plan may be made in the form of:	23			
		(a)	a declaration that the applicable mandatory provisions of a standard instrument are adopted, and	24 25			
		(b)	the prescription of the matters required to be prescribed for the purposes of the application of the mandatory provisions of the standard instrument (such as the adoption of land zoning or other maps or spatial datasets), and	26 27 28			
		(c)	the prescription of any other matters permitted to be prescribed by the planning control provisions of a local plan, including non-mandatory provisions of the standard instrument (with or without modification) or additional provisions.	29 30 31 32			
	(3)	When planning control provisions of a local plan are made with such a declaration, the provisions have the form and content of the applicable mandatory provisions of the standard instrument and the matters so prescribed.					
	(4)	If the mandatory provisions of a standard instrument so adopted are amended by a further order under subsection (1) or by an Act after they are adopted, the planning control provisions of a local plan are taken (without further amendment) to adopt the amended provisions of the standard instrument on and from the date the amendment to the standard instrument takes effect.					
	(5)	The order that amends a standard instrument may make provisions of a savings or transitional nature consequent on the amendment of the standard instrument.					
	(6)	local	re a standard instrument has been adopted, the planning control provisions of the plan (other than the mandatory provisions of the adopted standard instrument) be amended from time to time under this Part or in accordance with any Act.	43 44 45			

- (7)A standard instrument may:
 - (a) provide that a provision is a mandatory provision only in the circumstances specified in the instrument, and

1

2

3

4

5

6

7

8

q

10

11

12

13

14

15

16

22

24

25

26

30

31

32

33

36

- (b) contain requirements or guidance as to the form or content of a non-mandatory provision.
- (8) Subject to this Act and the regulations, the form and subject-matter of the planning control provisions of a local plan are (if there is no applicable standard instrument) to be as determined by the Minister.
- (9)The regulations may extend the application of this section to other provisions of a local plan that are not planning control provisions. In that case, a reference in this section to planning control provisions extends to those other provisions.

3.16 Relevant planning authority to prepare planning proposal for proposed planning control provisions

- Before planning control provisions of a local plan are made, the relevant planning (1)authority is required to prepare a document that describes the proposed provisions (the *planning proposal*).
- (2)A planning proposal is to contain such detail and information about the proposed 17 provisions as is proportionate and appropriate to the nature of the proposed 18 provisions. 19
- (3) The regulations may make provision with respect to the preparation of planning 20 proposals or different kinds of planning proposals (including the form and content of 21 planning proposals and the circumstances in which a planning proposal can take the form of draft planning control provisions). 23

3.17 **Gateway determination**

- After preparing a planning proposal, the relevant planning authority may forward it (1)to the Minister.
- (2)The Minister may, for the purposes of the review of a planning proposal (or part of a 27 planning proposal) under this section, refer the matter to the Planning Assessment 28 Commission or a regional planning panel for advice. 29
- (3) After reviewing the planning proposal, the Minister must determine the following:
 - whether the matter should proceed (with or without variation), (a)
 - (b) the minimum period of public exhibition for the planning proposal (or a determination that no such public exhibition is required).
- (4)After reviewing the planning proposal, the Minister may also determine the 34 following: 35
 - (a) whether the matter should be resubmitted for any reason (including for further studies or other information, or for the revision of the planning proposal), 37
 - (b) any consultation required with State or Commonwealth public authorities that 38 may be affected by the proposed planning control provisions, 39
 - whether a public hearing is to be held into the matter by the Planning (c) 40 Assessment Commission, 41
 - (d) the times within which the various stages of the procedure for the making of 42 the proposed planning control provisions are to be completed by the relevant 43 planning authority, 44
 - (e) if the relevant planning authority is a council—whether the council is 45 authorised to make the proposed planning control provisions instead of the 46

Minister and any conditions the council is required to comply with before the provisions are made.

1

2

3

4

5

6

7

8

9

10

11

12

13

22

23

24

25

26

27

30

31

32

33

34

35

36

- (5) A determination under this section is the *gateway determination* for a planning proposal.
- (6) The Minister may, at any time, alter a gateway determination.
- (7)A failure to comply with a requirement of a gateway determination in relation to proposed planning control provisions does not invalidate the provisions once they are made. However, this subsection does not apply to mandatory public exhibition requirements.

3.18 Relevant planning authority may vary proposals or not proceed

- (1)The relevant planning authority may, at any time, vary its proposals as a consequence of its consideration of any submission during public exhibition or for any other reason.
- The relevant planning authority may, at any time, forward a revised planning (2)14 proposal to the Minister. 15
- (3) If the relevant planning authority is a council and is authorised by the gateway 16 determination to make the proposed provisions instead of the Minister, the council is 17 required to forward the revised planning proposals to the Minister. The council 18 continues to be authorised to make the proposed provisions, subject to any conditions 19 of the gateway determination for the making of the provisions by the council and to 20 any revised gateway determination of the matter. 21
- (4) Further public exhibition of a revised planning proposal is not required unless the Minister so directs in a revised gateway determination.
- The relevant planning authority may also, at any time, request the Minister to (5) determine that the matter not proceed.

3.19 Final planning proposal

- After the completion of mandatory public exhibition and when submitting the final (1)planning proposal to the Minister, the relevant planning authority is to provide the 28 Minister with the following: 29
 - any public submissions (or summary of submissions) that have been made (a) about the planning proposal during public exhibition,
 - (b) any submissions that have been made by a public authority about the planning proposal,
 - if the Planning Assessment Commission has held a public hearing into the (c) planning proposal—a report of any findings and recommendations made by the Commission about the planning proposal.
- If the relevant planning authority is a council and is authorised by the gateway (2)37 determination to make the proposed planning control provisions instead of the 38 Minister, the council is not required to submit the final planning proposal to the 39 Minister, subject to any conditions of the gateway determination for the making of 40 the provisions by the council. 41
- (3) The NSW Planning Director-General is to make arrangements for the final drafting 42 of any required planning control provisions to give effect to the final proposals of the 43 relevant planning authority. 44

3.20	Making of planning control provisions				
	(1)	The Minister may:		2	
		(a) make planning control provisions of a local plan, with or without w the final proposals submitted by the relevant planning authority, or		3 4	
		(b) decide not to make the proposed planning control provisions.		5	
	(2)	The Minister may defer the inclusion of a matter in proposed planni provisions.	ng control	6 7	
	(3)	If the Minister does not make the proposed planning control provisions or inclusion of a matter in the proposed planning control provisions, the Mi specify which procedures under this Division the relevant planning auth comply with before the matter is reconsidered by the Minister.	nister may	8 9 10 11	
	(4)	If the relevant planning authority is a council and is authorised by th determination to make the proposed planning control provisions inste Minister, a reference in this section to the Minister includes a refere council, subject to any conditions of the gateway determination for the ma provisions by the council.	ead of the nce to the	12 13 14 15 16	
		Note. In addition to the gateway determination authorising a council to make the planning control provisions, the Minister may under the <i>Planning Administratic</i> delegate to the relevant planning authority or other authorised person the mat proposed planning control provisions.	n Act 2013	17 18 19 20	
3.21	Conc	urrent local plan amendments and applications for development con	sent	21	
	(1)	This section applies if an application is made to a consent authority for de consent to carry out development that may only be carried out if the planm or other provisions of the local plan applying to the land concerned are applicated.	ing control	22 23 24 25	
	(2)	Nothing in this Act prevents, subject to this section, the making of that a and its consideration by the consent authority.	application	26 27	
	(3)	Public exhibition required under Part 1 of Schedule 2 in connection with t of the proposed planning control or other provisions and public exh notification required under Part 1 of Schedule 2 in connection with that a are to be undertaken together or as closely together as is practicable.	nibition or	28 29 30 31	
	(4)	If the mandatory public exhibition requirements for the proposed planning other provisions are different to the public exhibition requirement application, the more extensive requirements apply for both of them.		32 33 34	
3.22		ension of covenants, agreements and instruments by planning contr sions	ol	35 36	
	(1)	For the purpose of enabling development to be carried out in accordance planning control provisions of a local plan (as in force from time to t accordance with a planning approval granted under this Act, a plannin provision of the local plan may provide that, to the extent necessary to purpose, any agreement, covenant or similar instrument specified in that does not apply to any such development or applies subject to the mo- specified in that provision.	ime) or in ng control serve that t provision	37 38 39 40 41 42 43	
	(2)	That planning control provision has effect according to its tenor.		44	
	(3)	If a Minister is responsible for the administration of the agreement, co similar instrument, that planning control provision is not to be made w concurrence of that Minister.		45 46 47	

	(4)	A declaration in that planning control provision as to the concurrence of that Minister is evidence of the concurrence.	1 2
	(5)	The provisions of this section have effect despite anything contained in section 42 of the <i>Real Property Act 1900</i> .	3 4
3.23	Misc	ellaneous provisions relating to planning control provisions	5
	(1)	Mandatory public exhibition is completed for the purposes of this Division when the relevant planning authority has considered any public submissions (or summary of submissions) duly made during public exhibition concerning the planning proposal and a report of the findings and recommendations of the Planning Assessment Commission following any public hearing.	6 7 8 9 10
	(2)	A development guide provision of a local plan (whenever made) has no effect to the extent that:	11 12
		(a) it is the same or substantially the same provision as a planning control or code provision of the local plan, or	13 14
		(b) it is inconsistent or incompatible with a planning control or code provision of the local plan.	15 16
	(3)	The planning control provisions of a local plan do not apply to development to which Part 5 applies unless the provisions determine whether development is subject to Part 5 or the provisions are expressly stated to apply to development of that kind.	17 18 19
	(4)	The planning control provisions of a local plan may make provision for any zoning of land or other provision to have effect only for a specified period or only in specified circumstances.	20 21 22
	(5)	The planning control provisions of a local plan that make provision for or with respect to protecting or preserving trees may make provision for the grant of permission to remove or otherwise affect trees, and for a refusal to grant permission to be treated as a refusal or failure to grant development consent under and for the purposes of Part 4.	23 24 25 26 27
	(6)	Sections 24, 28, 30, 42, 43, 45, 75 and 80 of the <i>Interpretation Act 1987</i> apply to the planning control provisions of a local plan in the same way as they apply to statutory rules.	28 29 30
Divi	sion	3.5 Strategic plans—miscellaneous provisions	31
3.24	Gen	eral considerations for preparation of proposed strategic plans	32
	(1)	A relevant planning authority that prepares a draft strategic plan (or proposal for a strategic plan) under this Part should have regard to the following:	33 34
		(a) that provisions of strategic plans are based on evidence,	35
		(b) the need to co-operate constructively with State agencies in preparing the draft plan (or proposal).	36 37
	(2)	A relevant planning authority that prepares draft provisions (or proposals) for a local plan under this Part should also have regard to the following:	38 39
		(a) the impact on the financial feasibility of future development,	40
		(b) that the extent and complexity of provisions of local plans should be proportionate to the likely impacts of future development.	41 42
3.25	Publ	lic exhibition of proposed strategic plans	43
	(1)	The regulations may make provision for the documents that are to accompany the public exhibition of a draft strategic plan (or proposal for a strategic plan). The	44 45

3.26

	asses	ations may, for example, require a neighbourhood impact statement or strategic sment statement to accompany the public exhibition of a plan or proposal for a lopment assessment code.	1 2 3
(2)		aft strategic plan (or proposal for a strategic plan) may be publicly exhibited her with another draft strategic plan (or proposal for a strategic plan).	4 5
(3)	propo	e mandatory public exhibition requirements for the draft strategic plans (or osals) are different, the more extensive requirements apply for all of them. See also Schedule 2 for provisions relating to public exhibition.	6 7 8
Publi	icatior	n of strategic plans	9
	A stra	ategic plan is made:	10
	(a)	in the case of planning control provisions of local plans—when published on the NSW legislation website, or	11 12
		in any other case—when published on the NSW planning portal. Planning control provisions are also to be published on or accessible from the NSW ing portal.	13 14 15
Disp	ensing	g with conditions precedent to making strategic plans	16
(1)		ategic plan may be made without compliance with the conditions precedent r the planning legislation to doing so in order to do any one or more of the wing:	17 18 19
	(a)	to correct an obvious error or misdescription or effect other minor law revision,	20 21
	(b)	to rezone land or make other changes as a consequence of any development that is made permissible with development consent by a strategic compatibility certificate if development consent has been granted for the development,	22 23 24 25
	(c)	to make changes that will not have any significant adverse impact on the environment or adjoining land.	26 27
(2)	under follo	ategic plan may be made without compliance with the conditions precedent r the planning legislation to doing so in order to do any one or more of the wing if the proposed plan has been publicly exhibited for the period determined e Minister:	28 29 30 31
	(a)	to give effect in an expeditious manner to a strategic plan or infrastructure plan,	32 33
	(b)	to make provision for matters that are, in the opinion of the Minister, of State, regional or subregional significance,	34 35
	(c)	to declare the development whose likely effect on threatened species may be assessed in accordance with a biodiversity assessment procedure adopted by the regulations (as referred to in clause 1.5 (2) of Schedule 1),	36 37 38
	(d)	to declare classes of development for which the consent authority functions of a council are to be exercised by (or an application for development consent is to be assessed by) an independent hearing and assessment panel of the council.	39 40 41
(3)		publication of a strategic plan made in reliance on subsection (2) is to contain a nent of the reasons why the plan is being made.	42 43
(4)		publication of a strategic plan made in reliance on subsection (1) or (2) is to in a statement that it is so made.	44 45

3.28 Periodic review and staged repeal of strategic plans 1 (1)The relevant planning authority is to keep strategic plans under regular and periodic 2 review to ensure they continue to achieve the objects of this Act to the maximum 3 extent possible, having regard to changing circumstances. 4 (2)The regulations may establish a staged repeal program for any class of provisions of 5 local plans or of other strategic plans to facilitate the replacement of transitional 6 planning control provisions or to ensure the regular and periodic review of strategic 7 plans. 8 (3)The staged repeal program may provide for the repeal of provisions of local plans or 9 of other strategic plans by the operation of the regulations and for the making of 10 replacement provisions or plans. 11

Part	: 4		velopment (other than infrastructure) assessment I consent	1 2
Division 4.1		i .1	Introductory	3
4.1	Appli	catior	n: Part 4	4
	(1)		Part applies to development that requires development consent under this Part carried out.	5 6
			See sections 1.15 and 1.17 for development that requires consent.	7
	(2)	certif for a	Part applies to development consent in the form of a complying development icate, except that Divisions 4.4, 4.5, 4.6 and 4.8 do not apply to an application complying development certificate or to a development consent in the form of a certificate.	8 9 10 11
		land a certific erection compliations	A complying development certificate for the erection of a building or the subdivision of also authorises building work or subdivision work without the need for a construction cate or subdivision works certificate under Part 8. Any other development consent for the on of a building or the subdivision of land generally does require such a certificate. A lying development certificate or other development consent for the erection of a building not generally authorise the occupation or use of a building without an occupation cate (see Part 8).	12 13 14 15 16 17 18
4.2	Interp	oretati	ion: Part 4	19
			s Part:	20
			<i>fier</i> means a building certifier or subdivision certifier.	21
		as ref	<i>assessment</i> means the determination of an application for development consent Ferred to in section 4.17.	22 23
			<i>tassessment</i> means the determination of an application for development consent ferred to in section 4.18.	24 25
		stage	<i>d development application</i> —see section 4.21.	26
4.3	Deve	lopme	ent assessment tracks: Part 4	27
		Appli	ications for development consent under this Part may be determined as follows:	28
		(a)	complying development (being the determination by a council or other certifier of an application for a complying development certificate),	29 30
		(b)	code assessment (being the code assessment of development by a consent authority for the determination of an application for development consent),	31 32
		(c)	merit assessment (being the merit assessment of development by a consent authority for the determination of an application for development consent).	33 34
		includ	The categories of development relevant to this Part are set out in section 1.13. They le State significant development, regionally significant development, EIS assessed opment and complying development.	35 36 37
Divis	sion 4	1.2	Consent authorities	38
4.4	Cons	ent au	uthorities	39
	(1)		<i>consent authority</i> for the purposes of development that requires development ent under this Part is as follows:	40 41
		(a)	for complying development the subject of an application for a complying development certificate—a certifier,	42 43
		(b)	for State significant development—the Minister,	44

		(c)	for regionally significant development—the regional planning panel for the area in which the development is to be carried out (subject to subsection (2)),	1 2
		(d)	for other development—a Minister or public authority declared to be the consent authority by the planning control provisions of a local plan that apply to the development (or the council for the area concerned if there is no consent authority so declared for the development).	3 4 5 6
	(2)	has t exerc	e case of regionally significant development, the regional planning panel only the function of determining the application for development consent (and cising other specified functions of the consent authority as provided by this Act he regulations). The council for the area is in all other respects the consent prity.	7 8 9 10 11
	(3)	asses only funct	e planning control provisions of a local plan require an independent hearing and ssment panel to determine an application for development consent, the panel has the function of determining the application (and exercising other specified tions of the consent authority as provided by this Act or the regulations). The cil for the area is in all other respects the consent authority.	12 13 14 15 16
	(4)		blic authority has the functions conferred or imposed on it as consent authority e planning legislation.	17 18
4.5	Dele	gation	of consent authority functions	19
			blic authority may delegate its functions as consent authority under the powers erred on it by an Act that authorises the delegation of the functions of the public ority.	20 21 22
Divi	sion	4.3	Complying development	23
4.6	Appl	icatio	n for complying development certificate	24
	(1)		application for a complying development certificate to carry out complying lopment may be made to a certifier.	25 26
	(2)		application may be made only by the owner of the land on which the lopment is to be carried out or by any other person with the consent of the owner.	27 28
4.7	Varia	ations	from complying development standards and requirements	29
	(1)	certif a stai	buncil may, on the application of an applicant for a complying development ficate, issue a certificate (a <i>variation certificate</i>) certifying that a variation from indard or requirement in the development guide provisions of the local plan that plicable to an aspect of the development is a permissible variation.	30 31 32 33
	(2)	satisf	buncil may issue a variation certificate for an aspect of a development if it is fied that non-compliance with the standard or requirement is not likely to have significant additional adverse impact on development on the surrounding land.	34 35 36
	(3)	certif	regulations may limit the standards or requirements for which a variation ficate may be issued (or limit the extent of permissible variations to standards or irements).	37 38 39
	(4)	A va	riation certificate relating to an application made to a council for a complying	40
		devel	lopment certificate may be issued by the council without an application being e for the variation certificate.	41 42
4.8		devel made	lopment certificate may be issued by the council without an application being	
4.8		devel made	lopment certificate may be issued by the council without an application being e for the variation certificate.	42

		(b)	by refusing to issue a complying development certificate.	1
	(2)	devel comp plan local	certifier must issue a complying development certificate if the proposed lopment is complying development and complies with the standards for that blying development in the development code or guide provisions of the local and any other applicable requirements of the planning control provisions of the plan or the regulations (subject to any permissible variation authorised by a tion certificate under this Division).	2 3 4 5 6 7
	(3)	any c comp	e proposed development is not complying development or does not comply with of those applicable standards or requirements, the certifier must refuse to issue a olying development certificate. However, the certifier must before doing so m the applicant of the applicant's rights to seek a variation certificate.	8 9 10 11
	(4)	must	regulations may provide that a determination of an application by the certifier be completed within a specified period after the application was made (or such er period as may be agreed to by the applicant).	12 13 14
4.9	Build	ding C	ode of Australia requirements to apply to complying development	15
	(1)	unles relev	rtifier must not issue a complying development certificate for building work as the proposed building (not being a temporary building) will comply with the ant requirements of the <i>Building Code of Australia</i> (as in force at the time the cation for the certificate was made).	16 17 18 19
	(2)	This	section is subject to the regulations.	20
4.10	Form	n of co	mplying development certificate	21
	(1)	A co	mplying development certificate is to be in the form of a certificate that:	22
		(a)	states that particular proposed development is complying development and (if carried out as specified in the certificate) will comply with all applicable standards and requirements, and	23 24 25
		(b)	in the case of development involving the erection of a building, identifies the classification of the building in accordance with the <i>Building Code of Australia</i> .	26 27 28
	(2)	The d build	certificate may indicate different classifications for different parts of the same ing.	29 30
4.11	Cond	ditions	of a complying development certificate	31
	(1)		mplying development certificate is subject to the conditions prescribed by the ations for the development concerned.	32 33
	(2)	deter	omplying development certificate is subject to any additional conditions mined by the certifier in accordance with the regulations and endorsed on the ficate.	34 35 36
4.12	Dura	tion o	f complying development certificate	37
	(1)		mplying development certificate has effect on and from the date it is registered e NSW planning portal.	38 39
	(2)	A co	mplying development certificate lapses 5 years after the date so registered.	40
	(3)	whic	ever, a complying development certificate does not lapse if the development to h it relates is physically commenced on the land to which the certificate applies n the period of 5 years after the date so registered.	41 42 43
	(4)	The tribut	5-year period cannot be extended by a certifier or a council or any court or nal.	44 45

4.13	Modification of complying development				
	(1)		on who has the benefit of a complying development certificate may apply to a or to modify the development the subject of the certificate or the terms of the ate.	2 3 4	
	(2)		ivision applies to an application for modification in the same way as it applies original application.	5 6	
Divi	ision	4.4 (Code and merit assessment	7	
4.14	Арр	ication f	for development consent	8	
	(1)		plication for development consent to carry out development may be made to evant consent authority.	9 10	
	(2)	The app	plication may be made only:	11	
		(a) t	by the owner of the land on which the development is to be carried out, or	12	
		(b) t	by any other person with the consent of that owner, or	13	
		(c) a	as otherwise authorised by the regulations.	14	
	(3)	A singl develop	le application for development consent may be made for one or more kinds of pment.	15 16	
	(4)	this Di	plication may be made to a consent authority for development consent under ivision to carry out complying development instead of for a complying pment certificate under Division 4.3.	17 18 19	
		Note. Papulot	art 1 of Schedule 2 requires certain applications for development consent to be exhibited and contains provisions for the notification of other applications.	20 21	
4.15	Dete	rminatio	on of applications for development consent	22	
	(1)	A cons	ent authority may determine an application for development consent:	23	
			by granting development consent, either unconditionally or subject to conditions, or	24 25	
		(b) t	by refusing to grant development consent.	26	
	(2)		nsent authority is to make that determination in accordance with this Division y other relevant provisions of this Act or the regulations.	27 28	
	(3)	subdivi plannin	onsent authority must refuse an application for development consent to the ision of land that would, if carried out, result in a contravention of this Act, the ng control provisions of the local plan or the regulations, whether arising in a to that or any other development.	29 30 31 32	
	(4)	an appl of any a or not g	ent authority has a general obligation in connection with the determination of lication for development consent to advise the applicant as soon as practicable aspects of the development that may lead to a refusal to grant consent. Giving giving advice on an aspect of development does not bind or affect the consent ty when determining the application.	33 34 35 36 37	
	(5)	A deve	lopment consent may be granted:	38	
		(a) f	for the development for which the consent is sought, or	39	
		(b) f	for that development, except for a specified aspect of that development, or	40	
		(c) f	for a specified aspect of that development.	41	
	(6)		nsent authority is not required to refuse consent to any specified aspect of pment for which development consent is not initially granted under	42 43	

subsection (5), but development consent may subsequently be granted for that aspect of the development.

(7) A development consent that authorises the erection of a building (but not the use of the building once erected) is sufficient to authorise the use of the building when erected for the purpose for which it was erected if that purpose was specified in the application for development consent. This section does not authorise the occupation of such a building if Part 8 requires an occupation certificate to be issued.

4.16 Development assessment codes

- (1) A *development assessment code* is a code for development set out in the development code provisions of a local plan.
- (2) A development assessment code is to set out the development standards for the development. The code may also set out performance criteria for the development and acceptable solutions for achieving those performance criteria.
- (3) If the planning control provisions of a local plan identify development subject to code assessment, the relevant development assessment code (and relevant performance criteria and acceptable solutions) apply to the code assessment of an application for development consent to carry out the development.
- (4) Relevant development assessment codes (including relevant performance criteria and acceptable solutions for those criteria) are also to be taken into consideration in merit assessment as a guide to appropriate development.
 18
 19
 20

4.17 Code assessment

21

27

33

34

37

1

2

3

4

5

6

7

8

9

10

11

12

- An application for development consent is subject to code assessment under this section if the development is identified in the planning control provisions of the local plan as development subject to code assessment and if the development meets all the development standards for the development under the applicable development assessment code, but is not subject to code assessment if:
 - (a) the development is EIS assessed development, or
 - (b) the application for development consent is made in reliance on a strategic 28 compatibility certificate under Division 4.7, or 29
 - (c) the development is subject to a requirement for concurrence or consultation
 under Division 6.2 of Part 6 because it is likely to significantly affect
 threatened species, or
 32
 - (d) the development requires an approval under Subdivision 1 of Division 3 of Part 4 of the *Heritage Act 1977*, or
 - (e) the development requires an Aboriginal heritage impact permit under section 90 of the National Parks and Wildlife Act 1974.
 36
- (2) If an application for development consent is code assessed:
 - (a) the consent authority cannot refuse to grant development consent on grounds
 related to the development standards for the development under the applicable
 development assessment code, and
 40
 - (b) the consent authority cannot impose conditions that are more onerous than those development standards. 41
- (3) Development that is subject to code assessment is to be assessed against the performance criteria in the applicable development assessment code for any aspect of the development. If acceptable solutions for those performance criteria have been adopted for that aspect of the development (or alternative solutions have been 46

adopted that, in the opinion of the consent authority, satisfy those performance 1 criteria for that aspect of the development): 2 (a) the consent authority cannot refuse to grant development consent on grounds 3 related to the performance criteria for that aspect of the development, and 4 (b) the consent authority cannot impose conditions that are more onerous than 5 those acceptable solutions. 6 (4)The consent authority can impose a condition on development consent to require 7 compliance with those performance criteria. 8 (5) If alternative solutions have been adopted that, in the opinion of the consent 9 authority, do not satisfy those performance criteria: 10 the consent authority is to take into consideration all relevant provisions of the (a) 11 local plan in determining the application for development consent, and 12 (b) the consent authority cannot impose conditions that are more onerous than the 13 acceptable solutions for those performance criteria. 14 Merit assessment 15 An application for development consent is subject to merit assessment under this (1)16 section if it is not subject to code assessment under section 4.17. 17 (2)In determining an application for development consent to which this section applies, 18 a consent authority is to take into consideration the following: 19 (a)the local plan (including proposed provisions of the local plan that have been 20 publicly exhibited under this Act), 21 any submissions (or a summary of submissions) duly received during public (b) 22 exhibition under this Act in connection with the application, 23

- (c) the likely impacts of the development, including:
 - (i) any environmental impacts on the natural and built environments, and

24

25

26

29

30

31

32

33

36

37

- (ii) any economic and social impacts in the locality, and
- (iii) in the case of regionally significant development, any economic and 27 social impacts in the region, and 28
- (iv) in the case of State significant development, any economic and social impacts in the State,
- (d) any planning agreement referred to in Part 7 that is or is to be entered into in connection with the development,
- (e) the public interest (in the context of the objects of this Act).
- (3) In the case of an application for development consent that is made in reliance on a strategic compatibility certificate under Division 4.7: 35
 - (a) the consent authority is also to take into consideration the terms and conditions of the certificate, and
 - (b) the determination of the application is subject to section 4.37 (Development 38 consents to be granted consistently with certificate). 39
- (4) The consent authority is not required, in the case of State significant development, to take into consideration the performance criteria for development in the development guide provisions of a local plan unless the criteria are expressly stated to apply to State significant development.
 40
 41
 42
 43
- (5) The regulations may further restrict refusal to grant consent or the imposition of
 conditions where the development meets developments standards, or adopts
 45

acceptable or alternative solutions for relevant performance criteria, set out in the development code or guide provisions of the local plan.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

32

33

35

36

37

38

44

- (6)If the planning control provisions of a local plan or the regulations identify a development standard as a non-discretionary development standard in respect of any aspect of development and proposed development complies with the development standard:
 - the consent authority cannot refuse consent on grounds related to the (a)development standard, and
 - (b) the consent authority cannot impose conditions that are more onerous than the development standard.

4.19 EIS assessed development that is not State significant development—additional requirements

- (1)An application for development consent for EIS assessed development must be accompanied by an environmental impact statement prepared by or on behalf of the applicant in accordance with the regulations. Note. Schedule 2 requires the application (and accompanying EIS) to be publicly exhibited for at least 28 days.
- (2)If submissions have been duly made during the period of public exhibition, the consent authority is not to determine the application until the expiry of the period of 21 days (or other prescribed period) following the date on which a copy of the submissions is forwarded to the NSW Planning Director-General.
- (3)Subsection (2) does not apply:
 - (a) if the consent authority is the Minister or the NSW Planning Director-General, or
 - (b) if the NSW Planning Director-General has waived the requirement that submissions be forwarded to the NSW Planning Director-General for a specified application or for a specified class of applications.
- (4)This section does not apply to EIS assessed development that is also State significant 28 development. 29 Note. Division 4.6 requires an EIS for EIS assessed development that is also State significant 30 development. 31

4.20 Determination must await any PAC public hearing or advice

- This section applies to an application for development consent where the advice of, (1)or a public hearing by, the Planning Assessment Commission in relation to all or any 34 aspect of the development has been requested by the Minister under the planning legislation.
- (2)The consent authority must not determine the application until:
 - the Planning Assessment Commission has dealt with the request, and (a)
 - (b) consent authority has considered the advice, findings and 39 recommendations of the Planning Assessment Commission, and 40
 - if the Minister is not the consent authority, the consent authority has (c) 41 considered any comments made by the Minister that accompanied the findings 42 and recommendations when they were forwarded to the consent authority. 43

4.21 Staged development applications and consents

For the purposes of this Part, a *staged development application* is an application for (1)45 development consent that sets out concept proposals for the development and for 46 which detailed proposals for separate parts of the development are to be the subject 47

of subsequent applications for development consent. The application may set out detailed proposals for the first stage of development.

1

2

7

8

9

10

11

12

13

27

32

33

34

35

36

37

38

39

40

41

- (2) An application for development consent is not to be treated as a staged development application unless the applicant requests it to be treated as a staged development pplication.
 (3) A local plan cannot require the making of a staged development application before 6
- (3) A local plan cannot require the making of a staged development application before development is carried out.
- (4) A staged development application is to be determined in accordance with the applicable provisions relating to applications for development consent (subject to any special provisions that apply to staged development applications).
- (5) If development consent is granted on the determination of a staged development application (a *staged development consent*), the consent does not authorise the carrying out of any part of the development unless:
 - (a) development consent is subsequently granted to carry out that part of the
 development following a further application for development consent in
 respect of that part of the development, or
 - (b) the staged development application also provided the requisite details of that
 part of the development and development consent is granted for that part of the
 development without the need for further consent.

The terms of a staged development consent are to reflect the operation of this 20 subsection. 21

(6) While any staged development consent remains in force, the determination of any further application for development consent for the development concerned cannot be inconsistent with the staged development consent. This subsection does not prevent the modification in accordance with this Act of a staged development consent.
 22 23 24
 23 24 25
 24 25 26

4.22 Date from which development consent has effect

- (1) A development consent for any development has effect on and from the date it is registered on the NSW planning portal, except as provided by subsection (2). 29
- A development consent for EIS assessed development has effect on and from the end of 28 days after the date it is registered on the NSW planning portal unless: 31
 - (a) the development consent was granted following a public hearing by the Planning Assessment Commission, or
 - (b) the development is State significant development.

Note. The date of effect of a consent for any such EIS assessed development is delayed by the period within which an objector may appeal to the Land and Environment Court against the grant of consent.

(3) This section is subject to Part 9.

Division 4.5 Conditions of development consent

4.23 Imposition of conditions—general

- (1) The following conditions of development consent may be imposed:
 - (a) a condition that relates to any matter that is required by this Part to be taken
 42 into consideration in the assessment of whether the development consent is to
 43 be granted,
 44

	(b)	a condition that requires the modification or surrender of a development consent, or an existing or continuing use right under Division 11.4 of Part 11, in relation to the land the subject of the application for development consent,	1 2 3
	(c)	a condition that requires the modification or cessation of development (including the removal of buildings and works used in connection with the development) carried out on the land subject to the application for	4 5 6
	(d)	development consent or on other land, a condition that limits the period during which development may be carried out in accordance with the consent so granted (or that requires the removal of buildings and works at the end of that period),	7 8 9 10
	(e)	a condition that requires the carrying out of works (relating to any matter that is required by this Part to be taken into consideration in the assessment of whether the development consent is to be granted) on land not subject to the application for development consent, if those works are reasonably required and the imposition of the condition complies with the regulations,	11 12 13 14 15
	(f)	a condition that modifies details of the development,	16
	(g)	a condition that requires a specified aspect of the development that is ancillary to the core purpose of the development to be carried out to the satisfaction of the consent authority or of a person specified by the consent authority (as determined in accordance with the regulations),	17 18 19 20
	(h)	a condition that is authorised to be imposed under any other provision of the planning legislation.	21 22
(2)	A con	ndition of a development consent has no effect to the extent that:	23
	(a)	it is inconsistent with a condition prescribed by the regulations to which the development consent is made subject by the regulations, or	24 25
	(b)	it is substantially the same condition as any such condition prescribed by the regulations.	26 27
(3)		ndition of a development consent has no effect to the extent that it requires a liance certificate under Part 8 to be obtained in respect of any development.	28 29
Cond	litions	that identify outcomes	30
	A dev mann	velopment consent may be granted subject to a condition that is expressed in a er that identifies both of the following:	31 32
	(a)	one or more express outcomes that the development or a specified aspect of the development must achieve,	33 34
	(b)	clear criteria against which the achievement of the outcome can be assessed.	35
"Def	erred o	commencement" conditions	36
(1)	to ope	velopment consent may be granted subject to a condition that the consent is not erate until the applicant satisfies the consent authority as to any matter specified e condition.	37 38 39
(2)		ing in the planning legislation prevents a person from doing anything necessary mply with the condition.	40 41
Revi	ewable	e conditions of development consent	42
(1)	made by the	velopment consent that is granted subject to a reviewable condition may also be subject to a further condition that the reviewable condition may be reviewed e consent authority at any time or at intervals specified in the consent and that eviewable condition may be changed on any such review.	43 44 45 46

4.24

4.25

	(2)	A decision by a consent authority to change a reviewable condition on a review is taken to be a determination of a development consent for the purposes of this Act. Note. As a consequence, the decision to change the reviewable condition may be subject to review or appeal under Part 9.	1 2 3 4
	(3)	In this section, a <i>reviewable condition</i> means any of the following:	5
		(a) a condition that permits extended hours of operation (in addition to other specified hours of operation),	6 7
		(b) a condition that increases the maximum number of persons permitted in a building (in addition to the maximum number otherwise permitted),	8 9
		(c) a condition of a kind prescribed by the regulations.	10
4.27	Cono dama		11 12
	(1)	may enter into an agreement with an applicant for development consent, that the	13 14 15 16
		(a) making good any damage caused to any property of the consent authority as a consequence of the doing of anything to which the consent relates,	17 18
		footway construction, stormwater drainage and environmental controls)	19 20 21
			22 23
	(2)		24 25
	(3)	The security may be provided, at the applicant's choice, by way of:	26
		(a) deposit with the consent authority, or	27
		(b) a guarantee satisfactory to the consent authority.	28
	(4)	development consent or at such other time as may be agreed to by the consent	29 30 31
	(5)		32 33
	(6)	The funds realised from a security may be paid out to meet any cost for which the security was provided. Any balance remaining is to be refunded to, or at the direction of, the persons who provided the security.	34 35 36
Divi	sion	.6 State significant development—additional provisions	37
4.28	"Cal	in" of development as State significant development	38
	(1)	The Minister may, by Ministerial planning order, declare specified development on specified land to be State significant development.	39 40
	(2)	made publicly available advice from the Planning Assessment Commission about the	41 42 43
	(3)	Any development for which consent is granted (or purports to be granted) by the Minister as State significant development is taken to be State significant	44 45

development and to have been such development for the purposes of any application or other matter under this Part in relation to the development.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

22

23

24

25

43

4.29 Environmental impact assessment for, and consent to, State significant development etc

- An application for development consent for State significant development is (if it is also EIS assessed development) to be accompanied by an environmental impact statement prepared by or on behalf of the applicant in accordance with the regulations.
 Note. Schedule 2 requires the application (and accompanying EIS) to be publicly exhibited for at least 28 days.
- (2) In the case of State significant development that is not also EIS assessed development and that is of a class prescribed by the regulations:
 - (a) the NSW Planning Director-General may issue environmental assessment requirements for an application for development consent for that development, and
 - (b) the application is to be accompanied by a statement of the environmental effects of the development prepared by or on behalf of the applicant in accordance with the regulations.
- (3) Development consent cannot be granted for State significant development if the development is wholly prohibited by the planning control provisions of the local plan.
 (3) Development consent cannot be granted for State significant development if the planning control provisions of the local plan.
- (4) However, development consent may be granted for State significant development despite the development being partly prohibited by the planning control provisions of the local plan. If development consent is granted, any such prohibition does not apply to the carrying out of the development.
- (5) If part of a single proposed development that is State significant development
 (5) If part of a single proposed development that is State significant development
 (5) 26
 (5) 27
 (6) 27
 (7) 28
 (7) 28
 (7) 28
 - (a) Divisions 5.1 and 5.2 of Part 5 do not apply to that other part of the proposed 29 development, and 30
 - (b) that other part of the proposed development is taken to be development 31 requiring development consent under this Part. 32

This subsection does not apply if the NSW Planning Director-General determines that the other part of the proposed development is not sufficiently related to the State significant development, or if it is specified development on specified land declared to be State infrastructure development by the Minister under Division 5.2 of Part 5.

- (6) The Minister may also grant development consent for State significant development
 37 with such modifications of the proposed development as the Minister may determine.
 38
- (7) The Minister in determining an application for development consent for State significant development may impose such conditions on the development consent as the Minister determines, despite any restriction in this Part on the kinds of conditions that may be imposed.
 (7) The Minister in determining an application for development consent for State significant development may impose such conditions on the development consent as the Minister determines, despite any restriction in this Part on the kinds of conditions 41 42

4.30 Staged development applications for State significant development

- When determining a staged development application to carry out State significant
 development, the Minister may make any (or any combination) of the following
 determinations:
 - (a) the Minister may determine the further environmental assessment 47 requirements for development consent to carry out the subsequent stages of the 48

development (in which case those requirements have effect for the purposes of this Part),

- (b) the Minister may determine that no further environmental impact assessment is required for development consent to carry out the development or subsequent stages of the development (in which case development consent may be granted or refused without further application or assessment under this Part),
- (c) the Minister may determine that a subsequent stage of the development is to be complying development, regionally significant development, EIS assessed development, development for which the council is the consent authority or Part 5 environmental impact assessment development (in which case that stage of the development ceases to be State significant development),
- (d) the Minister may determine the conditions that are to apply to any development consent for a subsequent stage of the development that the Minister has determined will not be dealt with as State significant development (in which case the subsequent development consent is to be generally consistent with those conditions),
- (e) the Minister may determine that the planning control provisions of a local plan
 that would otherwise prohibit any subsequent stage of the development do not
 apply when a council or regional planning panel determines an application to
 carry out that stage of the development (in which case those provisions have
 effect subject to that determination).
- (2) A determination under this section is taken to form part of the determination of the staged development application.
- (3) The Minister may vary or revoke a determination under this section on the application of the staged development applicant or a person having the benefit of the staged development consent.
- (4) When modifying a staged development consent for State significant development, the Minister may make, vary or revoke a determination under this section.

4.31 PAC to determine certain applications

The Minister is required to delegate to the Planning Assessment Commission the function of determining an application for development consent to any State significant development if the development would have been wholly prohibited but for planning control provisions of a local plan made by the Commission under delegation from the Minister. 35

Division 4.7 Strategic compatibility certificates—development not permissible under local plan

4.32 Prohibited development permissible with consent if strategic compatibility certificate issued

- A *strategic compatibility certificate* is a certificate issued under this Division that certifies that the carrying out of specified development on specified land is permissible with development consent under this Part, despite any prohibition on the carrying out of the development or any development standard under the planning control provisions of the local plan.
- A strategic compatibility certificate has effect as if it formed part of the planning control provisions of the local plan.
 45
 46
- (3) Despite anything to the contrary in the planning control provisions of the local plan,
 47 merit assessment under section 4.18 (Merit assessment) applies to the determination
 48

30

36

37

38

39

23

24

25

26

27

28

29

1

2

3

4

5

6

7

8

9

10

11

of an application for development consent made in reliance on the certificate. Those 1 planning control provisions determine whether the development is EIS assessed 2 development, regionally significant development or State significant development. 3 4.33 Who may issue certificate 4 A strategic compatibility certificate may be issued by: (1)5 the regional planning panel for the area in which the development is to be (a) 6 carried out, or 7 the NSW Planning Director-General, but only if satisfied that there are not (b) 8 more than 25 public objectors and that the relevant council has not objected to 9 the issue of the certificate when consulted under this Division. 10 (2)For the purposes of subsection (1), a public objector is a person who has duly made 11 a submission by way of objection during the public exhibition of the application for 12 the certificate. 13 4.34 Grounds on which certificate may be issued 14 A strategic compatibility certificate may be issued for development only if the 15 regional planning panel or NSW Planning Director-General (as the case requires) is 16 satisfied that: 17 a regional growth plan or subregional delivery plan has been made that applies (a)18 to the development, and 19 (b) the planning control provisions prohibiting the development have not yet been 20 amended to give effect to the relevant provisions of that plan, and 21 (c) the development is consistent with that plan, and 22 (d) the development will not have any significant adverse impact on likely future 23 uses of the surrounding land. 24 4.35 Requirements before certificate is issued 25 A strategic compatibility certificate is not to be issued unless an application is duly (1)26 made for the certificate by the owner of the land concerned or a person who has the 27 consent of the owner to make the application. 28 (2)An application for a certificate is to be made to the NSW Planning Director-General. 29 (3)The NSW Planning Director-General is to consult the relevant council before a 30 strategic compatibility certificate is issued. 31 Note. Schedule 2 requires an application for a certificate (and any application for development 32 consent made in reliance on a certificate) to be publicly exhibited for at least 28 days. 33 (4) Unless the NSW Planning Director-General determines the application, the NSW 34 Planning Director-General is to forward the application to the applicable regional 35 planning panel together with advice in relation to the determination of the 36 application. 37 (5)In determining an application for a certificate, the regional planning panel or the 38 NSW Planning Director-General (as the case requires) is to take into account: 39 any submissions (or a summary of submissions) duly received during public (a) 40 exhibition under Schedule 2 in connection with the application, and 41 (b) the views of the relevant council, and 42 in the case of a regional planning panel, the advice provided by the NSW (c) 43 Planning Director-General. 44 45

Note. Schedule 2 requires the panel or Director-General to give public notification of the determination of the application and the reasons for the determination.

4.36 Conditions and duration of certificate

	COIN	ditions and duration of certificate	1
	(1)	A strategic compatibility certificate may be issued subject to conditions relating to the standards for the development, environmental impact assessment or community participation requirements or other relevant matter.	2 3 4
	(2)	The conditions on any matter to which a strategic compatibility certificate is subject replace the otherwise applicable provisions of the planning control provisions of the local plan on that matter.	5 6 7
	(3)	An application under this Part for the carrying out of development permitted by a strategic compatibility certificate may only be made in reliance on the certificate within 12 months after the certificate is issued.	8 9 10
	(4)	If an application is not made within that period, or if development consent is refused on the determination of an application made within that period (including after any review or appeal), the certificate then ceases to have effect.	11 12 13
4.37	Dev	elopment consents to be granted consistently with certificate	14
	(1)	An application for development consent that is made in reliance on a strategic compatibility certificate cannot be refused, and conditions cannot be imposed on the development consent, if the grounds for refusal or the imposition of conditions relate to matters dealt with in the conditions of the certificate.	15 16 17 18
	(2)	This section does not affect a refusal to grant development consent or the imposition of conditions on other grounds.	19 20
	(3)	An application for development consent that is made in reliance on a proposed strategic compatibility certificate may be publicly exhibited together with the application for the certificate.	21 22 23
Divi	sion	4.8 Modification of development consents	24
4.38	Who	may make application for modification of development consent	
		may make application for mounication of development consent	25
		A development consent may be modified under this Division only on the application of the applicant for the development consent or of a person having the benefit of the development consent.	25 26 27 28
4.39		A development consent may be modified under this Division only on the application of the applicant for the development consent or of a person having the benefit of the	26 27
4.39		A development consent may be modified under this Division only on the application of the applicant for the development consent or of a person having the benefit of the development consent.	26 27 28
4.39 4.40	Wha	A development consent may be modified under this Division only on the application of the applicant for the development consent or of a person having the benefit of the development consent. t is a modification of development consent A modification of a development consent under this Division is a change to the terms of the consent, including the revocation or variation of a condition of the consent or	26 27 28 29 30 31
	Wha	A development consent may be modified under this Division only on the application of the applicant for the development consent or of a person having the benefit of the development consent. t is a modification of development consent A modification of a development consent under this Division is a change to the terms of the consent, including the revocation or variation of a condition of the consent or the imposition of additional conditions of consent.	26 27 28 29 30 31 32
	Wha Who	A development consent may be modified under this Division only on the application of the applicant for the development consent or of a person having the benefit of the development consent. t is a modification of development consent A modification of a development consent under this Division is a change to the terms of the consent, including the revocation or variation of a condition of the consent or the imposition of additional conditions of consent. may modify development consent A development consent may be modified under this Division by the consent authority that granted the consent, subject to and in accordance with this Act and the	26 27 28 30 31 32 33 33 34 35
4.40	Wha Who	A development consent may be modified under this Division only on the application of the applicant for the development consent or of a person having the benefit of the development consent. t is a modification of development consent A modification of a development consent under this Division is a change to the terms of the consent, including the revocation or variation of a condition of the consent or the imposition of additional conditions of consent. may modify development consent A development consent may be modified under this Division by the consent authority that granted the consent, subject to and in accordance with this Act and the regulations.	26 27 28 29 30 31 32 33 33 34 35 36
4.40	Wha Who Mod	A development consent may be modified under this Division only on the application of the applicant for the development consent or of a person having the benefit of the development consent. t is a modification of development consent A modification of a development consent under this Division is a change to the terms of the consent, including the revocation or variation of a condition of the consent or the imposition of additional conditions of consent. may modify development consent A development consent may be modified under this Division by the consent authority that granted the consent, subject to and in accordance with this Act and the regulations. ifications that may be made to development consents	26 27 28 29 30 31 32 33 34 35 36 37

any modification if the development is State significant development. (c)

(2)The mandatory public exhibition and notification requirements for applications for the modification of development consents (other than modifications of the kind referred to in subsection (1) (a)) are the requirements prescribed by the regulations made under Schedule 2.

1

2

3

4

5

6

7

8

9

14

23

24

25

26

27

28

29

30

31

32

33

34

35

36

- (3)A development consent may be modified only if the development as proposed to be modified is development for which the consent authority may grant development consent in accordance with this Part. Accordingly, the obligations of a consent authority under this Part in relation to the determination of an application for development consent extend to applications for the modification of development consents (other than modifications of the kind referred to in subsection (1) (a)). 10
- (4)However, the modification of a development consent is taken not to be the granting 11 of development consent under this Part, but a reference in this or any other Act to a 12 development consent is a reference to a development consent as so modified. 13

4.42 Provisions relating to Land and Environment Court

- The power of a consent authority to modify a development consent under this (1)15 Division extends to a development consent granted by the Land and Environment 16 Court, but only if the consent authority has notified, or made reasonable attempts to 17 notify, each person who made a submission in respect of the relevant application for 18 the development consent of the proposed modification by sending written notice to 19 the last address of the person known to the consent authority. 20
- (2)The Land and Environment Court may also exercise the powers of a consent 21 authority under this Division to modify a development consent granted by the Court. 22

Division 4.9 Miscellaneous

Long service leave levy 4.43

A certifier is not to give a completed complying development certificate to the applicant unless any long service levy payable under section 34 of the Building and Construction Industry Long Service Payments Act 1986 in respect of the development (or, where such a levy is payable by instalments, the first instalment of the levy) has been paid.

Maximum penalty: Tier 3.

Determination of Crown development applications 4.44

- (1)A consent authority (other than the Minister) must not:
 - refuse to grant development consent when determining a Crown development (a) application, or
 - (b) impose a condition on development consent when determining a Crown development application, except with the approval of the applicant.
- (2)If the consent authority fails to determine a Crown development application within 37 the period prescribed by the regulations, the applicant may refer the application to the 38 NSW Planning Director-General. 39
- A consent authority for a Crown development application may refer the application (3) 40 to the NSW Planning Director-General at any time. 41
- (4)If a Crown development application is referred to the NSW Planning 42 Director-General, the Minister may exercise the function of the consent authority in 43 determining the application. 44

	(5)	This section applies to applications for modification of development consent for development to be carried out by or on behalf of the Crown in the same way as it applies to Crown development applications.	1 2 3	
	(6)	In this section:	4	
		<i>Crown</i> has the meaning given to that expression by the regulations.	5	
		<i>Crown development application</i> means an application for development consent for development to be carried out by or on behalf of the Crown.	6 7	
		development consent does not include a complying development certificate.	8	
4.45	When species impact statement required			
	(1)	If proposed development is likely to significantly affect threatened species, the application for development consent (or an environmental impact statement accompanying the application) is to include or be accompanied by a species impact statement.	10 11 12 13	
	(2)	This section does not apply to State significant development.	14	
4.46	Laps	ing of development consent	15	
	(1)	A development consent (other than a complying development certificate) lapses 5 years after the date from which it takes effect. Note. For lapsing of complying development certificate see section 4.12.	16 17 18	
	(2)	The period specified by this section is subject to the reductions and changes permitted by Schedule 4.	19 20	

Part	5	Infr	astructure and environmental impact assessment	1
Divis	sion (5.1	Environmental impact assessment (except for State or public priority infrastructure)	2 3
5.1	Defin	itions	s: Division 5.1	4
		In thi	is Division:	5
		appro	oval includes:	6
		(a)	a consent, licence or permission or any form of authorisation, and	7
		(b)	a provision of financial accommodation by a determining authority to another person (not being a provision of any such financial accommodation that is excluded by the regulations from this definition),	8 9 10
		5.2, a	loes not include development consent under Part 4, an approval under Division a declaration of public priority infrastructure or any other approval excluded by egulations.	11 12 13
		autho Mini	<i>mining authority</i> for relevant development means the Minister or public ority by or on whose behalf the development is or is to be carried out or any ster or public authority whose approval is required in order to enable the lopment to be carried out.	14 15 16 17
			in an environmental impact statement, includes being furnished with such a ment.	18 19
5.2	Relev	vant d	evelopment to which this Division applies	20
	(1)	This	Division applies to any development other than the following:	21
		(a)	development for the purposes of public priority infrastructure,	22
		(b)	any act, matter or thing for which development consent or State infrastructure approval is required or has been obtained,	23 24
		(c)	any act, matter or thing that is prohibited under the planning control provisions of a local plan,	25 26
		(d)	exempt development,	27
		(e)	development carried out in compliance with a development control order,	28
		(f)	any project or development for which an approval from the Minister to carry out the project or development was obtained under the former Act,	29 30
		(g)	development that is excluded from this Division by the regulations.	31
	(2)	relev	is Division, the development to which this Division applies is referred to as <i>ant development</i> (but referred to elsewhere in this Act as <i>Part 5 environmental ect assessment development</i>).	32 33 34
5.3	Gene	ral du	ity to consider environmental impact of relevant development	35
	(1)	into a	termining authority is, in its consideration of any relevant development, to take account the matters affecting or likely to affect the environment because of the ring out of that development.	36 37 38
	(2)		obligation under this section is in addition to any obligation of a determining prity under the other provisions of or made under this Act or any other Act.	39 40

5.4 Determining authority to obtain and consider EIS for relevant development likely to significantly affect the environment

- (1) A determining authority is not to carry out any relevant development, or grant an approval for any relevant development, that is likely to significantly affect the environment unless:
 - (a) the determining authority has obtained and considered an environmental impact statement in respect of the development prepared by or on behalf of the proponent in accordance with the regulations, and

1

2

3

4

5

6

7

8

9

10

11

12

25

26

27

28

29

34

35

36

37

42

43

Note. Schedule 2 requires the EIS to be publicly exhibited for at least 28 days.

- (b) the determining authority has considered any submissions (or a summary of submissions) duly made during the period of public exhibition of the environmental impact statement, and
- (c) if the determining authority receives notice from the NSW Planning
 Director-General that the Planning Assessment Commission has been
 requested to provide advice or hold a public hearing in respect of the relevant
 development, the Commission has dealt with the request and the determining
 authority has considered the advice, findings and recommendations of the
 Commission and any advice of the Minister that has been provided to it, and
- (d) if the determining authority receives notice from the NSW Planning
 Director-General that the NSW Planning Director-General has decided to
 undertake an examination of the environmental impact statement, the
 examination has been undertaken and the determining authority has
 considered the findings and recommendations in the report of the examination
 that has been provided to it.

Note. See also Division 6.2 of Part 6 for consultation and concurrence requirements if relevant development is likely to significantly affect threatened species.

- (2) The regulations may declare that any relevant development or class of relevant development is taken to be likely to significantly affect the environment for the purposes of this section.
- (3) The determining authority is required to provide the NSW Planning Director-General 30 with: 31
 - (a) a copy of any environmental impact statement under this Division as soon as practicable after it obtains the statement, and 33
 - (b) as soon as practicable and not less than 21 days before carrying out the relevant development concerned or before granting an approval for the relevant development concerned—a copy of any submissions duly made to it during the public exhibition period of the environmental impact statement.
- (4) The determining authority must, at the time it forwards copies of those submissions to the NSW Planning Director-General, also forward copies of those submissions to the Environment Protection Authority if the development is a scheduled activity under the *Protection of the Environment Operations Act 1997*.
 (4) The determining authority must, at the time it forwards copies of those submissions to the NSW Planning Director-General, also forward copies of those submissions to the Environment Protection Authority if the development is a scheduled activity 40 under the *Protection of the Environment Operations Act 1997*.

5.5 Significant effect on environment requiring EIS includes significant effect on threatened species

- For the purposes of this Division, relevant development is to be regarded as development likely to significantly affect the environment if it is likely to significantly affect threatened species.
 44
 45
 46
- (2) In that case, the environmental impact statement is to include or be accompanied by
 47
 48
- (3) If the likely significant effect on threatened species is the only likely significant
 49 effect on the environment, a species impact statement may be obtained instead of an
 50

environmental impact statement and this Division applies as if references to an 1 environmental impact statement were references to a species impact statement. 2 Decision of determining authority where EIS obtained for relevant development 5.6 3 A determining authority that is not the proponent of relevant development for which (1)4 an environmental impact statement has been obtained and considered under this 5 Division may, if satisfied the development will adversely affect the environment: 6 on the grant of approval for the proposed development, impose such 7 (a)conditions or require such modifications as will in its opinion eliminate or 8 reduce the adverse effect of the development on the environment, or 9 refuse to grant approval for the proposed development. (b) 10 The determining authority is required to give the proponent notice in writing of its 11 reasons for any action taken under this subsection. 12 (2)A determining authority that is the proponent of relevant development for which an 13 environmental impact statement has been obtained and considered under this 14 Division may, if satisfied the development will adversely affect the environment: 15 modify the proposed development so as to eliminate or reduce the adverse (a) 16 effect of the development on the environment, or 17 (b) refrain from carrying out the proposed development. 18 (3)This section has effect despite any other Act or despite any instrument made under 19 this or any other Act. 20 5.7 Public hearing by Planning Assessment Commission 21 If the Planning Assessment Commission has been requested under the planning 22 legislation to provide advice or hold a public hearing in respect of relevant 23 development and has reported to the Minister: 24 the Minister is to consider the advice, findings and recommendations of the 25 (a) Commission and forward to relevant determining authorities a copy of the 26 advice, findings and recommendations and any advice the Minister wishes to 27 give as to whether, in the Minister's opinion: 28 there are no environmental grounds which would preclude the carrying (i) 29 out of the development to which the advice, findings or 30 recommendations relate in accordance with the proponent's proposal, 31 or 32 there are no environmental grounds which would preclude the carrying (ii) 33 out of the development subject to its being modified in the manner 34 specified in the advice, findings or recommendations, or 35 there are no environmental grounds which would preclude the carrying (iii) 36 out of the development subject to the observance of conditions specified 37 in the advice, findings or recommendations, or 38 there are environmental grounds which would preclude the carrying out (iv) 39 of the development, and 40 any public authority or body to which an appeal may be made by or under any (b) 41 Act in relation to the development is, in deciding the appeal, to consider and 42 take into account the advice, findings and recommendations of the Planning 43 Assessment Commission and any such advice given by the Minister. 44

5.8	Exan	ninatio	on of EIS by NSW Planning Director-General	1
	(1)	devel	section does not apply to relevant development (or any aspect of relevant lopment) in respect of which the Planning Assessment Commission has been ested to provide advice or hold a public hearing under the planning legislation.	2 3 4
	(2)	envir Divis	NSW Planning Director-General may undertake an examination of an conmental impact statement obtained by a determining authority under this sion and any submissions made under this Division in connection with the ant development.	5 6 7 8
	(3)	soon recor	NSW Planning Director-General is to forward to the determining authority, as as practicable, a report of the findings of the examination and any nmendations arising from those findings. The NSW Planning Director-General make the report public.	9 10 11 12
	(4)	in rel decid	public authority or body to which an appeal may be made by or under any Act lation to the relevant development that is the subject of the examination is, in ling the appeal, to consider and take into account the report forwarded to the mining authority.	13 14 15 16
5.9	Exer	nption	S	17
		-	Division does not apply to or in respect of the following:	18
		(a)	a modification of any relevant development (whose environmental effect has already been considered) that will reduce or not increase its overall environmental effect,	19 20 21
		(b)	a routine activity (such as the maintenance of infrastructure) if the Minister determines it has a low environmental effect and it is carried out in an approved manner determined by the Minister,	22 23 24
		(c)	any relevant development that has been approved, or is to be carried out, by another determining authority after the environmental impact assessment has been carried out in accordance with this Division.	25 26 27
Divi	sion	5.2	State infrastructure development	28
5.10	Decl	aratio	n of State infrastructure development	29
	(1)		he purposes of this Act, <i>State infrastructure development</i> is development that clared under this section to be State infrastructure development.	30 31
	(2)		planning control provisions of a local plan may declare any specified class of lopment to be State infrastructure development.	32 33
	(3)		Minister may, by Ministerial planning order, declare specified development on fied land to be State infrastructure development.	34 35
	(4)	infra	single proposed development comprises development that is only partly State structure development so declared, the remainder of the development (if it is not significant development):	36 37 38
		(a)	may be carried out without development consent under Part 4, and	39
		(b)	is also declared to be State infrastructure development.	40
		the N	subsection does not apply to so much of the remainder of the development that SW Planning Director-General determines is not sufficiently related to the State structure development.	41 42 43
	(5)		elopment which (but for this subsection) would be both State infrastructure lopment and State significant development is:	44 45

		(a)	State significant development and not State infrastructure development (subject to paragraph (b)), or	1 2
		(b)	State infrastructure development and not State significant development if it is specified development on specified land declared to be State infrastructure development by the Minister under subsection (3).	3 4 5
5.11	State restr	e infras	structure development not subject to Part 4 or to prohibitions or in planning control provisions	6 7
	(1)		infrastructure development does not require development consent under Part 4 carried out.	8 9
		of Sta	Section 1.16 requires the approval of the Minister under this Division for the carrying out ate infrastructure development (unless it is exempt development or public priority tructure).	10 11 12
	(2)	to any plan c	infrastructure development that is approved under this Division is not subject y prohibition or restriction in the planning control or other provisions of a local on the carrying out of the development (despite anything to the contrary in those sions).	13 14 15 16
		Note.	For application of other legislation in relation to State infrastructure development—see on 6.1 of Part 6.	17 18
		For pa	ayment of contributions where the proponent of State infrastructure development is not ic authority—see Part 7.	19 20
5.12	Appl	ication	n for approval of State infrastructure development	21
	(1)		roponent may apply for the approval of the Minister under this Division to carry tate infrastructure development.	22 23
	(2)	The a	pplication is to:	24
		(a)	describe the State infrastructure development, and	25
		(b)	contain any other matter required by the NSW Planning Director-General.	26
	(3)	The a	pplication is to be lodged with the NSW Planning Director-General.	27
5.13	Envi	ronme	ntal assessment requirements for approval	28
	(1)	devel	an application is made for the Minister's approval for State infrastructure opment, the NSW Planning Director-General is to prepare environmental sment requirements in respect of the development.	29 30 31
	(2)	requir	ne purposes of environmental impact assessment, the environmental assessment rements must require an environmental impact statement to be prepared by or half of the proponent in accordance with the regulations.	32 33 34
	(3)	Direc	reparing the environmental assessment requirements, the NSW Planning tor-General is to consult relevant public authorities and have regard to the need ess any key issues raised by those public authorities.	35 36 37
	(4)	assess	NSW Planning Director-General is to notify the proponent of the environmental sment requirements. The NSW Planning Director-General may modify those rements by further notice to the proponent.	38 39 40
5.14	Envi	ronme	ntal impact assessment and public consultation	41
	(1)	impao	roponent is to submit to the NSW Planning Director-General the environmental et statement required under this Division for approval to carry out the State structure development.	42 43 44
	(2)		SW Planning Director-General may require the proponent to submit a revised onmental impact statement to address the matters notified to the proponent.	45 46

	(3)	envir the N Note.	mandatory community participation requirements with respect to the commental impact statement under Part 1 of Schedule 2 are to be undertaken by ISW Planning Director-General. Part 1 of Schedule 2 requires the EIS to be publicly exhibited for at least 28 days and les submissions to be made during the public exhibition period.	1 2 3 4 5
	(4)	The recei	NSW Planning Director-General is to provide copies of submissions duly ved by the NSW Planning Director-General or a report of the issues raised in e submissions to:	6 7 8
		(a)	the proponent, and	9
		(b)	if the State infrastructure development will require an environment protection licence under Chapter 3 of the <i>Protection of the Environment Operations Act</i> 1997—the Environment Protection Authority, and	10 11 12
		(c)	any other public authority the NSW Planning Director-General considers appropriate.	13 14
	(5)		NSW Planning Director-General may require the proponent to submit to the / Planning Director-General:	15 16
		(a)	a response to the issues raised in those submissions, and	17
		(b)	a preferred State infrastructure development report that outlines any proposed changes to the State infrastructure development to minimise its environmental impact or to deal with any other issue raised during the assessment of the application concerned.	18 19 20 21
	(6)	propo Direc	the NSW Planning Director-General considers that significant changes are osed to the nature of the State infrastructure development, the NSW Planning ctor-General may make the preferred State infrastructure development report able to the public.	22 23 24 25
5.15	NSW	Planr	ning Director-General's environmental impact assessment report	26
	(1)	Tha	NSW Planning Director-General is to give a report on the State infrastructure	07
	(1)	devel	lopment to the Minister for the purposes of the Minister's consideration of the cation for approval to carry out the development.	27 28 29
	(1)	devel appli	lopment to the Minister for the purposes of the Minister's consideration of the	28
		devel appli	lopment to the Minister for the purposes of the Minister's consideration of the cation for approval to carry out the development.	28 29
		devel appli The l	lopment to the Minister for the purposes of the Minister's consideration of the cation for approval to carry out the development. NSW Planning Director-General's report is to include: a copy of the proponent's environmental impact statement and any preferred	28 29 30 31
		devel appli The I (a) (b) (c)	lopment to the Minister for the purposes of the Minister's consideration of the location for approval to carry out the development. NSW Planning Director-General's report is to include: a copy of the proponent's environmental impact statement and any preferred State infrastructure development report, and any advice provided by public authorities on the State infrastructure development, and a copy of any report or advice of the Planning Assessment Commission in respect of the State infrastructure development, and	28 29 30 31 32 33
		devel appli The I (a) (b)	lopment to the Minister for the purposes of the Minister's consideration of the location for approval to carry out the development. NSW Planning Director-General's report is to include: a copy of the proponent's environmental impact statement and any preferred State infrastructure development report, and any advice provided by public authorities on the State infrastructure development, and a copy of any report or advice of the Planning Assessment Commission in	28 29 30 31 32 33 34 35
5.16	(2)	devel appli The I (a) (b) (c) (d)	lopment to the Minister for the purposes of the Minister's consideration of the facation for approval to carry out the development. NSW Planning Director-General's report is to include: a copy of the proponent's environmental impact statement and any preferred State infrastructure development report, and any advice provided by public authorities on the State infrastructure development, and a copy of any report or advice of the Planning Assessment Commission in respect of the State infrastructure development, and any environmental impact assessment undertaken by the NSW Planning Director-General or other matter the NSW Planning Director-General	28 29 30 31 32 33 34 35 36 37 38
5.16	(2)	devel appli The I (a) (b) (c) (d)	lopment to the Minister for the purposes of the Minister's consideration of the facation for approval to carry out the development. NSW Planning Director-General's report is to include: a copy of the proponent's environmental impact statement and any preferred State infrastructure development report, and any advice provided by public authorities on the State infrastructure development, and a copy of any report or advice of the Planning Assessment Commission in respect of the State infrastructure development, and any environmental impact assessment undertaken by the NSW Planning Director-General or other matter the NSW Planning Director-General considers appropriate.	28 29 30 31 32 33 34 35 36 37 38 39
5.16	(2) Givir	devel appli The I (a) (b) (c) (d)	lopment to the Minister for the purposes of the Minister's consideration of the facation for approval to carry out the development. NSW Planning Director-General's report is to include: a copy of the proponent's environmental impact statement and any preferred State infrastructure development report, and any advice provided by public authorities on the State infrastructure development, and a copy of any report or advice of the Planning Assessment Commission in respect of the State infrastructure development, and any environmental impact assessment undertaken by the NSW Planning Director-General or other matter the NSW Planning Director-General considers appropriate.	28 29 30 31 32 33 34 35 36 37 38 39 40
5.16	(2) Givir	devel appli The I (a) (b) (c) (d) ng of a If: (a) (b)	 lopment to the Minister for the purposes of the Minister's consideration of the faction for approval to carry out the development. NSW Planning Director-General's report is to include: a copy of the proponent's environmental impact statement and any preferred State infrastructure development report, and any advice provided by public authorities on the State infrastructure development, and a copy of any report or advice of the Planning Assessment Commission in respect of the State infrastructure development, and any environmental impact assessment undertaken by the NSW Planning Director-General or other matter the NSW Planning Director-General considers appropriate. approval by Minister to carry out State infrastructure development the proponent makes an application for the approval of the Minister under this 	28 29 30 31 32 33 34 35 36 37 38 39 40 41 42

- (2)The Minister, when deciding whether or not to approve the carrying out of State infrastructure development, is to consider:
 - (a) the NSW Planning Director-General's report on the State infrastructure development and the reports, advice and recommendations contained in the report, and
 - (b) any advice provided by the Minister having portfolio responsibility for the proponent, and
 - any advice, findings or recommendations of the Planning Assessment (c) Commission following a request for advice or a public hearing in respect of the State infrastructure development.
- State infrastructure development may be approved under this Division with such (3)modifications of the development or on such conditions as the Minister may determine.

5.17 Staged State infrastructure development applications

13 14

22

23

24

25

26

27

28

29

32

33

34

35

36

41

42

1

2

3

4

5

6

7

8

9

10

11

12

- (1)For the purposes of this Division, a staged State infrastructure development 15 application is an application for approval of State infrastructure development under 16 this Division that sets out concept proposals for the proposed development, and for 17 which detailed proposals for separate parts of the development are to be the subject 18 of subsequent applications for approval. The application may set out detailed 19 proposals for the first stage. 20
- (2)If approval is granted under this Division on the determination of a staged State 21 infrastructure development application, the approval does not authorise the carrying out of any part of the State infrastructure development unless:
 - (a) approval is subsequently granted to carry out that part of the development following a further application for approval in respect of that part of the development, or
 - the staged State infrastructure development application also provided the (b) requisite details of that part of the development and approval is granted for that first stage without the need for further approval.
- (3)The terms of an approval granted on the determination of a staged State infrastructure 30 development application are to reflect the operation of subsection (2). 31

5.18 Status of staged State infrastructure development applications and approvals

- (1)The provisions of or made under this or any other Act relating to applications for approval and approvals under this Division apply, except as otherwise provided by or under this or any other Act, to a staged State infrastructure development application and an approval granted on the determination of any such application.
- An approval granted on the determination of a staged State infrastructure (2)37 development application for State infrastructure development does not have any 38 effect to the extent that it is inconsistent with the determination of any further 39 application for approval in respect of that development. 40

5.19 Determinations with respect to development subject to an approved staged State infrastructure development application

- When an approval is granted under this Division on the determination of a staged (1)43 State infrastructure development application, the Minister may make any (or any 44 combination) of the following determinations: 45
 - the Minister may determine the further environmental assessment (a) 46 requirements for approval to carry out the development or any particular stage 47

of the development under this Division (in which case those requirements have effect for the purposes of this Division),

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

22

23

24

25

26

27

28

29

33

34

35

36

37

38

39

40

41

45

46

- (b) the Minister may determine that approval to carry out the development or any particular stage of the development is to be subject to Part 4 of this Act (in which case it ceases to be State infrastructure development and subsection (2) applies),
- (c) the Minister may determine that approval to carry out the development or any particular stage of the development is to be subject to Division 5.1 (in which case it ceases to be State infrastructure development and subsection (3) applies),
- (d) the Minister may determine that no further environmental impact assessment is required for the development or any particular stage of the development (in which case the Minister may, under this Division, approve or disapprove its carrying out without further application, environmental impact assessment or report under this Division).
- (2) If the Minister determines under subsection (1) (b) that approval to carry out the development or any particular stage of the development is to be subject to Part 4, the following applies:
 - (a) the determination of a relevant application for development consent under
 Part 4 is to be generally consistent with the terms of the approval of the staged
 State infrastructure development application,
 - (b) any development consent under Part 4 is to be granted subject to such conditions as the Minister determines,
 - (c) the provisions of Part 4 relating to EIS assessed development do not apply,
 - (d) that stage of the development (or any part of it) is exempt or complying development for the purposes of Part 4 if the Minister so determines,
 - (e) a planning control or other provision of a local plan prohibiting or restricting the carrying out of the development or that stage of the development does not have effect if the Minister so determines.
- (3) If the Minister determines under subsection (1) (c) that approval to carry out the development or any particular stage of the development is to be subject to 31 Division 5.1 of this Part, the following applies: 32
 - (a) the development or that stage of the development becomes development that may be carried out without development consent under Part 4,
 - (b) any further environmental impact assessment is to be undertaken in accordance with any requirements determined by the Minister,
 - (c) a planning control or other provision of a local plan prohibiting or restricting the carrying out of the development or that stage of the development does not have effect if the Minister so determines.
- (4) The Minister may vary or revoke a determination under this section on the application of the proponent.
- (5) When modifying an approval granted on the determination of a staged State
 42 infrastructure development application, the Minister may make, vary or revoke a
 43 determination under this section.

5.20 Modification of Minister's approval

(1) In this section:

		under	ster's approval means an approval to carry out State infrastructure development this Division, and includes an approval granted on the determination of a d State infrastructure development application.	1 2 3
		revok	<i>fication</i> of an approval means changing the terms of the approval, including ting or varying a condition of the approval or imposing additional conditions on opproval.	4 5 6
	(2)	infras requi	proponent may request the Minister to modify the Minister's approval for State structure development. The Minister's approval for a modification is not red if the development as modified will be consistent with the existing approval this Division.	7 8 9 10
	(3)	Direc envir	request for the Minister's approval is to be lodged with the NSW Planning tor-General. The NSW Planning Director-General may notify the proponent of onmental assessment requirements with respect to the proposed modification he proponent must comply with before the matter will be considered by the ster.	11 12 13 14 15
	(4)		Anister may modify the approval (with or without conditions) or disapprove of odification.	16 17
	Valid	ity of a	action under this Division	18
		Minis this I	development that has been approved (or purports to be approved) by the ster under this Division is taken to be State infrastructure development to which Division applies, and to have been such development for the purposes of any cation or other matter under this Division in relation to the development.	19 20 21 22
2	Misc	ellane	ous provisions relating to approvals under this Division	23
	(1)	devel	following documents under this Division in relation to State infrastructure opment are to be made publicly available by the NSW Planning tor-General in accordance with the regulations:	24 25 26
		(a)	applications to carry out State infrastructure development,	27
		(b)	environmental assessment requirements for State infrastructure development,	28
		(c)	environmental impact statements placed on public exhibition and responses provided to the NSW Planning Director-General by the proponent after the end of the public exhibition period,	29 30 31
		(d)	environmental impact assessment reports of the NSW Planning Director-General to the Minister,	32 33
		(e)	any advice or reports received from the Planning Assessment Commission,	34
		(f)	approvals to carry out State infrastructure development given by the Minister,	35
		(g)	any determination made by the Minister in connection with the approval of a staged State infrastructure development application and any modification of any such determination,	36 37 38
		(h)	requests for modifications of approvals given by the Minister and any modifications made by the Minister,	39 40
		(i)	any reasons given to the proponent by the Minister as referred to in subsection (2),	41 42
		(j)	any other matter prescribed by the regulations.	43
	(2)	The N	Minister is to give reasons to the proponent for a decision:	44
		(a)	not to approve State infrastructure development under this Division, or	45
		(b)	to modify the State infrastructure development for which the proponent has sought approval under this Division, or	46 47

5.21

		(c)	not to approve a request by the proponent to modify a State infrastructure approval under this Division.	1 2
	(3)	speci (such	pproval under this Division may be subject to a condition that it lapses on a fied date unless specified action with respect to the approval has been taken as the commencement of work on the development). Any such condition may odified to extend the lapsing period.	3 4 5 6
	(4)		pproval under this Division may be surrendered, subject to and in accordance the regulations, by any person entitled to act on the approval.	7 8
	(5)		ndition of the approval of State infrastructure development under this Division require any one or more of the following:	9 10
		(a)	the surrender of any other planning approval under this Act (or the former Act) relating to the development or the land concerned,	11 12
		(b)	the surrender of any existing or continued use rights under Division 11.4 of Part 11 relating to the development or the land concerned.	13 14
Divi	sion	5.3	Public priority infrastructure	15
5.23	Decl	aratio	n of public priority infrastructure	16
	(1)		Minister may, by Ministerial planning order, declare that particular development blic priority infrastructure for the purposes of this Act.	17 18
	(2)	A de	claration under this section may only be made if:	19
		(a)	the particular development is generally of the kind that is identified in a strategic plan (other than a local plan) or in a growth infrastructure plan as priority infrastructure for the area to which the plan applies, or	20 21 22
		(b)	a Minister with portfolio responsibility for the carrying out of the particular development applies for the declaration and the Minister administering this Act is of the opinion that the development is essential for the economic, environmental or social well-being of the State.	23 24 25 26
	(3)		claration under this section may provide that future variations of a particular in relation to the development are covered by the declaration.	27 28
	(4)	prior prior the c	ingle proposed development comprises development that is only partly public ity infrastructure, the remainder of the development is also declared to be public ity infrastructure. This subsection does not apply to so much of the remainder of development that the NSW Planning Director-General determines is not ciently related to the public priority infrastructure.	29 30 31 32 33
	(5)		Minister is to make publicly available his or her reasons for making a declaration r this section.	34 35
	(6)		eclaration under this section is not affected by any subsequent change to a egic or infrastructure plan on which the declaration was based.	36 37
5.24	Ame	ndmei	nt or revocation of declaration	38
	(1)		Minister may, from time to time, amend or revoke a declaration of public priority structure by further Ministerial planning order.	39 40
	(2)	decla	declaration may be amended or revoked at any time after it is made. However, a tration cannot be revoked during or after the carrying out of development for the poses of the public priority infrastructure concerned.	41 42 43
	(3)		regulations may make provision with respect to the application of the planning lation as a consequence of the amendment or revocation of a declaration.	44 45

5.25	Decl	ared p	public priority infrastructure may be carried out	1
	(1)	with unde	elopment for the purposes of public priority infrastructure may be carried out out any planning approval under this Act and despite any provision of or made er the planning legislation, other than this Division. • For application of other legislation—see Division 6.1 of Part 6.	2 3 4 5
	(2)	or an the 1 contr	Minister may, by an instrument published on the NSW legislation website, make nend the planning control provisions of a local plan so that the local plan reflects land use authorised by this section or to change land use and development rols for land in the vicinity of the public priority infrastructure. Any such ument may zone land for use for the infrastructure.	6 7 8 9 10
	(3)	com	such planning control provisions may be made or amended without the need to ply with any conditions precedent under the planning legislation relating to the ing or amendment of those provisions.	11 12 13
5.26	Proj	ect de	finition report by proponent of public priority infrastructure	14
	(1)		bre the carrying out of development for the purposes of public priority astructure:	15 16
		(a)	the proponent is required to prepare a report on the carrying out of that development (the <i>project definition report</i>), and	17 18
		(b)	the report is to be publicly exhibited for a period of at least 28 days, and	19
		(c)	the report (or revised report) is to be published on the NSW planning portal.	20
	(2)		project definition report is to set out the following:	21
		(a)	a description of the development (including any staging of the carrying out of the development),	22 23
		(b)	the measures that the proponent will take to avoid, minimise or mitigate any adverse impacts of the development,	24 25
		(c)	the monitoring, auditing and reporting that the proponent will undertake in relation to the environmental impacts of the development during the construction and operation stages of the development,	26 27 28
		(d)	any other matter prescribed by the regulations.	29
	(3)) Before publicly exhibiting the project definition report, the proponent is to submit copy of the report to the NSW Planning Director-General and to revise the report to address any matters notified to the proponent by the NSW Planning Director-General and submit a copy of the revised report to the NSW Planning Director-General.		30 31 32 33
	(4)	exhil prop	py of the project definition report, as revised by the proponent following public bition of the report (including to address any further matters notified to the onent by the NSW Planning Director-General), is to be published on the NSW ning portal.	34 35 36 37
5.27	App	licatio	n of provisions of this Act to public priority infrastructure	38
	(1)	relate	following provisions of this Act (and the ancillary Schedules to this Act that e to those provisions) do not apply to development for the purposes of public rity infrastructure:	39 40 41
		(a)	Division 1.3 of Part 1,	42
		(b)	Part 3 (other than provisions relating to the identification of priority infrastructure by strategic plans),	43 44
		(c)	Part 4,	45
		(d)	Divisions 5.1 and 5.2 of this Part,	46

(2)

(e)	Part 6 (other than Division 6.1),	1
(f)	Part 7 (other than provisions relating to the identification of priority infrastructure by growth infrastructure plans),	2 3
(g)	Parts 8 and 9,	4
(h)	Part 10 (other than Division 10.3 and provisions relating to development control orders to the extent they apply to a kind of order that refers specifically to public priority infrastructure).	5 6 7
Note	. See general order, item 14 of Division 1 of Part 2 of Schedule 10.	8
parti	bite subsection (1), the regulations may apply (with or without modification) cular provisions excluded by subsection (1) to development for the purposes of ic priority infrastructure.	9 10 11

Part	6		icurrences, cons rovals	ultation and other legislative	1 2		
Divis	ion (6.1		ther legislation—public priority te infrastructure development and State oment	3 4 5		
6.1	Appli	ication and interpretation: Division 6.1					
	(1)	This l	Division applies to the ca	rrying out of:	7		
		(a)	development for the pur	poses of public priority infrastructure, and	8		
		(b)	State infrastructure deve	elopment approved under this Act, and	g		
		(c)		pment approved under this Act.	10		
		Divisio	Because of the definition of extends to the on-going opment.	of the " <i>carrying out of development</i> " under this Act, this operation of public priority infrastructure or other specified	11 12 13		
	(2)	infras carrie	tructure includes any in	ion, development for the purposes of public priority vestigative or other activities that are required to be the preparation of the project definition report for the	14 15 16 17		
	(3)	For th	e purposes of this Divisi	on:	18		
		(a)	State infrastructure deve approved the carrying or	Plopment is approved under this Act if the Minister has at of the development under Division 5.2 of Part 5, and	19 20		
		(b)		pment is approved under this Act if the Minister has onsent to the carrying out of the development under	21 22 23		
		to be requir	carried out for the purpo	es any investigative or other activities that are required ose of complying with any environmental assessment a connection with an application for approval to carry	24 25 26 27		
	(4)	devel) to an approval or consent to the carrying out of an approval or consent as modified from time to time	28 29 30		
	(5)		e purposes of this Division of authorisation of authorisa	on, <i>approval</i> includes a consent, licence or permission tion.	31 32		
6.2	Approvals that do not apply to public priority infrastructure, State infrastructure development or State significant development				33 34		
	(1)			out in Table 1 to this section are not required for the which this Division applies.	35 36		
	(2)		rdingly, the provisions of ut any such approval do	or made under any Act that would prohibit an activity not apply.	37 38		
		Table	1: Approvals that do n	ot apply	39		
		Coast	al Protection Act 1979	Concurrence of Minister administering Act under Part 3			
		Fishe	ries Management Act 1994	Permit referred to in section 201, 205 or 219			
			age Act 1977	Approval under Subdivision 1 of Division 3 of Part 4 or excavation permit under section 139			

National Parks and Wildlife Act 1974	Aboriginal heritage impact permit under section 90
Native Vegetation Act 2003	Approval referred to in section 12 to clear native vegetation (including approval to clear vegetation on State protected land under repealed provisions continued in force under Schedule 3)
Rural Fires Act 1997	Bush fire safety authority under section 100B
Water Management Act 2000	Approval under Part 3 of Chapter 3 (including water use approval, water management work approval, controlled activity approval or aquifer interference approval)

6.3 Approvals that must be issued consistently for public priority infrastructure, State infrastructure development or State significant development

(1) An approval of the kind set out in Table 2 to this section cannot be refused if the approval is necessary for the carrying out of development to which this Division applies.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

- (2) In the case of development for the purposes of public priority infrastructure, any such approval is to be substantially consistent with the authority conferred by Division 5.3 of Part 5 for the carrying out of the development and with the project definition report published under that Division.
- (3) In the case of State infrastructure development or State significant development, any such approval is to be substantially consistent with an approval or consent under this Act for the carrying out of the development.
- (4) This section does not apply to:
 - (a) an application for the renewal of any such approval or a renewed approval, or
 - (b) an application for a further such approval or a further approval following the expiry or lapsing of any such approval, or
 - (c) in the case of an environment protection licence under the *Protection of the Environment Operations Act 1997*—any period after the first review of the
 18
 19
- (5) A reference in this section to an approval includes a reference to any conditions of the approval. 20
- (6) This section applies to a person, court or tribunal that deals with an objection, appeal or review conferred on a person in relation to an approval in the same way as it applies to the person giving the approval.
 22
 23
 24

Table 2: Approvals that must be issued consistently			
Fisheries Management Act 1994	Aquaculture permit under section 144		
Mine Subsidence Compensation Act 1961	Approval under section 15		
Mining Act 1992	Mining lease under section 63 or 64		
Petroleum (Onshore) Act 1991	Production lease under section 42		
Pipelines Act 1967	Licence under Part 3		
Protection of the Environment Operations Act 1997	Environment protection licence under Chapter 3		
Roads Act 1993	Consent under Division 3 of Part 9		

6.4	Statu	itory d	lirections, orders etc that cannot be made or given	1
	(1)	or int	following directions, orders or notices cannot be made or given so as to prevent terfere with the carrying out of development for the purposes of public priority structure:	2 3 4
		(a)	an interim protection order (within the meaning of the National Parks and Wildlife Act 1974 or the Threatened Species Conservation Act 1995),	5 6
		(b)	an order under Division 1 (Stop work orders) of Part 6A of the <i>National Parks</i> and Wildlife Act 1974, Division 1 (Stop work orders) of Part 7 of the <i>Threatened Species Conservation Act 1995</i> or Division 7 (Stop work orders) of Part 7A of the <i>Fisheries Management Act 1994</i> ,	7 8 9 10
		(c)	a remediation direction under Division 3 (Remediation directions) of Part 6A of the <i>National Parks and Wildlife Act 1974</i> ,	11 12
		(d)	an order under section 124 of the Local Government Act 1993.	13
	(2)	of th	nvironment protection notice cannot be given under Chapter 4 of the <i>Protection</i> <i>e Environment Operations Act 1997</i> so as to prevent the carrying out of lopment for the purposes of public priority infrastructure.	14 15 16
	(3)		rder cannot be made under section 136 of the <i>Heritage Act 1977</i> so as to prevent erfere with the carrying out of any development to which this Division applies.	17 18
	(4)	A dir effec	rection, order or notice made or given in contravention of this section has no t.	19 20
Divi	sion	6.2	Concurrences, consultation and other legislative approvals—general provisions	21 22
6.5	Cons is co	sultation nsent	on with Minister administering threatened species legislation if a Minister authority or determining authority	23 24
	(1)	This	section applies to the following:	25
		(a)	development (not being State significant development or complying development) that requires development consent under Part 4 when a Minister is the consent authority,	26 27 28
		(b)	development that requires environmental impact assessment under Division 5.1 of Part 5 when a Minister is the determining authority.	29 30
	(2)	that of Minis	he purposes of determining the application for that development consent or of environmental impact assessment, the Minister concerned is to consult the ster administering the threatened species legislation if the development is likely mificantly affect threatened species.	31 32 33 34
	(3)	provi any r	consulting, the Minister administering the threatened species legislation is to de the Minister who is the consent authority or the determining authority with ecommendations made by the head of the Public Service agency to which the nistration of that legislation is assigned.	35 36 37 38
6.6			ce of or consultation with responsible agency administering threatened gislation if a Minister is not consent authority or determining authority	39 40
	(1)	This	section applies to the following:	41
		(a)	development (not being State significant development or complying development) that requires development consent under Part 4 when a Minister is not the consent authority,	42 43 44
		(b)	development that requires environmental impact assessment under Division 5.1 of Part 5 when a Minister is not the determining authority.	45 46

(2) The consent authority is not to grant development consent if the development is likely to significantly affect threatened species, unless the consent authority has obtained the concurrence of the head of the Public Service agency to which the administration of the threatened species legislation is assigned.

1

2

3

4

21

22

23

24

25

26

27

28

43

44

- (3) The determining authority is not to carry out the development, or grant an approval to carry out the development, if the development is likely to significantly affect threatened species, unless the determining authority has obtained the concurrence of the head of the Public Service agency to which the administration of the threatened species legislation is assigned.
 5
 5
 6
 7
 8
 9
- (4) However, if the Minister administering the threatened species legislation considers that it is appropriate, that Minister may elect to act in the place of the relevant head of the Public Service agency. That Minister is required, in giving any concurrence, to consult that agency head, to provide the consent authority or the determining authority with any recommendations made by that agency head and to give public notice of any such recommendation that the Minister has not accepted.
- (5) In determining whether to give a concurrence under this section, the relevant head of the Public Service agency or Minister (as the case requires) is to have regard to the following:
 16
 17
 18
 - (a) any species impact statement prepared for the development and submissions 19 made in response to it, 20
 - (b) any assessment report prepared by or on behalf of the proponent,
 - (c) any relevant recovery plan or threat abatement plan,
 - (d) whether the development is likely to reduce the long-term viability of the threatened species, populations or ecological communities in the region,
 - (e) whether the development is likely to accelerate the extinction of the threatened species, populations or ecological communities or place them at risk of extinction,
 - (f) the object of this Act of the promotion of sustainable development.
- (6) A concurrence under this section may be conditional on the taking of action that the relevant head of the Public Service agency or Minister (as the case requires) 30 considers will significantly benefit threatened species and to which the person 31 required to take the action has agreed. Any such action may (without limitation) 32 include the provision of biodiversity offsets. 33
- (8) A consent authority that grants consent, or a determining authority that grants approval, to the carrying out of development for which a concurrence under this section has been granted must grant the consent or approval subject to any conditions of the concurrence. This does not affect the right of the consent authority or determining authority to impose other conditions not inconsistent with the conditions of the concurrence or to refuse consent or approval.

6.7 Other consultation and concurrence requirements under planning control provisions

 This section applies to development (not being State significant development or complying development) that requires development consent under Part 4 when the planning control provisions of a local plan require the consent authority to consult with or to obtain the concurrence of a person before granting consent to the development. (2) The consent authority is required to consult with or obtain the concurrence of that person unless the consent authority determines to refuse to grant development consent.

1

2

3

4

5

6

7

8

9

10

11

12

22

- (3) However, if the consent authority is a Minister, then that Minister is required only to consult with that person.
- (4) A person whose concurrence to development is required may:
 - (a) grant concurrence to the development, either unconditionally or subject to conditions, or
 - (b) refuse concurrence to the development.

In deciding whether to grant concurrence, the person must take into consideration only the matters that the planning control provisions concerned require that person to take into account.

- (6) If the person whose concurrence to development is required fails to inform the consent authority of the decision concerning concurrence within the time allowed by the regulations for doing so, the consent authority may determine the application for development consent without the concurrence of that person.
 (6) If the person whose concurrence to development is required fails to inform the second second

6.8 Consultation and development consent—certain bush fire prone land

- This section applies to development (not being State significant development or complying development) that requires development consent under Part 4, being development for any purpose other than:
 - (a) a subdivision of land that could lawfully be used for residential or rural residential purposes, or 27
 - (b) development for a special fire protection purpose (within the meaning of section 100B of the *Rural Fires Act 1997*). 29
- (2) Development consent cannot be granted for the carrying out of development to which
 30 this section applies on bush fire prone land unless the consent authority:
 31
 - (a) is satisfied that the development conforms to the specifications and requirements prescribed or adopted by the regulations for the purposes of this section, or
 32
 33
 34
 - (b) has been provided with a certificate by a person who is recognised by the NSW
 Rural Fire Service as a qualified consultant in bush fire risk assessment stating
 that the development conforms to the relevant specifications and
 requirements.
- (3) If the consent authority is satisfied that the development does not conform to the relevant specifications and requirements, the consent authority may, despite subsection (2), grant consent to the carrying out of the development but only if it has consulted with the Commissioner of the NSW Rural Fire Service concerning measures to be taken with respect to the development to protect persons, property and the environment from danger that may arise from a bush fire.

(4)	subje	regulations may exclude development from the application of this section ect to compliance with any requirements of the regulations. The regulations may hout limiting the requirements that may be made):	1 2 3
	(a)	require the issue of a certificate by the Commissioner of the NSW Rural Fire Service, or a qualified consultant in bush fire risk assessment, in relation to the bush fire risk of the land concerned, and	4 5 6
	(b)	authorise the payment of a fee for the issue of any such certificate.	7
	ster m	nay amend local plan for purpose of removing consultation, concurrence legislative provisions	8 9
(1)		section applies to:	10
(1)	(a)	any consultation or concurrence requirement imposed by this Division, or	11
	(b)	any consultation or concurrence requirement imposed by planning control provisions referred to in this Division, or	12 13
	(c)	any provision of or made under this Act relating to environmental assessment requirements for State significant development or State infrastructure development, or	14 15 16
	(d)	the statutory provisions that require any approval of the kind set out in Table 3 to Division 6.3, or	17 18
	(e)	any provision of or made under any other Act that relates to the carrying out of development to which Part 4 or Part 5 applies (being a provision declared by the regulations to be a statutory provision to which this section applies).	19 20 21
(2)	deve prov an ei	Minister may, for the purpose of facilitating the carrying out of any particular elopment or any particular class of development, amend the planning control isions of a local plan to provide that a consultation or concurrence requirement, nvironmental assessment requirement or a relevant statutory provision, to which section applies, does not apply to that development.	22 23 24 25 26
(3)	respo conc not a	such amendment requires the approval of the Minister with portfolio onsibility for the matter that is the subject of the relevant consultation or currence requirement or for the relevant statutory provision. This subsection does apply to an amendment of consultation or concurrence requirements referred to absection (1) (b).	27 28 29 30 31
(4)		Minister may, under Part 3, amend the planning control provisions of the vant local plan for any of the following purposes:	32 33
	(a)	to impose additional or alternative environmental assessment requirements or conditions that apply to that development,	34 35
	(b)	to impose biodiversity offset requirements that apply to that development,	36
	(c)	to remove or modify environmental assessment requirements of or made under this Act that apply to that development,	37 38
	(d)	to prescribe matters to be included in an application for planning approval for that development,	39 40
	(e)	to prescribe matters to be taken into account in determining any such application,	41 42
	(f)	to prescribe standard conditions of development consent that are to apply to any consent for that development,	43 44
	(g)	to prescribe any other matter relating to the assessment, determination or carrying out of that development.	45 46

	(5)	If an	y such amendment prescribes standard conditions of a development consent:	1
		(a)	the development consent is taken to be subject to those conditions, whether or not they are set out in the development consent, and	2 3
		(b)	any other conditions of development consent imposed by the consent authority that are inconsistent with the standard conditions, or that deal with the same matter as and are more onerous than the standard conditions, do not have any effect, and	4 5 6 7
		(c)	the standard conditions imposed cannot be modified without the approval of the Minister.	8 9
	(6)	with	Minister may amend the provisions of a local plan for the purposes of this section out compliance with the provisions of the planning legislation relating to the litions precedent to doing so.	10 11 12
	(7)	relev	section has effect despite any provisions of or made under this Act requiring the vant consultation or concurrence and despite the relevant statutory provisions of ade under any other Act.	13 14 15
Divi	sion	6.3	One stop referrals and decisions for other legislative approvals and for concurrences and consultation	16 17
6.10	App	licatio	n and interpretation: Division 6.3	18
	(1)	This (othe	Division applies to development that requires development consent under Part 4 er than State significant development or complying development) and that:	19 20
		(a)	requires any approval of the kind set out in Table 3 to this Division, or	21
		(b)	is subject to any consultation or concurrence requirement imposed by Division 6.2 or by any planning control provision of a local plan referred to in Division 6.2.	22 23 24
		The	regulations may prescribe circumstances in which this Division does not apply.	25
	(2)	In th	is Division:	26
			<i>roval body</i> means the person or body that may, under the relevant Act, grant an oval of the kind set out in Table 3 to this Division.	27 28
		unde	<i>renewal</i> of an approval means, in the case of an environment protection licence or the <i>Protection of the Environment Operations Act 1997</i> , the first review of the face under section 78 of that Act.	29 30 31
			an approval, includes grant or issue an approval.	32
		one s	stop referrals development means development to which this Division applies.	33
6.11	Refe	erral to	and consultation by NSW Planning Director-General	34
	(1)	conse Diree	consent authority is required to refer a copy of any application for development ent for proposed one stop referrals development to the NSW Planning ctor-General (unless the consent authority has determined to refuse to grant lopment consent).	35 36 37 38
	(2)	Direc	regulations may require the provision of information to the NSW Planning ctor-General and the payment of any part of the fee for the application to the V Planning Director-General.	39 40 41
	(3)	Divis relev	NSW Planning Director-General is, when exercising functions under this sion, required to have regard to any advice on the matter that is provided by the vant approval body or other relevant person or body for whom the NSW Planning ctor-General acts under this Division.	42 43 44 45

(4) The NSW Planning Director-General may, when exercising functions under this Division in respect of more than one legislative approval or matter, issue or give combined general terms of approval, concurrence or advice.

1

2

3

4

5

6

7

8

9

6.12 One stop referrals and decisions for other legislative approvals

- If proposed one stop referrals development requires an approval of the kind set out (1)in Table 3 to this Division, the NSW Planning Director-General is to determine whether the approval body should give approval under the relevant Act and, if so, the general terms of any such approval. For that purpose, the NSW Planning Director-General is to act as if he or she were the approval body that may give that approval and is to have regard to any State assessment requirements prescribed by 10 the regulations for the purposes of this section. 11
- (2)If the consent authority grants development consent for the proposed one stop 12 referrals development, the consent is to be consistent with the general terms of 13 approval of the NSW Planning Director-General of which the consent authority has 14 been informed in accordance with the regulations. For that purpose, the consent 15 authority has the power to impose any condition that could be imposed under the 16 relevant Act by the approval body on the giving of the approval under that Act. 17
- (3)The consent authority is not to grant development consent for the proposed one stop 18 referrals development if the NSW Planning Director-General informs the consent 19 authority that general terms of approval will not be determined because he or she has 20 determined that the approval should not be given under the relevant Act. 21
- (4) If the consent authority grants development consent for the proposed one stop 22 referrals development after the NSW Planning Director-General has determined the 23 general terms of approval, the approval body must give the approval under the 24 relevant Act if an application for that approval is made within 3 years of the grant of 25 development consent and the development consent is still in force. The approval is 26 to be substantially consistent with the general terms of approval determined by the 27 NSW Planning Director-General. 28
- (5) This section does not prevent a consent authority from determining an application for 29 development consent where the NSW Planning Director-General fails to make a 30 determination with respect to a request to determine general terms of approval. 31
- (6)This section does not limit the imposition of conditions (including conditions as to 32 security) by the approval body on the giving of an approval that are not inconsistent 33 with the development consent for the one stop referrals development. 34
- This section does not apply to or limit the giving of approval by the approval body (7)35 pursuant to an application for renewal of an approval or the exercise by the approval 36 body of any of its other functions, such as the issuing of orders, the suspension or 37 cancellation of an approval or the prosecution of offences. 38
- (8)However, an approval body must not vary the terms of an approval subject to this 39 section before the expiration, lapsing or first renewal of the approval (whichever first 40 occurs) in a manner that is inconsistent with the relevant general terms of approval. 41 This subsection does not apply to a variation that is consistent with a modification 42 made to the conditions of the development consent. 43
- (9) The regulations may provide that any public or other notice requirements for an 44 application for an approval under a relevant Act are taken to be satisfied by public 45 notice given in relation to the determination of the application for development 46 consent. 47
- (10)This section does not affect any rights of objection, review or appeal of an applicant 48 for approval under a relevant Act, but this section applies to a person, court or 49

tribunal that deals with any such objection, review or appeal in the same way as it applies to the person giving the approval.

- (11) The NSW Planning Director-General is to notify the relevant approval body of a determination made by the NSW Planning Director-General under this section.
- (12) The consent authority is to notify the NSW Planning Director-General and relevant approval bodies of its determination of an application to carry out one stop referrals development.
- (13) This section has effect despite the provisions of or made under any other Act that relates to the giving of an approval to which this section applies.

6.13 One stop referrals and decisions for consultation and concurrences

- If proposed one stop referrals development requires any concurrence imposed by Division 6.2 or by the planning control provisions of a local plan referred to in Division 6.2, the NSW Planning Director-General is, in accordance with this Act or the relevant planning control provisions, to give or refuse to give that concurrence in the place of (and as if he or she were) the person or body that would otherwise give or refuse to give concurrence.
- (2) If the proposed one stop referrals development requires any consultation imposed by Division 6.2 or by the planning control provisions of a local plan referred to in Division 6.2, the NSW Planning Director-General is, in accordance with this Act or the relevant planning control provisions, to give consideration to the matter and provide any relevant advice in the place of (and as if he or she were) the person or body with whom the consent authority was otherwise required to consult.
 17
 18
 19
 20
 21
 22
- (3) The consent authority is to notify the NSW Planning Director-General and relevant persons or bodies on whose behalf the NSW Planning Director-General has acted under this section of his or her determination of an application to carry out one stop referrals development.
 23 24 24 25 25 26
- (4) This section has effect despite any other provisions of, or made under, this Act requiring the relevant consultation or concurrence.

6.14 Special heritage provision

A consent authority is not to refuse development consent on heritage grounds for one stop referrals development if the same development is the subject of an approval under Subdivision 1 of Division 3 of Part 4 of the *Heritage Act 1977*. 32

Table 3: One stop referrals and decisions for legislative approvals								
Fisheries Management Act 1994	Permit referred to in section 144, 201, 205 or 219							
Heritage Act 1977	Approval under Subdivision 1 of Division 3 of Part 4							
Mine Subsidence Compensation Act 1961	Approval under section 15							
Mining Act 1992	Mining lease under section 63 or 64							
National Parks and Wildlife Act 1974	Aboriginal heritage impact permit under section 90							
Petroleum (Onshore) Act 1991	Production lease under section 42							
Protection of the Environment Operations Act 1997	Environment protection licence under Chapter 3							
Roads Act 1993	Consent under Division 3 of Part 9							

29 30

33

27

28

1

2

3

4

5

6

7

8

9

10

Rural Fires Act 1997 Water Management Act 2000 Bush fire safety authority under section 100B

Approval under Part 3 of Chapter 3 (including water use approval, water management work approval, controlled activity approval or aquifer interference approval)

Par	t 7	Infrastructure and other contributions	1
Divi	sion 7	7.1 Introductory	2
7.1	Defin	nitions	3
		In this Part:	4
		<i>biodiversity offset contribution</i> means a contribution imposed under Division 7.4.	5
		growth infrastructure plan means a growth infrastructure plan made under	6
		Division 7.3 for regional infrastructure in a region or subregion of the State.	7
		<i>infrastructure contribution</i> means a local infrastructure contribution or regional infrastructure contribution.	8 9
		<i>local infrastructure</i> means each of the following:	10
		(a) local roads,	11
		(b) local drainage works,	12
		(c) open space,	13
		(d) community facilities.	14
		<i>local infrastructure contribution</i> means a contribution imposed under Division 7.2.	15
		<i>local infrastructure plan</i> means a local infrastructure plan made under Division 7.2 for local infrastructure in a local government area.	16 17
		<i>local plan</i> means the contribution provisions of a local plan.	18
		<i>planning agreement</i> means a planning agreement provided for in Division 7.5.	19
		<i>public authority</i> includes any Minister.	20
		regional infrastructure means each of the following:	21
		(a) regional or State roads,	22
		(b) land for drainage,	23
		(c) transport infrastructure,	24
		(d) regional open space,	25
		(e) educational establishments.	26
		<i>regional infrastructure contribution</i> means a contribution imposed under Division 7.3.	27 28
7.2	Appli	ication of Part	29
	(1)	This Part applies to development that requires development consent under Part 4.	30
		Note. This Part accordingly applies where a consent authority grants development consent, including where a council or other certifier grants a complying development certificate.	31 32
	(2)	This Part also applies to State infrastructure development that requires approval under Part 5 and that is not carried out by or on behalf of a public authority in the same way as it applies to development that requires development consent under Part 4, subject to any necessary modifications and any modifications prescribed by the regulations.	33 34 35 36 37
Divi	sion 7	7.2 Local infrastructure contributions	38
7.3	Loca	I infrastructure contributions for proposed development	39
	(1)	A consent authority can impose a local infrastructure contribution on development to fund the provision by a council of local infrastructure in the area in which development is proposed to be carried out.	40 41 42

	(2)	In this Divis	sion, the <i>provision</i> of local infrastructure includes:	1
			xtension or augmentation of local infrastructure and recoupment of the of providing, extending or augmenting local infrastructure, and	2 3
		of thi and t	sercise of any function of a council in connection with the administration s Division (including the carrying out of any research or investigation he preparation of any report, study or instrument in connection with the nistration of this Division).	4 5 6 7
	(3)		frastructure contribution is imposed by means of a condition of at consent for the development concerned.	8 9
7.4	Cont	ributions to	be in accordance with contribution provisions of local plan	10
	(1)	A local infrarea, that is:	astructure contribution must be in accordance with the local plan for the	11 12
		devel	cal infrastructure contribution can only be imposed in respect of opment or a class of development that the local plan provides can be the ct of a local infrastructure contribution, and	13 14 15
			mount of a local infrastructure contribution is to be determined in dance with the local plan.	16 17
	(2)	that the lo infrastructur	astructure contribution is imposed in respect of the local infrastructure cal infrastructure plan provides is to be the subject of the local re contribution. Accordingly, a local infrastructure contribution cannot unless there is a local infrastructure plan.	18 19 20 21
	(3)	can impose plan for the	sections (1) and (2), if the Minister is the consent authority, the Minister a local infrastructure contribution that is not in accordance with the local area but must have regard to that local plan and the local infrastructure mposing the contribution.	22 23 24 25
7.5	Kind	s of local inf	rastructure contributions—direct and indirect	26
	(1)	There are 2	kinds of local infrastructure contribution, as follows:	27
			<i>ect contribution</i> , which is a contribution requiring the payment of money easonable contribution towards the provision of local infrastructure,	28 29
		amou propo	<i>direct contribution</i> , which is a contribution requiring the payment of an int of money that is a percentage of the capital investment value of the osed development or that is calculated by reference to the area of the osed development or calculated in a manner authorised by the regulations.	30 31 32 33
	(2)	An indirect to a direct c	contribution cannot be imposed in respect of development that is subject ontribution (that is, only one or the other can be imposed).	34 35
	(3)	the subdivision can be improved development	ct contribution has been imposed in respect of development comprising sion of land (the <i>initial development</i>), no direct or indirect contribution osed in respect of other development on that land unless that other at will or is likely to increase the demand for local infrastructure beyond in demand attributable to the initial development.	36 37 38 39 40
	(4)		rposes of a direct contribution, the cost of providing existing local re is that cost as indexed in accordance with the regulations.	41 42
	(5)	or with resp	poses of an indirect contribution, the regulations may make provision for ect to the means by which the capital investment value or the area of the evelopment is to be estimated or determined.	43 44 45

7.6 Nexus for contributions

infras that th	rect contribution for the provision, extension or augmentation of local tructure within an area can only be imposed if the consent authority is satisfied he development concerned will or is likely to require the provision of or increase	2
the de	emand for that local infrastructure.	3 4 5
	rect contribution for recoupment of the cost of providing existing local tructure within the area can be imposed only if:	6 7
(a)	the consent authority is satisfied that the development concerned will, if carried out, benefit from the provision of the existing local infrastructure, and	8 9
(b)	the existing local infrastructure was (at any time, whether before or after the commencement of this Act) provided within the area by a council in preparation for or to facilitate the carrying out of development in the area.	10 11 12
devel	n indirect contribution there is not required to be any connection between the opment the subject of the contribution and the object of expenditure of any y required to be paid.	13 14 15
7.7 Payment of	local infrastructure contributions	16
the ar	al infrastructure contribution is payable to the council (the <i>relevant council</i>) of ea in which the development concerned is to be carried out. Payment to the council is required even if the council is not the consent authority.	17 18 19
(2) The ti plan.	me for payment of a local infrastructure contribution is as required by the local	20 21
works	relevant council can accept the dedication of land or the carrying out of s-in-kind in part or full payment of a local infrastructure contribution. Land ated in satisfaction of a local infrastructure contribution is to be transferred to levant council.	22 23 24 25
under index the co	ocal plan can provide for the indexation of local infrastructure contributions the plan by providing for a contribution to increase at a specified rate of ation between the date the contribution is imposed and the date of payment of ontribution. The amount payable in satisfaction of a contribution is to be ased in accordance with the indexation provisions of the local plan.	26 27 28 29 30
7.8 Use of loca	I infrastructure contributions	31
contri mone	uncil is to hold any money paid to it in satisfaction of a local infrastructure ibution for the purpose for which the contribution was imposed and apply the y towards that purpose. Money paid for different purposes may be pooled and ed progressively for those purposes.	32 33 34 35
	dedicated in satisfaction of a local infrastructure contribution is to be made able by the council for the purpose for which the contribution was imposed.	36 37
additi	noney paid in satisfaction of a local infrastructure contribution includes any onal amount earned from investment of that money and any proceeds of sale of received in satisfaction of a local infrastructure contribution.	38 39 40
mann to be subre	Minister considers that a council is not complying with this section in a timely er, the Minister may direct the council on how the money held by the council is applied to further the local infrastructure requirements of the relevant region or gion and the council is required to comply with any such direction. A direction be consistent with the infrastructure priorities in the regional growth plan or	41 42 43 44 45

7.10

subregional delivery plan and may require money collected in one local government 1 area to be applied in another local government area in the region or subregion. 2 Note. The Planning Administration Act 2013 enables the Minister to delegate to a regional 3 planning panel or subregional planning board the power to give directions to a council under 4 this subsection. 5 (5) Money is to be applied, and land made available, under this section within 5 years for 6 the purposes for which the contribution was imposed. The Minister may, having 7 regard to the relevant local infrastructure plan, extend that period in a particular case 8 at the request of the council. 9 **Complying development** 10 The local plan must specify whether or not a local infrastructure contribution for (1)11 which the plan provides is to be imposed on development when a complying 12 development certificate is issued for the development. 13 A certifier must impose a condition on a complying development certificate (2)14 requiring payment of a local infrastructure contribution if the local plan so requires 15 (with the amount of the contribution to be as required by the local plan). 16 (3) If a certifier fails to impose the necessary condition, such a condition has effect as if 17 it had been imposed by the certifier. 18 (4) A certifier (other than a council) cannot accept the dedication of land or the carrying 19 out of works-in-kind in payment of a local infrastructure contribution. 20 Local infrastructure plans 21 (1)A council may prepare a draft local infrastructure plan for its area and submit the 22 draft plan to the Minister. 23 (2)In preparing a draft local infrastructure plan, the council must have regard to any 24 applicable NSW planning policy. 25 A local infrastructure plan is to identify the local infrastructure for which a local (3)26 infrastructure contribution is to be imposed. The plan may also identify other 27 infrastructure for the area. 28 (4)A local infrastructure plan is to specify when contributions of money or land are to 29 be applied or made available for the purposes for which the contributions were 30 imposed. 31 (5)The Minister may make a local infrastructure plan in the form in which it was 32 submitted or with such modifications as the Minister considers appropriate. The 33 Minister may decide not to make the draft plan. 34 Note. Schedule 2 requires any such draft plan to be publicly exhibited for at least 28 days. 35 In making a local infrastructure plan, the Minister must have regard to any applicable (6)36 NSW planning policy. 37 (7)A local infrastructure plan is required to be published on the NSW planning portal. 38 A local infrastructure plan may be amended or repealed by a further local (8) 39 infrastructure plan published on the NSW planning portal. 40 **Cross-boundary issues** 41 A local infrastructure contribution may be imposed for the benefit (or partly for the (1)42 benefit) of an area that adjoins the local government area in which the development 43 concerned is to be carried out. 44

	(2)		money payable pursuant to such a contribution is to be apportioned among the vant councils:	1 2
		(a)	in accordance with the local plans for the areas concerned, or	3
		(b)	if provision is not made for the apportionment in those local plans—in accordance with the terms of the development consent for the development.	4 5
	(3)	Dire	dispute between the councils concerned is to be referred to the NSW Planning ctor-General and resolved in accordance with any direction given by the NSW ning Director-General.	6 7 8
7.12	Loca	l infra	astructure may be provided outside NSW	9
		the p in w	rect contribution may, with the written approval of the Minister, be imposed for provision of local infrastructure on land in another State or Territory if the area hich the development the subject of the contribution is to be carried out adjoins other State or Territory.	10 11 12 13
7.13	Арре	eals		14
	(1)	as a c Cour circu	ndition of development consent that imposes a local infrastructure contribution direct contribution may be disallowed or amended by the Land and Environment rt on appeal under Part 9 because it is not reasonable in the particular imstances of the case, even if it was in accordance with the local plan and the l infrastructure plan. The Court cannot disallow or amend the local plan.	15 16 17 18 19
	(2)	as a	ndition of development consent that imposes a local infrastructure contribution n indirect contribution that is in accordance with the local plan cannot be lowed or amended by the Land and Environment Court on appeal under Part 9.	20 21 22
Divi	ision	7.3	Regional infrastructure contributions	23
7.14	Impo	sition	n of regional infrastructure contributions	24
	(1)		cal plan can impose a regional infrastructure contribution on development as a ribution towards the provision of regional infrastructure by the State.	25 26
	(2)	In th	is Division, the <i>provision</i> of regional infrastructure includes:	27
		(a)	the extension or augmentation of regional infrastructure and recoupment of the cost of providing, extending or augmenting regional infrastructure, and	28 29
		(b)	any action of a public authority in connection with the exercise of any statutory function under the planning legislation in connection with regional infrastructure, including the carrying out of any research or investigation and the preparation of any report, study or instrument.	30 31 32 33
	(3)	whic calcu	local plan is to specify the development or class of development in respect of ch a regional infrastructure contribution is imposed, the amount (or method of ulation) of the contribution, the time for payment of the contribution and any xation of that amount at the date of payment.	34 35 36 37
	(4)	capit the a regu whic	amount of a regional infrastructure contribution can only be a percentage of the tal investment value of the proposed development or calculated by reference to area of the proposed development or calculated in a manner authorised by the lations. The regulations may make provision for or with respect to the means by the capital investment value or area of proposed development is to be mated or determined.	38 39 40 41 42 43
	(5)	the r	re is not required to be any connection between the development the subject of regional infrastructure contribution and the object of expenditure of any money ired to be paid.	44 45 46

	(6)	The regional infrastructure for which a regional infrastructure contribution can be imposed extends to regional infrastructure outside the region or subregion concerned or outside New South Wales.	1 2 3
	(7)	A regional infrastructure contribution is in addition to any local infrastructure contribution imposed under Division 7.2.	4 5
7.15	How	regional infrastructure contributions are imposed	6
	(1)	The consent authority for development on which a regional infrastructure contribution is imposed is to impose the contribution as a condition of development consent for the development.	7 8 9
	(2)	If a consent authority fails to impose the necessary condition on development consent, such a condition has effect as if it had been imposed by the consent authority.	10 11 12
7.16	Payr	nent of regional infrastructure contributions	13
	(1)	A regional infrastructure contribution (and any proceeds of sale of land accepted in payment of a regional infrastructure contribution) are payable into:	14 15
		(a) the Regional Contributions Fund under Division 7.6, subject to paragraph (b), or	16 17
		(b) in the case of regional infrastructure contributions for land for drainage or land for regional open space—the Planning Growth Fund for the region under Schedule 7.	18 19 20
		Note. Division 7.6 provides for payments from the Regional Contributions Fund for the provision of regional infrastructure identified in a growth infrastructure plan.	21 22
	(2)	The Minister may accept the dedication of land or the carrying out of works-in-kind in part or full payment of a regional infrastructure contribution but only if the land or work relates to the regional infrastructure to which the contribution relates.	23 24 25
	(3)	The Minister may sell all or part of any land received under this Division in respect of a contribution or direct the transfer of any such land to a public authority that is to provide, or has provided, infrastructure in relation to the development or class of development to which the contribution relates.	26 27 28 29
7.17	Com	plying development	30
	(1)	The local plan must specify whether or not a regional infrastructure contribution for which the local plan provides is to be imposed on development when a complying development certificate is issued for the development.	31 32 33
	(2)	A certifier must impose a condition on a complying development certificate requiring payment of a regional infrastructure contribution if the local plan so requires (with the amount of the contribution to be as required by the local plan).	34 35 36
	(3)	If a certifier fails to impose the necessary condition, such a condition has effect as if it had been imposed by the certifier.	37 38
	(4)	A certifier cannot accept the dedication of land or the carrying out of works-in-kind in payment of a regional infrastructure contribution.	39 40
7.18	Plan infra	ning control provisions of local plan for satisfactory arrangements for regional structure	41 42
	(1)	The planning control provisions of a local plan can include provision to the effect that development consent is not to be granted for specified development or development of a specified class unless arrangements satisfactory to the NSW Planning	43 44 45

Director-General have been made for the making of a contribution towards the provision of regional infrastructure by the State in relation to the development.

- (2)In deciding for the purposes of any such provision whether satisfactory arrangements 3 have been made for the making of a contribution towards the provision of regional 4 infrastructure by the State in relation to development, the NSW Planning 5 Director-General must have regard to any applicable NSW planning policy. 6
- (3) If a regional infrastructure contribution is imposed in respect of development as a condition of development consent, a contribution towards the provision of regional infrastructure in respect of the development cannot be required under the planning control provisions of the local plan. 10

7.19 Growth infrastructure plans

1

2

7

8

9

11

- The NSW Planning Director-General may prepare a draft growth infrastructure plan (1)12 for any region or subregion of the State and submit the draft plan to the Minister. 13 Note. Schedule 2 requires any such draft plan to be publicly exhibited for at least 28 days. 14
- (2)In preparing a draft growth infrastructure plan, the NSW Planning Director-General 15 must have regard to any applicable NSW planning policy or regional growth plan. 16
- A growth infrastructure plan is to identify the regional infrastructure for which a (3)17 regional infrastructure contribution may be imposed. The plan may also identify 18 priority infrastructure and other infrastructure for the region or subregion. 19
- (4)A growth infrastructure plan is to include a contestability assessment, being an 20 assessment of the opportunities for infrastructure identified in the plan to be provided 21 and operated by the private sector. 22
- The Minister may make a growth infrastructure plan in the form in which it was (5) 23 submitted or with such modifications as the Minister considers appropriate. The 24 Minister may decide not to make the draft plan. 25
- The making of a growth infrastructure plan requires the concurrence of the Treasurer (6)26 or (if the cost of the infrastructure concerned is less than \$30 million) the Secretary 27 of the Treasury. 28
- A growth infrastructure plan is required to be published on the NSW planning portal. (7)29
- A growth infrastructure plan may be amended or repealed by a further growth (8) 30 infrastructure plan published on the NSW planning portal. 31

7.20 Restrictions on appeals and changes to conditions

- A condition of development consent that imposes a regional infrastructure (1)33 contribution in accordance with the local plan cannot be disallowed or amended by 34 the Land and Environment Court on appeal under Part 9. 35
- (2)A condition of development consent that imposes a regional infrastructure 36 contribution cannot be modified without the approval of the Minister. 37

Division 7.4 Biodiversity offset contributions

38 39

32

7.21 Imposition of biodiversity offset contributions

- (1)A local plan can impose a biodiversity offset contribution on development as a 40 contribution towards biodiversity offsets for the conservation or enhancement of the 41 natural environment of the State. 42
- The local plan is to specify the development or class of development in respect of (2)43 which a biodiversity offset contribution is imposed, the amount (or method of 44

calculation) of the contribution, the time for payment of the contribution and any indexation of that amount at the date of payment.

- (3) There is not required to be any connection between the development the subject of the biodiversity offset contribution and the object of expenditure of any money required to be paid.
- (4) A biodiversity offset contribution can be imposed for the conservation or enhancement of the natural environment of or outside the site of the proposed development.
- (5) A biodiversity offset contribution is in addition to any local or regional infrastructure contribution imposed under this Part.

7.22 How biodiversity offset contributions are imposed

- (1) The consent authority for development on which a biodiversity offset contribution is imposed is to impose the contribution as a condition of development consent for the development.
- (2) If a consent authority fails to impose the necessary condition on development consent, such a condition has effect as if it had been imposed by the consent 16 authority.

7.23 Payment of biodiversity offset contributions

- A biodiversity offset contribution (and any proceeds of sale of land accepted in payment of a biodiversity offset contribution) are payable into:
 - (a) except in the case of biodiversity offsets for the aquatic environment—the Biodiversity Offset Contributions Fund under the *Threatened Species Conservation Act 1995*, or
 - (b) in the case of biodiversity offsets for the aquatic environment—the Fish Conservation Trust Fund under the *Fisheries Management Act 1994*. 25

Note. Under the relevant Acts, other money is payable into those Funds (eg money required to be paid into the Funds under planning agreements) and provision is made for the purposes for which money in the Fund may be used.

- (2) A Minister administering the threatened species legislation may accept the dedication of land, or the carrying out of environmental works or activities that conserve or enhance the natural environment of the State, in part or full payment of a biodiversity offset contribution.
- (3) A Minister administering the threatened species legislation may sell all or part of any land received under this Division in respect of a contribution.
- (4) The Ministers administering the *Threatened Species Conservation Act 1995* and the *Fisheries Management Act 1994* may, for the purposes of this section, determine whether any biodiversity offset is or is not a biodiversity offset for the aquatic are nvironment.
 (4) The Ministers administering the *Threatened Species Conservation Act 1995* and the section *Act 1995* and the section *Act 1994* may, for the purposes of this section, determine are nvironment.

7.24 Complying development

- The local plan must specify whether or not a biodiversity offset contribution for which the plan provides is to be imposed on development when a complying development certificate is issued for the development.
- (2) A certifier must impose a condition on a complying development certificate
 43 requiring payment of a biodiversity offset contribution if the local plan so requires
 44 (with the amount of the contribution to be as required by the local plan).
- (3) If a certifier fails to impose the necessary condition, such a condition has effect as if
 46 it had been imposed by the certifier.
 47

18

21

22

23

26

27

28

33

34

1

2

3

4

5

6

7

8

9

10

11

12

13

14

39

	(4)		ertifier cannot accept the dedication of land or the carrying out of works or ities in payment of a biodiversity offset contribution.	1 2
7.25	Rest	rictior	ns on appeals and changes to conditions	3
	(1)	in ac	ndition of development consent that imposes a biodiversity offset contribution cordance with a local plan cannot be disallowed or amended by the Land and ronment Court on appeal under Part 9.	4 5 6
	(2)	cann	ndition of development consent that imposes a biodiversity offset contribution ot be modified without the approval of a Minister administering the threatened ies legislation.	7 8 9
Divi	sion	7.5	Planning agreements	10
7.26	Deve	loper	s can enter into planning agreements	11
	(1)	a per of co	<i>anning agreement</i> is an agreement between one or more public authorities and rson (the <i>developer</i>) under which the developer is required to dedicate land free ost, pay money, carry out public works or provide any other material public fit (or any combination of them) towards the following:	12 13 14 15
		(a)	the provision of infrastructure that is identified in a local infrastructure plan or growth infrastructure plan,	16 17
		(b)	the provision of infrastructure, or any other public purpose, that is identified in a Ministerial planning order made under this section,	18 19
		(c)	the provision of affordable housing that is identified in a strategic plan,	20
		(d)	the conservation or enhancement of the natural environment of the State.	21
	(2)	The o	developer must be:	22
		(a)	a person who has sought a change to the planning control provisions of a local plan, or	23 24
		(b)	a person who has made, or proposes to make, an application for development consent, or	25 26
		(c)	a person who has entered into an agreement with, or is otherwise associated with, a person to whom paragraph (a) or (b) applies.	27 28
	(3)	In thi	is Division, the <i>provision</i> of infrastructure includes:	29
		(a)	the extension or augmentation of infrastructure and the recoupment of the cost of providing, extending or augmenting infrastructure, and	30 31
		(b)	the funding of recurrent expenditure relating to the provision, extension or augmentation of infrastructure, and	32 33
		(c)	any action of a public authority in connection with the exercise of any statutory function under the planning legislation, including the carrying out of any research or investigation and the preparation of any report, study or instrument.	34 35 36 37
7.27	Plan	ning a	greements can limit contribution requirements	38
	(1)	provi infra	anning agreement can exclude the application in respect of development of any ision of Division 7.2 (Local infrastructure contributions), Division 7.3 (Regional structure contributions) or Division 7.4 (Biodiversity offset contributions), ect to the following restrictions:	39 40 41 42
		(a)	a planning agreement cannot exclude the application of a provision of Division 7.2 in respect of development unless the consent authority for the development or the Minister is a party to the agreement,	43 44 45

		(b)	a public authority is not to enter into a planning agreement excluding the application of Division 7.3 unless the public authority is the Minister or does so with the approval of the Minister,	1 2 3
		(c)	a public authority is not to enter into a planning agreement excluding the application of Division 7.4 unless the public authority is a Minister administering the threatened species legislation or does so with the approval of any such Minister.	4 5 6 7
	(2)	or 7.4 respe	lanning agreement excludes the application of any provision of Division 7.2, 7.3 4 to particular development, a consent authority cannot impose a contribution in ect of that development under the excluded provisions (except in respect of the cation of any part of those provisions that is not excluded by the agreement).	8 9 10 11
7.28	Use o	of plar	nning agreement contributions	12
	(1)	purpo purpo	blic authority is to hold any money paid to it under a planning agreement for the ose for which the payment was required and apply the money towards that ose. Money paid for different purposes may be pooled and applied progressively nose purposes.	13 14 15 16
	(2)	into a	a specified statutory fund, the public authority is to paid to a public authority is to be paid a specified statutory fund, the public authority is to pay the money into that fund apply the money for the purposes of that fund.	17 18 19
	(3)		dedicated under a planning agreement is to be made available by the public prity for the purpose for which the dedication was required.	20 21
	(4)		noney paid under a planning agreement includes any additional amount earned investment of that money.	22 23
7.29	Parti		planning agreements	24
7.29	Partie (1)	es to p Any l be ar	Dianning agreements Minister, public authority or other person approved by the Minister is entitled to a additional party to a planning agreement and to receive a benefit under the ement on behalf of the State.	
7.29		es to p Any be an agree A cou	Minister, public authority or other person approved by the Minister is entitled to a additional party to a planning agreement and to receive a benefit under the	24 25 26
7.29	(1) (2)	es to p Any be an agree A cou count any p	Minister, public authority or other person approved by the Minister is entitled to a additional party to a planning agreement and to receive a benefit under the ement on behalf of the State. uncil is not precluded from entering into a joint planning agreement with another cil or other public authority merely because it applies to any land not within, or	24 25 26 27 28 29
	(1) (2)	Any Design of the second secon	Minister, public authority or other person approved by the Minister is entitled to a additional party to a planning agreement and to receive a benefit under the ement on behalf of the State. uncil is not precluded from entering into a joint planning agreement with another cil or other public authority merely because it applies to any land not within, or burposes not related to, the area of the council.	24 25 26 27 28 29 30
	(1) (2) Limit	es to p Any l be ar agree A cou cound any p cations A pla devel relati of a l A pla anyth plann	Minister, public authority or other person approved by the Minister is entitled to a additional party to a planning agreement and to receive a benefit under the ement on behalf of the State. uncil is not precluded from entering into a joint planning agreement with another cil or other public authority merely because it applies to any land not within, or burposes not related to, the area of the council. S on planning agreements anning agreement cannot impose an obligation on a public authority to grant lopment consent, or to exercise any function under the planning legislation in on to a change to or the making or revocation of the planning control provisions	24 25 26 27 28 29 30 31 32 33 34
	 (1) (2) Limit (1) (2) 	Any I be ar agree A cou cound any p rations A pla devel relati of a 1 A pla anyth plann land	Minister, public authority or other person approved by the Minister is entitled to a additional party to a planning agreement and to receive a benefit under the ement on behalf of the State. uncil is not precluded from entering into a joint planning agreement with another cil or other public authority merely because it applies to any land not within, or burposes not related to, the area of the council. S on planning agreements anning agreement cannot impose an obligation on a public authority to grant lopment consent, or to exercise any function under the planning legislation in on to a change to or the making or revocation of the planning control provisions ocal plan. anning agreement is void to the extent, if any, to which it requires or allows the done that, when done, would breach a provision of this Act, the thing control provisions of a local plan or a development consent applying to the	24 25 26 27 28 29 30 31 32 33 34 35 36 37 38
7.30	 (1) (2) Limit (1) (2) 	Any l be an agree A cou cound any p ations A pla devel relati of a l A pla anyth plann land eents o	Minister, public authority or other person approved by the Minister is entitled to a additional party to a planning agreement and to receive a benefit under the ement on behalf of the State. uncil is not precluded from entering into a joint planning agreement with another cil or other public authority merely because it applies to any land not within, or burposes not related to, the area of the council. S on planning agreements anning agreement cannot impose an obligation on a public authority to grant lopment consent, or to exercise any function under the planning legislation in on to a change to or the making or revocation of the planning control provisions ocal plan. anning agreement is void to the extent, if any, to which it requires or allows ting to be done that, when done, would breach a provision of this Act, the ting control provisions of a local plan or a development consent applying to the concerned.	24 25 26 27 28 29 30 31 32 33 34 35 36 37 38 39
7.30	 (1) (2) Limit (1) (2) Cont 	Any l be an agree A cou cound any p ations A pla devel relati of a l A pla anyth plann land eents o	Minister, public authority or other person approved by the Minister is entitled to a additional party to a planning agreement and to receive a benefit under the ement on behalf of the State. uncil is not precluded from entering into a joint planning agreement with another cil or other public authority merely because it applies to any land not within, or burposes not related to, the area of the council. S on planning agreements anning agreement cannot impose an obligation on a public authority to grant lopment consent, or to exercise any function under the planning legislation in on to a change to or the making or revocation of the planning control provisions ocal plan. anning agreement is void to the extent, if any, to which it requires or allows hing to be done that, when done, would breach a provision of this Act, the ting control provisions of a local plan or a development consent applying to the concerned.	24 25 26 27 28 29 30 31 32 33 34 35 36 37 38 39 40

		(c)	the nature and extent of the contribution to be made by the developer under the agreement, the time or times by which the contribution is to be made and the manner by which the contribution is to be made,	1 2 3
		(d)	in the case of development, whether the agreement excludes (wholly or in part) or does not exclude the application of Divisions 7.2, 7.3 and 7.4 to the development,	4 5 6
		(e)	if the agreement does not exclude the application to the development of provisions of Division 7.2 for imposing a direct contribution, whether benefits under the agreement are or are not to be taken into consideration in connection with imposing such a contribution,	7 8 9 10
		(f)	the terms of an offer referred to in section 7.33 (3) relating to the agreement,	11
		(g)	a mechanism for the resolution of disputes under the agreement,	12
		(h)	the enforcement of the agreement by a suitable means, such as the provision of a bond or guarantee, in the event of a breach of the agreement by the developer.	13 14 15
	(2)	plann	e is not required to be any connection between the development to which a ning agreement applies and the object of expenditure of any money required to id by the agreement.	16 17 18
		Note. for the	See section 7.28, which requires money paid under a planning agreement to be applied e purpose for which it was paid.	19 20
7.32	Regi	stered	planning agreements to run with land	21
	(1)		anning agreement can be registered under this section if the following persons to its registration:	22 23
		(a)	if the agreement relates to land under the <i>Real Property Act 1900</i> —each person who has an estate or interest in the land registered under that Act,	24 25
		(b)	if the agreement relates to land not under the <i>Real Property Act 1900</i> —each person who is seised or possessed of an estate or interest in the land.	26 27
	(2)		odgment of an application for registration in a form approved by the strar-General, the Registrar-General is to register the planning agreement:	28 29
		(a)	by making an entry in the relevant folio of the Register kept under the <i>Real Property Act 1900</i> if the agreement relates to land under that Act, or	30 31
		(b)	by registering the agreement in the General Register of Deeds if the agreement relates to land not under the <i>Real Property Act 1900</i> .	32 33
	(3)	sectio	anning agreement that has been registered by the Registrar-General under this on is binding on, and is enforceable against, the owner of the land from time to as if each owner for the time being had entered into the agreement.	34 35 36
	(4)		ference in this section to a planning agreement includes a reference to any adment or revocation of a planning agreement.	37 38
7.33	Circu	umstar	nces in which planning agreements can or cannot be required to be made	39
	(1)		anning control provision of a local plan (other than a provision of a transitional ting instrument made before 8 July 2005) has no effect to the extent that the sion:	40 41 42
		(a)	expressly requires a planning agreement to be entered into before an application for development consent can be made, considered or determined, or	43 44 45
		(b)	expressly prevents a development consent from being granted or having effect unless or until a planning agreement is entered into.	46 47

	(2)	A consent authority cannot refuse to grant development consent on the ground that a planning agreement has not been entered into in relation to the proposed development or that the developer has not offered to enter into such an agreement.	1 2 3
	(3)	However, a consent authority can require a planning agreement to be entered into as a condition of a development consent, but only if it requires a planning agreement that is in the terms of an offer made by the developer in connection with the application for development consent, or a change to the planning control provisions of the local plan sought by the developer for the purposes of making the application.	4 5 7 8
	(4)	An offer referred to in subsection (3) must include all the matters for which a planning agreement must make provision under this Part.	9 10
	(5)	In this section, <i>planning agreement</i> includes any agreement (however described) containing provisions similar to those contained in a planning agreement.	11 12
7.34	No a	ppeals	13
	(1)	A person cannot appeal to the Land and Environment Court under Part 9 against the failure of a public authority to enter into a planning agreement or against the terms of a planning agreement.	14 15 16
	(2)	This section does not affect the jurisdiction of the Land and Environment Court under Division 10.3 of Part 10 (Civil enforcement proceedings).	17 18
7.35	Proc	edures and other requirements relating to planning agreements	19
	(1)	The Minister may determine:	20
		(a) the procedures to be followed in negotiating a planning agreement, or	21
		(b) the publication of those procedures, or	22
		(c) other standard requirements with respect to planning agreements.	23
		The procedures or other requirements may include consideration of the value of the contribution provided under the planning agreement compared to the value of the contribution that would be required under this Part but for the agreement.	24 25 26
	(2)	A determination made in respect of a particular case is to be made by notice in writing served on the public authority concerned.	27 28
	(3)	A determination made generally or in a particular class of cases is to be made by Ministerial planning order.	29 30
Divi	sion	7.6 Regional Contributions Fund	31
7.36	Defin	nition	32
		In this Division: <i>the Fund</i> means the Regional Contributions Fund established under this Division.	33 34
7.37	Estal	blishment of Fund	35
	(1)	There is to be established in the Special Deposits Account a fund called the Regional Contributions Fund.	36 37
	(2)	The Fund is to be administered by the Secretary of the Treasury. The Secretary is to consult the NSW Planning Director-General in relation to the administration of the Fund.	38 39 40
	(3)	The Secretary of the Treasury is also responsible for investing or making arrangements for investment of money in the Fund.	41 42

7.38 **Payments into Fund**

The following is to be paid into the Fund:

money, and the proceeds of sale of any land, received in payment of a regional (a) 3 infrastructure contribution under Division 7.3 (unless required by that 4 Division to be paid into the Planning Growth Fund for the region established 5 under Schedule 7), 6 7

1

2

8

9

10

11

17

- (b) any money appropriated by Parliament for the purposes of the Fund,
- (c) the proceeds of the investment of money in the Fund,
- (d) any other money required to be paid into the Fund by the planning legislation or any other legislation.

7.39 Payments out of Fund

- The assets of the Fund can be applied for the following purposes (and only those (1)12 purposes): 13
 - payments to public authorities and other persons for the provision of regional (a) 14 infrastructure identified in a growth infrastructure plan to be the subject of a 15 regional infrastructure contribution, 16
 - any money required to meet administrative expenses in relation to the Fund, (b)
 - (c) all other money directed or authorised to be paid from the Fund by the 18 planning legislation. 19
- (2)Priorities for expenditure from the Fund in a region are to be determined in 20 consultation with any relevant subregional planning board. 21

Par	t 8	Bui	Iding and subdivision	1
Divi	sion	8.1	Introductory	2
8.1	Inter	pretati	on: Part 8	3
		In thi	s Part:	4
		build	ing work—see section 1.9.	5
		certif	<i>ier</i> means a building certifier or subdivision certifier.	6
		subdi accrec a cert	See the Dictionary in Schedule 1 for the definitions of building certifier and ivision certifier , which generally include a council or the holder of a certificate of ditation as the relevant certifier under the <i>Building Professionals Act 2005</i> . The holder of ificate of accreditation may act as a subdivision certifier only if the planning control ions of a local plan authorise such an accredited certifier to act.	7 8 9 10 11
		of bu	<i>ge of building use</i> means a change of the use of a building from a use as a class ilding recognised by the <i>Building Code of Australia</i> to a use as a different class ilding recognised by the <i>Building Code of Australia</i> .	12 13 14
			building includes an altered part of, or an extension to, an existing building.	15
			<i>ipal contractor</i> for building work means the person responsible for the overall dination and control of the carrying out of the building work.	16 17
		reside Act 1	ential building work, owner-builder, contractor licence—see Home Building 989.	18 19
			vision work—see section 1.9.	20
			The Dictionary in Schedule 1 includes a complying development certificate in the ion of <i>development consent</i> for the purposes of the Act.	21 22
8.2	Kind	s of ce	ertificates under Part 8	23
		There	e are the following kinds of certificates under this Part:	24
		(a)	<i>construction certificate</i> —a certificate to the effect that building work completed in accordance with specified plans and specifications or standards will comply with the requirements of the regulations.	25 26 27
		(b)	<i>subdivision works certificate</i> —a certificate to the effect that subdivision work completed in accordance with specified plans and specifications will comply with the requirements of the regulations.	28 29 30
		(c)	occupation certificate—a certificate that authorises:	31
			(i) the occupation and use of a new building in accordance with a development consent, or	32 33
			(ii) a change of building use for an existing building in accordance with a development consent.	34 35
			When issued, an occupation certificate is taken to be part of the development consent to which it relates.	36 37
		(d)	<i>subdivision certificate</i> —a certificate that authorises the registration of a plan of subdivision under Part 23 of the <i>Conveyancing Act 1919</i> .	38 39
			When issued, a subdivision certificate is taken to be part of the development consent that authorised the carrying out of the subdivision.	40 41
			Note. Section 195A of the <i>Conveyancing Act</i> 1919 requires a person to lodge a subdivision certificate when lodging a plan of subdivision for registration under that Act.	42 43
		(e)	<i>compliance certificate</i> —a certificate to the effect that:	44
			(i) specified building work or subdivision work has been completed as specified in the certificate and complies with specified plans and	45 46

				specifications or with specified standards or requirements (a <i>completion of work compliance certificate</i>), or	1 2
				Note. A completion of work compliance certificate may be an authorised alternative in certain cases to an occupation certificate under section 8.7 (2) (a) (ii).	3 4 5
			(ii)	a condition with respect to specified building work or subdivision work (being a condition attached to a planning approval) has been complied with, or	6 7 8
			(iii)	a specified building or proposed building has a specified classification identified in accordance with the <i>Building Code of Australia</i> , or	9 10
			(iv)	any specified aspect of development (including design of development) complies with standards or requirements specified in the certificate with respect to the development, or	11 12 13
			(v)	any specified aspect of development complies with the requirements of provisions prescribed by the regulations.	14 15
		under develo requir	Part 4 opment red for b	plying development certificate is a form of development consent that is issued that authorises the carrying out of complying development. Unlike other consents, construction certificates or subdivision works certificates are not building or subdivision work authorised by a development consent in the form of development certificate.	16 17 18 19 20
8.3	Func	tions	of buil	ding certifiers and subdivision certifiers	21
	(1)	A bui	ilding c	certifier has the following functions:	22
		(a)	issuin	g construction certificates for building work,	23
		(b)	carryi	ing out inspections of building work,	24
		(c)	issuin	g occupation certificates,	25
		(d)	issuin	g completion of work compliance certificates,	26
		(e)	ensur	ing building manuals are provided to the owners of buildings,	27
		(f)		ther function conferred or imposed on a building certifier under this or ther Act.	28 29
	(2)	A sub	odivisio	on certifier has the following functions:	30
		(a)	issuin	g subdivision works certificates for subdivision work,	31
		(b)	carryi	ing out inspections of subdivision work,	32
		(c)	issuin	g subdivision certificates,	33
		(d)		ther function conferred or imposed on a subdivision certifier under this y other Act.	34 35
	(3)	in an	y case i	certifier or subdivision certifier must not issue a certificate under this Part in which this Part provides that the certificate is not to be issued. Denalty: Tier 3.	36 37 38
	(4)	The I subdi	NSW P ivision	lanning Director-General may provide guidance to building certifiers or certifiers on the exercise of their functions.	39 40
Divi	sion	8.2	Buil	ding work and certificates relating to building	41
8.4	Requ	iireme	nts be	fore building work commences	42
	(1)			boses of this section, the building certifier for building work is:	43
	. /	(a)	if the	e development consent for the work is a complying development icate—the building certifier who issued the certificate, or	44 45

	(b)	by (o	e case of any other development consent—a building certifier appointed or with the approval of) the person having the benefit of the development ent or other person authorised by the regulations.	1 2 3
(2)			ing requirements apply before the commencement of building work in with a development consent:	4 5
	(a)	com	building certifier has, no later than 2 days before the building work nences, notified the consent authority and the council (if the council is not onsent authority) of his or her appointment as the building certifier,	6 7 8
	(b)	com	building certifier has, no later than 2 days before the building work nences, notified the person having the benefit of the development consent y inspections that are required to be carried out in respect of the building	9 10 11 12
	(c)		erson carrying out the building work has notified the building certifier he person will carry out the building work as an owner-builder, if that is ase,	13 14 15
	(d)		erson having the benefit of the development consent, if not carrying out vork as an owner-builder, has:	16 17
		(i)	appointed a principal contractor for the building work who must be the holder of a contractor licence if any residential building work is involved, and	18 19 20
		(ii)	notified the building certifier of the appointment, and	21
		(iii)	unless that person is the principal contractor, notified the principal contractor of any inspections that are required to be carried out in respect of the building work,	22 23 24
	(e)	2 day	erson having the benefit of the development consent has given at least vs' notice to the council, and the building certifier if not the council, of the on's intention to commence the erection of the building,	25 26 27
	(f)	any o	other requirements of the regulations have been complied with.	28
(3)	A per section		nust not fail to give a notice that the person is required to give under this	29 30
	Maxi	mum j	penalty: Tier 3.	31
(4)	conse build	ent do ing wo	poses of subsection (1), the person having the benefit of a development es not include any contractor or other person who will carry out the ork unless the contractor or other person is the owner of the land on which to be carried out.	32 33 34 35
(5)			n does not apply to Crown building work that is certified under this Part with the <i>Building Code of Australia</i> .	36 37
Requ	ireme	nt for	construction certificate	38
(1)			ion certificate is required for the erection of a building in accordance with ent consent.	39 40
(2)	How	ever, a	construction certificate is not required for the following:	41
	(a)		erection of a building in accordance with a complying development ficate,	42 43
	(b)	Build	In building work that is certified under this Part to comply with the ling Code of Australia.	44 45
	Note. certifi	Sectio cate tha	n 1.18 makes it an offence if a person does building work without a construction at is required by this section.	46 47

8.6	Restriction on issue of construction certificate						
	(1)	A construction certificate must not be issued with respect to the plans and specifications for any building work unless:					
		(a)	the r	equirements of the regulations have been complied with, and	4		
		(b)	Cons	long service levy payable under section 34 of the <i>Building and</i> struction Industry Long Service Payments Act 1986 (or, where such a levy yable by instalments, the first instalment of the levy) has been paid.	5 6 7		
	(2)	it rel		tion certificate has no effect if it is issued after the building work to which physically commenced on the land to which the relevant development plies.	8 9 10		
8.7	Req	uireme	ent for	occupation certificate	11		
	(1)	An o	ccupat	tion certificate is required for:	12		
		(a)	the c builc	commencement of the occupation or use of the whole or any part of a new ling, or	13 14		
		(b)	exist	commencement of a change of building use for the whole or any part of an ting building.	15 16		
		Note. occup	Section of	on 1.18 makes it an offence if a person engages in any activity without an certificate that is required by this section.	17 18		
	(2)	How	ever, a	an occupation certificate is not required:	19		
		(a)		he commencement of the occupation or use of a new building:	20		
			(i)	for any purpose if the erection of the building is or forms part of exempt development or development that does not otherwise require development consent, or	21 22 23		
			(ii)	that is the subject of a completion of work compliance certificate in circumstances in which that certificate is an authorised alternative to an occupation certificate (such as a swimming pool or altered part of an existing building), or	24 25 26 27		
			(iii)	by such persons or in such circumstances as may be prescribed by the regulations, or	28 29		
			(iv)	that has been erected by or on behalf of the Crown or by or on behalf of a person prescribed by the regulations, or	30 31		
		(b)		he commencement of a change of building use for the whole or any part n existing building:	32 33		
			(i)	if the change of building use is or forms part of exempt development or development that does not otherwise require development consent, or	34 35		
			(ii)	by such persons or in such circumstances as may be prescribed by the regulations, or	36 37		
			(iii)	if the existing building has been erected by or on behalf of the Crown or by or on behalf of a person prescribed by the regulations.	38 39		
8.8	Rest	riction	i s on i	issue of occupation certificates	40		
	(1)			tion certificate must not be issued unless any preconditions to the issue of ate that are specified in a development consent have been complied with.	41 42		
	(2)			tion certificate must not be issued to authorise a person to commence or use of a new building (or part of a new building) unless:	43 44		
		(a)	a de	velopment consent is in force with respect to the building (or part of the ling), and	45 46		

		(b)	in the case of a building erected pursuant to a development consent (other than a complying development certificate), a construction certificate has been issued with respect to the plans and specifications for the building (or part of the building), and	1 2 3 4
		(c)	the completed building (or part of the building) is suitable for occupation or use in accordance with its classification under the <i>Building Code of Australia</i> , and	5 6 7
		(d)	such other requirements as are required by the regulations to be complied with before such a certificate may be issued have been complied with.	8 9
	(3)	new	ccupation certificate must not be issued to authorise a person to commence a use of a building (or of part of a building) resulting from a change of building or an existing building unless:	10 11 12
		(a)	a development consent is in force with respect to the change of building use, and	13 14
		(b)	the building (or part of the building) is suitable for occupation or use in accordance with its classification under the <i>Building Code of Australia</i> , and	15 16
		(c)	such other requirements as are required by the regulations to be complied with before such a certificate may be issued have been complied with.	17 18
8.9	Effec	ct of o	ccupation certificate on earlier occupation certificates	19
	(1)		ccupation certificate for a building revokes any earlier occupation certificate for building.	20 21
	(2)		occupation certificate for a part of a building revokes any earlier occupation ficate to the extent to which it applies to that part.	22 23
Divi	ision	8.3	Subdivision work and certificates relating to subdivision	24
Divi 8.10			Subdivision work and certificates relating to subdivision ents before subdivision work commences	24 25
		For t subdi	-	
	Requ	For t subdi benet regul The f	ents before subdivision work commences he purposes of this section, the subdivision certifier for subdivision work is a ivision certifier appointed by (or with the approval of) the person having the fit of the development consent for the work or other person authorised by the	25 26 27 28
	Req ((1)	For t subdi benet regul The f	ents before subdivision work commences he purposes of this section, the subdivision certifier for subdivision work is a ivision certifier appointed by (or with the approval of) the person having the fit of the development consent for the work or other person authorised by the ations. Following requirements apply before the commencement of subdivision work in	25 26 27 28 29 30
	Req ((1)	For t subdi benef regul The f accor	ents before subdivision work commences the purposes of this section, the subdivision certifier for subdivision work is a ivision certifier appointed by (or with the approval of) the person having the fit of the development consent for the work or other person authorised by the ations. Following requirements apply before the commencement of subdivision work in redance with a development consent: the subdivision certifier has, no later than 2 days before the subdivision work commences, notified the consent authority and the council (if the council is not	25 26 27 28 29 30 31 32 33
	Req ((1)	For t subdi benet regul The f accon (a)	 Ants before subdivision work commences The purposes of this section, the subdivision certifier for subdivision work is a invision certifier appointed by (or with the approval of) the person having the fit of the development consent for the work or other person authorised by the ations. Collowing requirements apply before the commencement of subdivision work in redance with a development consent: The subdivision certifier has, no later than 2 days before the subdivision work commences, notified the consent authority and the council (if the council is not the consent authority) of his or her appointment as the subdivision certifier, the subdivision certifier has, no later than 2 days before the subdivision work commences, notified the person having the benefit of the development consent of any inspections that are required to be carried out in respect of the 	25 26 27 28 29 30 31 32 33 34 35 36 37
	Req ((1)	For t subdi benef regul The f accor (a) (b) (c)	ents before subdivision work commences the purposes of this section, the subdivision certifier for subdivision work is a tivision certifier appointed by (or with the approval of) the person having the fit of the development consent for the work or other person authorised by the ations. Following requirements apply before the commencement of subdivision work in redance with a development consent: the subdivision certifier has, no later than 2 days before the subdivision work commences, notified the consent authority and the council (if the council is not the consent authority) of his or her appointment as the subdivision certifier, the subdivision certifier has, no later than 2 days before the subdivision work commences, notified the person having the benefit of the development consent of any inspections that are required to be carried out in respect of the subdivision work, the person having the benefit of the development consent has given at least 2 days' notice to the council, and the subdivision work. rson must not fail to give a notice that the person is required to give under this	25 26 27 28 29 30 31 32 33 34 35 36 37 38 39 40
	Requ (1) (2)	For t subdi benet regul The f accor (a) (b) (c) A per section	ents before subdivision work commences the purposes of this section, the subdivision certifier for subdivision work is a tivision certifier appointed by (or with the approval of) the person having the fit of the development consent for the work or other person authorised by the ations. Following requirements apply before the commencement of subdivision work in redance with a development consent: the subdivision certifier has, no later than 2 days before the subdivision work commences, notified the consent authority and the council (if the council is not the consent authority) of his or her appointment as the subdivision certifier, the subdivision certifier has, no later than 2 days before the subdivision work commences, notified the person having the benefit of the development consent of any inspections that are required to be carried out in respect of the subdivision work, the person having the benefit of the development consent has given at least 2 days' notice to the council, and the subdivision work. rson must not fail to give a notice that the person is required to give under this	25 26 27 28 29 30 31 32 33 34 35 36 37 38 39 40 41 42

			ivision work unless the contractor or other person is the owner of the land on h the work is to be carried out.	1 2
	(5)		section does not apply to Crown building work that is certified under this Part mply with the <i>Building Code of Australia</i> .	3 4
8.11	Requ	uireme	ent for subdivision works certificate	5
	(1)		bdivision works certificate is required for the carrying out of subdivision work cordance with a development consent.	6 7
	(2)	How	ever, a subdivision works certificate is not required for the following:	8
		(a)	subdivision work carried out in accordance with a complying development certificate,	9 10
		(b)	Crown building work that comprises subdivision work and that is certified under this Part to comply with the <i>Building Code of Australia</i> .	11 12
		Note. works	Section 1.18 makes it an offence if a person does subdivision work without a subdivision scertificate that is required by this section.	13 14
8.12	Rest	riction	on issue of subdivision works certificate	15
	(1)		bdivision works certificate must not be issued with respect to the plans and fications for any subdivision work unless:	16 17
		(a)	the requirements of the regulations have been complied with, and	18
		(b)	any long service levy payable under section 34 of the <i>Building and</i> <i>Construction Industry Long Service Payments Act 1986</i> (or, where such a levy is payable by instalments, the first instalment of the levy) has been paid.	19 20 21
	(2)	to w	bdivision works certificate has no effect if it is issued after the subdivision work hich it relates is physically commenced on the land to which the relevant lopment consent applies.	22 23 24
8.13	Rest	riction	is on issue of subdivision certificates	25
	(1)	A sul	bdivision certificate must not be issued for a subdivision unless:	26
		(a)	the subdivision is not prohibited by or under this Act, and	27
		(b)	in the case of subdivision that cannot be carried out except with development consent, a development consent is in force with respect to the subdivision, and	28 29
		(c)	in the case of subdivision for which a development consent has been granted, all the conditions of the development consent that, by its terms, are required to be complied with before a subdivision certificate may be issued in relation to the plan of subdivision have been complied with, and	30 31 32 33
		(d)	in the case of subdivision of land to which a planning agreement referred to in Part 7 applies, all the requirements of the agreement that, by its terms, are required to be complied with before a subdivision certificate is issued in relation to the plan of subdivision have been complied with, and	34 35 36 37
		(e)	in the case of subdivision for which the operation of the development consent has been deferred under Part 4, the applicant has satisfied the consent authority concerning all matters as to which the consent authority must be satisfied before the development consent can operate, and	38 39 40 41
		(f)	 in the case of subdivision the subject of a development consent for which the consent authority is required by or under this Act to notify any objector: (i) at least 28 days have elapsed since the objector was notified, or (ii) if an appeal has been made by the objector within that time, the appeal has been finally determined. 	42 43 44 45 46

	(2)		hout limiting subsection (1), a subdivision certificate must not be issued for a livision that involves subdivision work unless:	1
		(a)	the work has been completed, or	3
		(b)	agreement has been reached between the applicant for the certificate and the consent authority:	4 5
			(i) as to the payment by the applicant to the consent authority of the cost of carrying out the work, and	6 7
			(ii) as to when the work will be completed by the consent authority, or	8
		(c)	agreement has been reached between the applicant for the certificate and the consent authority:	9 10
			(i) as to the security to be given by the applicant to the consent authority with respect to the work to be completed, and	11 12
			(ii) as to when the work will be completed by the applicant,	13
			such other requirements as are required by the regulations to be complied with ore such a certificate may be issued have been complied with.	14 15
	(3)	land	section (2) does not prevent the issue of a subdivision certificate for part only of that may be subdivided in accordance with a development consent as long as the irements of that subsection have been complied with in relation to that part.	16 17 18
Divi	sion	8.4	Compliance certificates	19
3.14	Requ	uireme	ent for compliance certificate	20
	(1)		ompliance certificate is required in relation to building work or subdivision work ach circumstances as are prescribed by the regulations.	21 22
		fire sa certifi obtair	2 1. For example, the regulations require compliance certificates to be obtained for certain safety aspects of development before a complying development certificate, construction ficate or occupation certificate can be issued and require compliance certificates to be ined for certain alternative solutions to the BCA before a complying development ficate can be issued. See also section 8.7 (2) (a) (ii).	23 24 25 26 27
		Note exten	2. Section 4.23 (3) provides that a condition of development consent has no effect to the nt it requires a compliance certificate to be obtained in respect of any development.	28 29
	(2)	build	ertifier may obtain a compliance certificate from another person in relation to ding work or subdivision work for which the certifier is responsible even if a pliance certificate is not required.	30 31 32
3.15	Pers	ons w	vho may issue compliance certificates	33
		A co	ompliance certificate may be issued by:	34
		(a)	a certifier, or	35
			Note. A certifier includes a council.	36
		(b)	a person of a class prescribed by the regulations as being authorised to issue a compliance certificate in relation to the matters to be certified.	37 38
3.16	Rest	riction	n on issue of compliance certificates	39
		for b	regulations may prevent the issue of particular kinds of compliance certificates building work or subdivision work unless a planning approval is in force with ect to the building or subdivision to which the work relates.	40 41 42

8.15

Division 8.5		8.5	Liability for defective building or subdivision work			
8.17	Definitions		3	2		
		build comp build civil subd of a	is Division: ding work includes the design or inspection of building work and the issue of a plying development certificate or a certificate under this Part in respect of ding work. action includes a counter-claim. livision work includes the design or inspection of subdivision work and the issue complying development certificate or a certificate under this Part in respect of livision work.	3 4 5 6 7 8 9 10		
8.18	Limi brou		on time when action for defective building or subdivision work may be	11 12		
	(1)	builc	vil action for loss or damage arising out of or in connection with defective ding work or defective subdivision work cannot be brought more than:	13 14		
		(a)	in the case of residential building work—6 years, or	15		
		(b)	in any other case—10 years,	16		
	$\langle 0 \rangle$		the date of completion of the work.	17		
	(2)	Build (a)	ding work is taken to be completed on: the date on which an occupation certificate is issued that authorises the occupation of the building or part of the building for which the work was carried out (or if an occupation certificate is not required, the date on which a compliance certificate is issued for the completed building work), or	18 19 20 21 22		
		(b)	if no such certificate has been issued—the date on which a required inspection of the completed building work was carried out by a building certifier, or	23 24		
		(c)	if no such certificate has been issued and no such inspection carried out—the date on which the building or part of the building for which the work was carried out is first occupied or used.	25 26 27		
	(3)	Subc	division work is taken to be completed on:	28		
		(a)	if the work was completed before the issue of a subdivision certificate in respect of the subdivision for which the work was carried out—the date on which that certificate is issued, or	29 30 31		
		(b)	if the work was completed after the issue of that certificate—the date on which a compliance certificate is issued that certifies the work has been completed.	32 33		
	(4)		section has effect despite any other Act or law, but does not operate to extend period of limitation under the <i>Limitation Act 1969</i> .	34 35		
8.19	Divis	sion n	ot to affect rights to recover damages for death or personal injury	36		
			ning in this Division applies to or affects any right to recover damages for death ersonal injury arising out of or concerning defective building work or subdivision <i>k</i> .	37 38 39		
Divi	sion	8.6	Miscellaneous	40		
8.20	Own	ers bu	uilding manual	41		
	(1)		ertifier is not to issue an occupation certificate for a building that is of a class cribed by the regulations unless a building manual for the building has been	42 43		

prepared and provided to the owner of the building in accordance with the 1 requirements of the regulations. 2 (2)The regulations may make provision for or with respect to building manuals and, in 3 particular, for or with respect to the following: 4 (a) the preparation, form and maintenance of building manuals, 5 (b) the content of a building manual (including requirements that a building 6 manual identify in a consolidated format matters for on-going compliance in 7 relation to the building concerned), 8 (c) the inspection of building manuals, 9 (d) extending the circumstances in which a building manual is required to be 10 prepared and provided under this section. 11 Crown building, demolition and incidental work 12 (1)In this section: 13 *Crown* has the meaning given to that expression by the regulations. 14 *Crown building work* means development (other than exempt development), or 15 Part 5 environmental impact assessment development, by the Crown that comprises: 16 the erection of a building, or (a) 17 the demolition of a building or work, or (b) 18 the doing of anything that is incidental to the erection of a building or the (c) 19 demolition of a building or work. 20 (2)Crown building work cannot be commenced unless the Crown building work is 21 certified by or on behalf of the Crown to comply with the Building Code of Australia 22 in force as at: 23 (a) the date of the invitation for tenders to carry out the Crown building work, or 24 in the absence of tenders, the date on which the Crown building work (b) 25 commences, except as provided by this section. 26 (3)A Minister may at any time, by Ministerial planning order, determine in relation to 27 buildings generally or a specified building or buildings of a specified class that a 28 specified provision of the Building Code of Australia: 29 (a) does not apply, or 30 does apply, but with such exceptions and modifications as may be specified. (b) 31 The determination has effect according to its tenor. 32 (4)A determination of a Minister applies only to: 33 a building erected on behalf of the Minister, or (a) 34 a building erected by or on behalf of a person appointed, constituted or (b) 35 regulated by or under an Act administered by the Minister. 36 (5)The application of this section is subject to the regulations. 37 Certifiers may be satisfied as to certain matters 38 For the purpose of enabling a certificate under this Part (or a complying development (1)39 certificate) to be issued, the regulations may provide that any requirement for a 40 consent authority or council to be satisfied as to any specified matter is taken to have 41 been complied with if the person or body issuing the certificate is satisfied as to that 42 matter. 43

	(2)	This section applies whether the requirement is imposed by or under this Act, the regulations or a local plan or the terms of a development consent.	1 2						
8.23	Satis	Satisfaction as to compliance with conditions precedent to the issue of certificates							
	(1)	A person who exercises functions under the planning legislation in reliance on a certificate under this Part or complying development certificate is entitled to assume:							
		(a) that the certificate has been duly issued, and	6						
		(b) that all conditions precedent to the issuing of the certificate have been duly complied with, and	7 8						
		(c) that all things that are stated in the certificate as existing or having been done do exist or have been done,	9 10						
			11 12						
	(2)		13 14						
8.24	Directions by certifiers								
	(1)	which this section applies in respect of the aspect of development, the certifier must issue (or, if the certifier is a council, may issue) a notice in writing to the person	16 17 18 19						
			20 21						
			22 23						
	(2)	If a certifier gives a direction under this section and the direction is not complied with within the time specified in the notice containing the direction, the certifier who issued the direction (if not the consent authority) is, within the period prescribed by the regulations, to send a copy of the notice to the consent authority and to notify the consent authority of the fact that the direction has not been complied with.							
	(3)	The regulations may make provision for or with respect to the following:	29						
		(a) non-compliances to which this section applies,	30						
		(b) the procedure for issuing notices under this section,	31						
		(c) requirements in relation to follow-up action,	32						
		(d) the keeping of records in relation to notices given and follow-up action taken,	33						
			34 35						
	(4)	In this section:	36						
			37 38						

Division 9.1		Reviews and appeals					
		9.1	Introductory				
9.1	Interp	pretation: Part 9					
		In thi	s Part:	4			
		apped	al means an appeal to the Court under Divisions 9.3, 9.4, 9.5 and 9.6.	5			
		Cour	t means the Land and Environment Court.	6			
		revie	<i>w</i> means a review by a consent authority under Division 9.2.	7			
Divis	sion §	9.2	Reviews	8			
9.2	Deter	erminations and decisions subject to review					
	(1)		following determinations or decisions of a consent authority under Part 4 are act to review under this Division:	10 11			
		(a)	the determination of an application for development consent by a council, by an independent hearing and assessment panel, by a regional planning panel or by the Planning Assessment Commission or other person acting as delegate of the Minister,	12 13 14 15			
		(b)	the determination of an application for the modification of a development consent by a council, by an independent hearing and assessment panel, by a regional planning panel or by the Planning Assessment Commission or other person acting as delegate of the Minister,	16 17 18 19			
		(c)	the decision of a council to reject and not determine an application for development consent.	20 21			
	(2)		ever, a determination or decision in connection with an application relating to oblowing is not subject to review under this Division:	22 23			
		(a)	a complying development certificate,	24			
		(b)	EIS assessed development,	25			
		(c)	development that is subject to one stop referrals and decisions under Division 6.3 of Part 6,	26 27			
		(d)	Crown development (referred to in section 4.44),	28			
		(e)	a determination or decision that was made after a public hearing into the matter by the Planning Assessment Commission.	29 30			
	(3)		termination or decision reviewed under this Division is not subject to further w under this Division.	31 32			
9.3	Appli	catior	n for and conduct of review	33			
	(1)	deter	pplicant for development consent may request a consent authority to review a mination or decision made by the consent authority. The consent authority is to w the determination or decision if duly requested to do so under this Division.	34 35 36			
	(2)	A det	termination or decision cannot be reviewed under this Division:	37			
		(a)	after the period within which any appeal may be made to the Court has expired if no appeal was made, or	38 39			
		(b)	after the Court has disposed of an appeal against the determination or decision.	40			
	(3)	subje	questing a review, the applicant may amend the proposed development the ect of the original application for development consent or for modification of lopment consent. The consent authority may review the matter having regard to	41 42 43			

the amended development, but only if it is satisfied that it is substantially the same development.

1

2

3

4

5

6

7

8

9

10

11

12

13

18

21

22

23

24

25

26

27

28

29

30

31

32

33

34

35

36

37

- (4) The review of a determination or decision made by a delegate of a council is to be conducted by the council or by another delegate of the council who is not subordinate to the delegate who made the determination or decision. The review of a determination or decision made by the council is to be conducted by the council and not by a delegate.
- (5) The review of a determination or decision made by an independent hearing and assessment panel acting for or as the delegate of the council is also to be conducted by the panel.
- (6) The review of a determination or decision made by the Planning Assessment Commission as the delegate of the Minister is also to be conducted by the Commission.
- (7) The review of a determination or decision made by a delegate of the Minister (other than the Planning Assessment Commission) is to be conducted by the Planning Assessment Commission or by another delegate of the Minister who is not subordinate to the delegate who made the determination or decision.

9.4 Outcome of review

After conducting its review of a determination or decision, the consent authority may confirm or change the determination or decision.

9.5 Miscellaneous provisions relating to reviews

- (1) The regulations may make provision for or with respect to reviews under this Division, including:
 - (a) specifying the person or body with whom applications for reviews are to be lodged and by whom applications for reviews and the results of reviews are to be notified, and
 - (b) setting the period within which reviews must be finalised, and
 - (c) declaring that a failure to finalise a review within that time is taken to be a confirmation of the determination or decision subject to review.

Note. Division 11.2 of Part 11 enables regulations to prescribe the fee for a request for a review.

- (2) The functions of a consent authority in relation to a matter subject to review under this Division are the same as the functions in connection with the original application or determination.
- (3) If a decision to reject an application for development consent is changed on review, the application is taken to have been lodged on the date the decision is made on the review.
- (4) If a determination is changed on review, the changed determination replaces the earlier determination on the date the decision made on the review is registered on the NSW planning portal.
 38
 39
 40
- (5) Notice of a decision on a review to grant or vary development consent is to specify the date from which the consent (or the consent as varied) operates.
 41
 42
- (6) A decision after the conduct of a review is taken for all purposes to be the decision
 43 of the consent authority.
 44
- (7) If on a review of a determination the consent authority grants development consent or varies the conditions of a development consent, the consent authority is entitled (with the consent of the applicant and without prejudice to costs) to have an appeal
 45
 46
 47

			ast the determination made by the applicant to the Court under this Part drawn at any time prior to the determination of that appeal.	1 2
Divis	sion	9.3	Appeals—development consents	3
9.6	Decis	sions	subject to appeal to Court under this Division	4
	(1)	devel	excision of a consent authority under Part 4 in relation to an application for lopment consent or a development consent is (if this Division so provides) ext to appeal to the Court under this Division.	5 6 7
	(2)		ecision subject to appeal includes a decision made after a review under sion 9.2.	8 9
	(3)	There	e is no right of appeal under this Division against the following decisions:	10
		(a)	a decision of a consent authority under this Act in relation to the carrying out of any development that is made after a public hearing by the Planning Assessment Commission into the carrying out of that development,	11 12 13
		(b)	the determination of, or a failure to determine, an application for a complying development certificate,	14 15
		(c)	a decision of a council to issue or not to issue a variation certificate for complying development.	16 17
		Note. of dev	See sections 7.13 and 7.20 for restrictions on the power of the Court to vary conditions velopment consent relating to infrastructure contributions.	18 19
9.7	Appe	al by	applicant—applications for development consent	20
	(1)	the a	pplicant for development consent who is dissatisfied with the determination of application by the consent authority may appeal to the Court against the mination.	21 22 23
	(2)		he purposes of this section, the determination of an application by a consent ority includes:	24 25
		(a)	any decision subsequently made by the consent authority or other person about an aspect of the development that under the conditions of development consent was required to be carried out to the satisfaction of the consent authority or other person, or	26 27 28 29
		(b)	any decision subsequently made by the consent authority as to a matter of which the consent authority must be satisfied before a deferred commencement consent can operate.	30 31 32
	(3)	carry this I	ppeal under this section relating to an application for development consent to out EIS assessed development in respect of which an objector may appeal under Division cannot be heard until after the expiration of the period within which the etor may appeal to the Court.	33 34 35 36
9.8	Appe	al by	an objector—EIS assessed development applications	37
	(1)	for E signit	section applies to the determination of an application for development consent EIS assessed development (including any such development that is State ficant development), being a determination to grant development consent, either nditionally or subject to conditions.	38 39 40 41
	(2)	exhit dissa	erson who duly made a submission by way of objection during the public bition of the application for development consent (an <i>objector</i>) and who is tisfied with the determination of the consent authority to grant consent may al to the Court against the determination.	42 43 44 45

9.9	Арре	al by	applicant—modifications of development consent	1
		the d	pplicant for the modification of a development consent who is dissatisfied with etermination of the application by the consent authority may appeal to the Court ast the determination.	2 3 4
9.10	Time	withi	n which appeals may be made	5
	(1)	6 mo	ppeal under this Division (except by an objector) may only be made within nths after the date the decision appealed against is notified or registered on the <i>l</i> planning portal or after the date of deemed refusal under section 9.11.	6 7 8
	(2)		ppeal under this Division by an objector may only be made within 28 days after ate the objector is notified of the decision appealed against.	9 10
9.11	Circu right:		nces in which consent taken to have been refused for purposes of appeal	11 12
	(1)	(or for the re Divis	nsent authority that has not determined an application for development consent or the modification of a development consent) within the period prescribed by egulations for the determination of the application is, for the purpose only of this sion, taken to have determined the application by refusing development consent o modify development consent) when that period ends.	13 14 15 16 17
	(2)		ection (1) does not prevent a consent authority from determining an application the end of that period.	18 19
	(3)	deter	such determination of an application does not affect the continuation or mination of an appeal made under this Division against the deemed refusal of ent (or modification of consent) under subsection (1).	20 21 22
	(4)	conse entitl	y such determination of an application results in the grant of development ent (or the modification of development consent), the consent authority is led, with the consent of the applicant and without prejudice to costs, to have the al withdrawn at any time prior to the determination of the appeal.	23 24 25 26
9.12	Notic	e of a	ppeals to be given and right to be heard	27
	(1)	The f	following are entitled to be given notice of an appeal made under this Division:	28
		(a)	an objector, in the case of an appeal by an applicant concerning an application for development consent in respect of which the objector has a right of appeal under this Division,	29 30 31
		(b)	an applicant for development consent and the consent authority, in the case of an appeal under this Division by an objector concerning the application for development consent,	32 33 34
		(c)	a Minister or public authority, in the case of an appeal concerning an application for development consent in respect of which the concurrence of the Minister or public authority is required under this Act,	35 36 37
		(d)	an approval body for a legislative approval to which one stop referrals and decisions apply under Division 6.3 of Part 6, in the case of an application for development consent that involves the approval body.	38 39 40
	(2)	Any	such notice of appeal is to be given by the relevant consent authority.	41
	(3)	28 da	one who is given any such notice of appeal is, on application to the Court within ays after the notice is given, entitled to be heard at the hearing of the appeal if lready a party to the proceedings.	42 43 44
	(4)		is section, a reference to an application for development consent includes an cation to modify a development consent.	45 46

Effect of appeals on operation of consents 9.13 1 If the granting of a development consent for development (other than State 2 (1)significant development) is the subject of an appeal made under this Division, the 3 consent ceases to have effect. 4 (2)A development consent that is granted, or confirmed, as a result of a decision on an 5 appeal under this Division is taken to be a development consent duly granted under 6 Part 4. Any such development consent takes effect, subject to any order of the Court, 7 on and from the date the decision is registered on the NSW planning portal (except 8 in the case of the confirmation of a development consent for State significant 9 development). 10 (3) If the effect of a decision on appeal is that development consent is refused, any 11 development consent granted ceases to have any effect. 12 (4) Despite anything to the contrary in this section, a development consent is taken to 13 have effect on and from the date fixed by: 14 a court (whether or not the Land and Environment Court) that finally (a) 15 determines an appeal on a question of law which confirms the validity of, or 16 results in the granting of, the development consent, or 17 the Land and Environment Court, if the validity of a development consent (b) 18 granted by that Court is confirmed by, or the development consent is granted 19 by that Court as a result of, such a final determination made by another court 20 that has not fixed that date. 21 Powers of Court on appeals 9.14 22 In addition to any other functions and discretions that the Court has apart from this (1)23 subsection, the Court has, for the purposes of hearing and disposing of an appeal 24 under this Division, all the functions and discretions which the consent authority 25 whose decision is the subject of the appeal had in respect of the matter the subject of 26 the appeal. 27 (2)The decision of the Court on an appeal under this Division is, for the purposes of this 28 or any other Act or instrument, taken to be the final decision of that consent authority 29 and is to be given effect to accordingly. 30 If the consent authority was under this Act required to consult or obtain the (3)31 concurrence of another person or body before making the decision the subject of an 32 appeal under this Division: 33 the Court may determine the appeal whether or not the consultation has taken (a) 34 place and whether or not the concurrence has been granted, and 35 in a case where the concurrence has been granted-the Court may vary or (b) 36 revoke any conditions imposed by that person or body or may impose any 37 conditions that could have been imposed by that person or body. 38 (4) If an appeal under this Division relates to one stop referrals development under 39 Division 6.3 of Part 6: 40 the Court may determine the appeal whether or not the consent authority has (a) 41 obtained general terms of approval from the NSW Planning Director-General, 42 and 43 (b) the Court is not bound to refuse an application for development consent 44 because the NSW Planning Director-General has decided that general terms of 45 approval will not be determined or a relevant approval body has decided not 46 to grant a relevant approval, and 47

		(c) the Court may determine an appeal even though a development consent granted as a result of the appeal is inconsistent with the general terms of approval determined by the NSW Planning Director-General.	1 2 3
9.15	Misce	ellaneous provisions relating to appeals under this Division	4
	(1)	Separate appeals under this Division with respect to the determination of an application for development consent are, as far as practicable, to be heard together.	5 6
	(2)	On an appeal under this Division, the Court may, at any time on the application of a person or of its own motion, order the joinder of a person as a party to the appeal if the Court is of the opinion:	7 8 9
		 (a) that the person is able to raise an issue that should be considered in relation to the appeal but would not be likely to be sufficiently addressed if the person were not joined as a party, or 	10 11 12
		(b) that:	13
		(i) it is in the interests of justice, or	14
		(ii) it is in the public interest,	15
		that the person be joined as a party to the appeal.	16
	(3)	If the Court on an appeal by an applicant under this Division allows the applicant to file an amended application for development consent (other than to make a minor amendment), the Court must make an order for the payment by the applicant of those costs of the consent authority that have been thrown away as a result of the amendment of the application for development consent. This subsection does not apply to proceedings to which section 34AA of the <i>Land and Environment Court Act 1979</i> applies.	17 18 19 20 21 22 23
	(4)	If the determination or decision appealed against under this Division was made by a regional planning panel or an independent hearing and assessment panel, the council for the area concerned is to be the respondent to the appeal but is subject to the control and direction of the panel in connection with the conduct of the appeal. The council is to give notice of the appeal to the panel.	24 25 26 27 28
	(5)	If the Minister exercised the functions of the council as consent authority (for Crown development) in respect of a determination or decision appealed against under this Division, the council is to be the respondent to the appeal but is subject to the control and direction of the Minister in connection with the conduct of the appeal. The council is to give notice of the appeal to the Minister.	29 30 31 32 33
Divi	sion §	9.4 Appeals—Part 8 certificates	34
9.16	Appe	eals against failure or refusal to issue Part 8 certificates	35
	(1)	An appeal may be made to the Court against the following decisions of a council:	36
		(a) a decision to refuse to issue a construction certificate, occupation certificate, subdivision works certificate or subdivision certificate,	37 38
		(b) a decision to issue any such certificate subject to conditions.	39
	(2)	The appeal may be made by the applicant for the certificate concerned.	40
	(3)	An appeal may only be made within 6 months after the date on which the decision was made.	41 42

9.17	Deer	ned re	efusal for purposes of appeal	1
	(1)	made	the purposes only of an appeal under this Division, a council is taken to have e a decision to refuse to issue a certificate (<i>a deemed refusal</i>) if it has failed to the certificate to the applicant within the period prescribed by the regulations.	2 3 4
	(2)	cons subd	ning in subsection (1) prevents a council from determining an application for a truction certificate, occupation certificate, subdivision works certificate or livision certificate after the expiration of the applicable period specified in that ection.	5 6 7 8
	(3)	the c	etermination made after the expiration of that applicable period does not affect continuance or determination of an appeal made under this Division in respect of emed refusal.	9 10 11
	(4) If a determination is made after the applicable period to grant the certific concerned, the council is entitled, with the consent of the applicant and with prejudice to costs, to have any appeal under this Division against a deemed refu withdrawn at any time prior to the determination of that appeal.			
Divi	ision	9.5	Appeals—development control orders	16
9.18	App	eals c	oncerning orders	17
	(1)		erson who is given a development control order may appeal to the Court against order.	18 19
	(2)		vever, a person may not appeal against a fire safety order given by an authorised officer (other than an order that prevents a person using or entering premises).	20 21
	(3)	The	appeal may only be made:	22
		(a)	within 28 days after the development control order is given to the person, or	23
		(b)	if an order is given subsequently that forms part of the development control order, within 28 days after the subsequent order is given to the person.	24 25
	(4)	On h	nearing an appeal, the Court may:	26
		(a)	revoke the development control order, or	27
		(b)	modify the development control order, or	28
		(c)	substitute for the development control order any other order that the relevant enforcement authority who gave the order could have given, or	29 30
		(d)	find that the development control order is sufficiently complied with, or	31
		(e)	make such order with respect to compliance with the development control order as the Court thinks fit, or	32 33
		(f)	make such other order with respect to the development control order as the Court thinks fit.	34 35
9.19	Awa	rding	of compensation concerning orders	36
	(1)	comp expe any i	Court, on the hearing of an appeal or otherwise, has a discretion to award pensation to a person to whom a development control order is given for any onse incurred by the person as a consequence of the order, including the cost of investigative work or reinstatement carried out by the person as a consequence e order.	37 38 39 40 41
	(2)	the C	pensation is to be awarded only if the person seeking the compensation satisfies Court that the giving of the development control order was unsubstantiated or the s of the order were unreasonable.	42 43 44

	(3)	the C	aim for compensation cannot be made more than 28 days after the date on which Court gives its decision on the appeal or more than 3 months after the date of the lopment control order if an appeal is not made against the order.	1 2 3
	(4)		pensation under this section is to be awarded against the relevant enforcement prity who gave the development control order.	4 5
9.20	Effe	ct of a	ppeal on order	6
			appeal is duly made to the Court against a development control order, the appeal not effect a stay of the order.	7 8
Divi	ision	9.6	Appeals—miscellaneous	9
9.21	Арр	eal co	ncerning decisions on security for development requirements or damage	10
	(1)		section applies in connection with a decision of a consent authority or council ing to security of the kind referred to in section 4.27.	11 12
	(2)	havir	applicant for development consent to which the security relates, or a person ng the benefit of the consent, who is dissatisfied with the decision may appeal to Court as follows:	13 14 15
		(a)	an appeal may be made against a decision of the consent authority with respect to the provision of the security (otherwise than by the imposition of a condition of development consent),	16 17 18
		(b)	an appeal may be made against the failure or refusal of the consent authority to release a security held by it,	19 20
		(c)	an appeal may be made against the failure or refusal of a council to release a security held by it that has been provided in accordance with a condition of a complying development certificate.	21 22 23
		Note. exclu	The right to appeal against the imposition of a condition of development consent is ded because this Part provides separately for appeals against any such condition.	24 25
	(3)		ppeal under subsection (2) (a) may only be made within 6 months after the cant for development consent received notice of the decision.	26 27
	(4)	An a	ppeal under subsection (2) (b) or (c) may only be made:	28
		(a)	except as provided by paragraph (b), within 6 months after the work to which the security relates has been completed, or	29 30
		(b)	if the security is provided in respect of contingencies that may arise on or after completion of the work to which the security relates, not earlier than 3 months and not later than 6 months after the completion of the work.	31 32 33
9.22	Арр	eals ag	gainst refusal to extend consent lapsing period	34
	(1)		section applies to an application under clause 4.1 of Schedule 4 for the extension e period after which a development consent lapses.	35 36
	(2)	appli	applicant for the extension who is dissatisfied with the determination of the cation or the failure of the consent authority to determine the application within eriod prescribed by the regulations, may appeal to the Court.	37 38 39
	(3)	giver	appeal can only be made within 6 months after the date on which the person is a notice of the decision appealed against or the end of the deemed refusal period red to in subsection (2).	40 41 42
9.23	Арр	eals co	oncerning compliance cost notices	43
	(1)		rson on whom a compliance cost notice is served may appeal against the notice e Court within 28 days after the service of the notice on the person.	44 45

(2)	If an appeal is lodged against an order in relation to which a compliance cost notice has been issued:				
	(a)	an appeal may be lodged against the compliance cost notice in the same way as, and at the same time as, the appeal against the development control order concerned, and	3 4 5		
	(b)	the Court may deal with the appeal against the compliance cost notice at the same time as it deals with the appeal against the development control order.	6 7		
(3)	On h	earing an appeal against a compliance cost notice, the Court may:	8		
	(a)	revoke the notice, or	9		
	(b)	modify the notice, or	10		
	(c)	make any other order with respect to the notice as the Court thinks fit.	11		
Арре	als wi	ith respect to building information certificates	12		
(1)	An aj	pplicant:	13		
	(a)	who is dissatisfied with a council's refusal to issue a building information certificate, or	14 15		
	(b)	who is dissatisfied with a council's failure to issue a building information certificate within the period prescribed by the regulations, or	16 17		
	(c)	who is dissatisfied with a notice from the council to supply information in connection with the application,	18 19		
	may	appeal to the Court.	20		
(2)	given	appeal can only be made within 6 months after the date on which the person is a notice of the decision appealed against or the end of the deemed refusal period red to in subsection (1).	21 22 23		
(3)	On h	earing the appeal, the Court may do any one or more of the following:	24		
	(a)	direct the council to issue a building information certificate in such terms and on such conditions as the Court thinks fit,	25 26		
	(b)	revoke, alter or confirm a notice to supply information,	27		
	(c)	make any other order that it considers appropriate.	28		

Par	t 10	Enforcement	1
Divi	sion 1	10.1 Ministerial enforcement powers	2
10.1	Direc	tions by the Minister	3
	(1)	The Minister may direct a public authority or a person or body having functions under the planning legislation (including under a local plan) to exercise those functions at or within such times as are specified in the direction.	4 5 6
	(2)	The Minister may direct a council to provide the Minister, in the manner and at the times specified in the direction, with reports, containing such information as the Minister may direct, on the council's performance in relation to planning and development matters.	7 8 9 10
	(3)	A public authority, person, body or council to whom a direction is given under this section must comply, and is authorised by this section to comply, with the direction in accordance with the terms of the direction.	11 12 13
	(4)	A direction under this section may be given:	14
		(a) if it is given to a particular body or person—by written notice, or	15
		(b) if it is given to a class of bodies or persons—by Ministerial planning order.	16
	(5)	Before giving a direction under this section, the Minister is to consult with the responsible Minister concerned.	17 18
	(6)	Before giving a direction under subsection (2), the Minister is also to consult with the Local Government and Shires Association of New South Wales and any other industry organisation the Minister considers to be relevant, in relation to the information that the Minister is proposing to seek.	19 20 21 22
	(7)	A failure to comply with a direction under this section in relation to the exercise of a function does not invalidate the decision once it is made.	23 24
10.2	Loca	I government investigation into council's performance	25
	(1)	The NSW Planning Director-General may request that an investigation be authorised under section 430 of the <i>Local Government Act 1993</i> into any aspect of a council's performance of its functions under the planning legislation that the NSW Planning Director-General considers requires investigation.	26 27 28 29
	(2)	The NSW Planning Director-General is to be provided with advice on the outcome of any such investigation from the person who authorised the investigation.	30 31
	(3)	The NSW Planning Director-General may report and make recommendations to the Minister following the receipt of that advice.	32 33
10.3	Settle	ement of planning disputes	34
	(1)	In this section, a <i>planning dispute</i> is a dispute about the operation of, or the exercise of functions under, the planning legislation.	35 36
	(2)	If a planning dispute arises between the NSW Planning Director-General and a public authority (other than a council), a party to the dispute may submit that dispute to the Premier for settlement in accordance with this section.	37 38 39
	(3)	If a planning dispute arises between a public authority (other than a council) and another public authority (other than a council), a party to the dispute may submit that dispute to the Premier for settlement in accordance with this section.	40 41 42

	(4)	publi	planning dispute arises between the NSW Planning Director-General or other c authority and a council, a party to the dispute may submit that dispute to the ster for settlement in accordance with this section.	1 2 3
	(5)	sectio Com respe	the submission of a planning dispute to the Premier or the Minister under this on, the Premier or Minister may appoint a member of the Planning Assessment mission to hold an inquiry and make a report to the Premier or the Minister with ext to that dispute or may himself or herself arrange for an inquiry to be held with ext to the dispute.	4 5 6 7 8
	(6)	Mini	the completion of the inquiry and the report of the inquiry, the Premier or the ster may make such order with respect to the planning dispute, having regard to ublic interest and to the circumstances of the case, as the Premier or the Minister as fit.	9 10 11 12
	(7)		order made by the Premier or the Minister under this section may direct the nent of any costs or expenses of or incidental to the holding of the inquiry.	13 14
	(8)	A part of the	rty to a planning dispute is to comply (and is authorised to comply) with an order e Premier or Minister under this section.	15 16
	(9)		provisions of any other Act relating to the settlement of disputes that are asistent with this section do not apply to the settlement of a planning dispute.	17 18
Divi	sion	10.2	Development control orders	19
10.4	Orde	ers tha	t may be given	20
	(1)	The o	development control orders that may be given under this Act are as follows:	21
		(a)	general orders in accordance with the table to Division 1 of Part 2 of Schedule 10,	22 23
		(b)	fire safety orders in accordance with the table to Division 2 of Part 2 of Schedule 10,	24 25
		(c)	brothel closure orders in accordance with the table to Division 3 of Part 2 of Schedule 10.	26 27
	(2)	The 1	regulations may amend those tables.	28
10.5	Rele	vant e	nforcement authorities who may give orders	29
	(1)		elopment control orders may be given by the following (a <i>relevant enforcement ority</i>):	30 31
		(a)	the Minister or the NSW Planning Director-General, but only in connection with public priority infrastructure, State infrastructure development, State significant development or any other development for which the Minister or NSW Planning Director-General is or has been the consent authority,	32 33 34 35
		(b)	a council,	36
		(c)	a consent authority (not being a council or a certifier), but only in connection with development for which the authority is or has been the consent authority,	37 38
		(d)	in the case of fire safety orders (and without limiting the authority of other persons or bodies to give those orders)—the Commissioner of Fire and Rescue NSW or a member of staff of Fire and Rescue NSW, or a member of a permanent fire brigade, who is for the time being authorised by the Minister administering the <i>Fire Brigades Act 1989</i> to give fire safety orders (an <i>authorised fire officer</i>),	39 40 41 42 43 44
		(e)	in the case of brothel closure orders (and without limiting the authority of other persons or bodies to give those orders)—a person or body exercising planning	45 46

		or regulatory functions in respect of the area in which the premises are situated and authorised by the Minister to give brothel closure orders.	1 2
	(2)	A development control order in connection with State infrastructure development or public priority infrastructure may be given only by the Minister or the NSW Planning Director-General.	3 4 5
	(3)	A development control order cannot be given in respect of the following land unless the written consent of the Minister has first been obtained:	6 7
		(a) vacant Crown land within the meaning of the Crown Lands Act 1989,	8
		(b) a reserve within the meaning of Part 5 of the Crown Lands Act 1989,	9
		(c) a common within the meaning of the <i>Commons Management Act 1989</i> .	10
		The Minister must not give consent in respect of vacant Crown land or a reserve within the meaning of Part 5 of the <i>Crown Lands Act 1989</i> until after the Minister has consulted the Minister administering the <i>Crown Lands Act 1989</i> .	11 12 13
	(4)	A copy of any development control order given by a relevant enforcement authority other than a council is to be provided by that authority to the council for the area concerned.	14 15 16
10.6	Prov	isions relating to orders	17
		Part 2 of Schedule 10 contains provisions relating to the giving of orders and related matters.	18 19
10.7	Failu	ire to comply with order—offence	20
	(1)	A person to whom a development control order is given or is taken to have been given must comply with the terms of the order.	21 22
	(2)	It is a sufficient defence to a prosecution for an offence against this section if the defendant satisfies the court that the defendant was unaware of the fact that the matter in respect of which the offence arose was the subject of an order.	23 24 25
		Maximum penalty: Tier 1.	26
		Note. For civil enforcement—see Division 10.3. Note. Schedule 10 provides that a development control order that is given to a person binds a successor in title or occupation of the land concerned and is taken to have been given to the successor. Information about outstanding orders can be obtained under this Act by prospective successors—see Division 11.3.	27 28 29 30 31
Divi	sion	10.3 Civil enforcement proceedings	32
10.8	Defir	nitions	33
		In this Division:	34
		breach of this Act means:	35
		(a) a contravention of or failure to comply with this Act, and	36
		(b) a threatened or an apprehended contravention of or a threatened or apprehended failure to comply with this Act.	37 38
		<i>third-party environmental appeal proceedings</i> means proceedings for an order under section 252 or 253 of the <i>Protection of the Environment Operations Act 1997</i> .	39 40
		this Act includes:	41
		(a) the <i>Planning Administration Act 2013</i> , and	42
		(b) the regulations under this Act or under the <i>Planning Administration Act 2013</i> , and	43 44
		(c) the planning control provisions of a local plan, and	45

10.7

		(d)	a planning approval, including a condition of a planning approval, and	1
		(e)	a development control order, and	2
		(f)	a planning agreement within the meaning of Part 7.	3
10.9	Rest	raint e	etc of breaches of this Act	4
	(1)		person may bring proceedings in the Land and Environment Court for an order medy or restrain a breach of this Act.	5 6
	(2)	Proc	eedings under this section may be brought:	7
		(a)	whether or not any right of the person has been or may be infringed by or as a consequence of that breach, and	8 9
		(b)	whether or not proceedings have been instituted for an offence against this Act.	10 11
	(3)	Proc	eedings under this section may be brought by a person:	12
		(a)	on the person's own behalf, or	13
		(b)	on behalf of another person (with their consent), or of a body corporate or unincorporate (with the consent of its committee or other controlling or governing body), having like or common interests in those proceedings.	14 15 16
	(4)	provi	person on whose behalf proceedings are brought is entitled to contribute to or ide for the payment of the legal costs and expenses incurred by the person ging the proceedings.	17 18 19
	(5)	Note	section is subject to the other provisions of this Division. . Part 3 of Schedule 10 contains special evidentiary provisions relating to proceedings r this Division (eg proceedings with respect to backpackers' hostels and brothels).	20 21 22
10.10	Orde	ers of t	the Court	23
10.10	Orde	If the coming the coming of th	the Court e Land and Environment Court is satisfied that a breach of this Act has been mitted or that a breach of this Act will, unless restrained by order of the Court, pommitted, the Court may make such order as it thinks fit to remedy or restrain reach.	23 24 25 26 27
10.10		If the community of the be control with	e Land and Environment Court is satisfied that a breach of this Act has been mitted or that a breach of this Act will, unless restrained by order of the Court, prommitted, the Court may make such order as it thinks fit to remedy or restrain	24 25 26
10.10	(1)	If the community of the be control with	e Land and Environment Court is satisfied that a breach of this Act has been nitted or that a breach of this Act will, unless restrained by order of the Court, ommitted, the Court may make such order as it thinks fit to remedy or restrain reach.	24 25 26 27 28
10.10	(1)	If the com be co the b With section	e Land and Environment Court is satisfied that a breach of this Act has been nitted or that a breach of this Act will, unless restrained by order of the Court, ommitted, the Court may make such order as it thinks fit to remedy or restrain reach. Nout limiting the powers of the Court under this section, an order made under this on may: if the breach comprises a use of any building, work or land—restrain that use,	24 25 26 27 28 29 30
10.10	(1)	If the communication of the best of the be	e Land and Environment Court is satisfied that a breach of this Act has been mitted or that a breach of this Act will, unless restrained by order of the Court, committed, the Court may make such order as it thinks fit to remedy or restrain reach. Nout limiting the powers of the Court under this section, an order made under this on may: if the breach comprises a use of any building, work or land—restrain that use, or if the breach comprises the erection of a building or the carrying out of a	24 25 26 27 28 29 30 31 32
10.10	(1)	If the communication be contracted by With section (a) (b) (c) If a b	 e Land and Environment Court is satisfied that a breach of this Act has been mitted or that a breach of this Act will, unless restrained by order of the Court, ommitted, the Court may make such order as it thinks fit to remedy or restrain reach. nout limiting the powers of the Court under this section, an order made under this on may: if the breach comprises a use of any building, work or land—restrain that use, or if the breach comprises the erection of a building or the carrying out of a work—require the demolition or removal of that building or work, or if the breach has the effect of altering the condition or state of any building, work or land—require the reinstatement, so far as is practicable, of that building, work or land to the condition or state the building, work or land was 	24 25 26 27 28 29 30 31 32 33 34 35 36
10.10	(1)	If the communication be contracted by With section (a) (b) (c) If a b	 e Land and Environment Court is satisfied that a breach of this Act has been mitted or that a breach of this Act will, unless restrained by order of the Court, formitted, the Court may make such order as it thinks fit to remedy or restrain reach. nout limiting the powers of the Court under this section, an order made under this on may: if the breach comprises a use of any building, work or land—restrain that use, or if the breach comprises the erection of a building or the carrying out of a work—require the demolition or removal of that building or work, or if the breach has the effect of altering the condition or state of any building, work or land—require the reinstatement, so far as is practicable, of that building, work or land to the condition or state the building, work or land was in immediately before the breach. 	24 25 26 27 28 29 30 31 32 33 34 35 36 37 38
10.10	(1)	If the communication be contracted by With section (a) (b) (c) If a b planm	 a Land and Environment Court is satisfied that a breach of this Act has been mitted or that a breach of this Act will, unless restrained by order of the Court, pommitted, the Court may make such order as it thinks fit to remedy or restrain reach. bout limiting the powers of the Court under this section, an order made under this on may: if the breach comprises a use of any building, work or land—restrain that use, or if the breach comprises the erection of a building or the carrying out of a work—require the demolition or removal of that building or work, or if the breach has the effect of altering the condition or state of any building, work or land—require the reinstatement, so far as is practicable, of that building, work or land to the condition or state the building, work or land was in immediately before the breach. 	24 25 26 27 28 29 30 31 32 33 34 35 36 37 38 39 40

	(5)	This section is subject to the other provisions of this Division.	1
10.11		vithin which validity of planning control provisions, planning approvals etc e challenged	2 3
	(1)	In this section:	4
		environmental appeal proceedings or any other kind of proceedings (other than	5 6 7
			8 9
	(2)	proceedings, except those commenced in the Land and Environment Court within 1	10 11 12
	(3)	validity of the planning approval cannot be questioned in any legal proceedings, except those commenced in the Land and Environment Court within 3 months after	13 14 15 16
	(4)		17 18
	(5)	proceedings, the remaining provisions are not invalid merely because they are	19 20 21
	(6)	excluded by this Division merely because this section refers to the commencement	22 23 24
10.12	Exclu	sion of legal proceedings 2	25
	(1)	mandatory in connection with the validity of a strategic plan, an infrastructure plan or a planning approval are the mandatory community participation requirements	26 27 28 29
		(a) the provisions of Part 2 and Schedule 2,	30
			31 32
			33 34
			35 36
	(2)	Schedule 5) relating to public priority infrastructure that are mandatory in connection with the validity of anything done under those provisions is the requirement to	37 38 39 40
	(3)	proceedings or any other kind of proceedings (other than criminal proceedings) 4 cannot be instituted in respect of any of the following after the project definition 4 report for the public priority infrastructure has been published on the NSW planning 4 portal:	41 42 43 44 45
			46 47

(b)	a breach of Division 5.3 of Part 5 (including in relation to the project definition report)
	report),

1 2

3

4

5

6

7

8

9

10

11

12

24

25

47

- (c) a breach of this or any other Act arising in respect of the giving of an approval of the kind referred to in Table 2 to section 6.3 in relation to public priority infrastructure (or in respect of the conditions of such an approval).
- (4) Subsection (3) does not apply to proceedings for an order under this Division, or third-party environmental appeal proceedings, instituted by or with the approval of the Minister.
- (5) This section applies despite any other provision of this Act or any other Act or law.

10.13 Special provision where development consent tainted by corruption

- (1) For the purposes of this section, a decision of a consent authority to grant or modify a development consent is tainted by corrupt conduct:
 - (a) if the Independent Commission Against Corruption, in a report referred to in section 74C of the *Independent Commission Against Corruption Act 1988*, 14 recommends that consideration be given to the suspension of the development 15 consent or modification with a view to its revocation because of serious 16 corrupt conduct by the consent authority or by a councillor or other officer or 17 member of staff of the consent authority in connection with the grant of the 18 consent or modification, or 19
 - (b) if criminal proceedings are instituted against the consent authority or against a councillor or other officer or member of staff of the consent authority for serious corrupt conduct in connection with the grant of the consent or modification, or
 20
 21
 22
 23
 - (c) if the consent authority, councillor or other officer or member of staff makes an admission of such serious corrupt conduct.
- (2) A breach of this Act that may be remedied or restrained in proceedings instituted under this Division includes a decision of a consent authority to grant or modify a development consent that is tainted by corrupt conduct. Section 10.11 (Time within which validity of planning control provisions, planning approvals etc may be challenged) does not apply to any such proceedings.
 26
 27
 28
 29
 30
- (3) If a decision of a consent authority to grant or modify a development consent is tainted by corrupt conduct, the Minister may, without prior notice or inquiry, suspend the decision pending the institution and determination of proceedings for an order under this Division in respect of the decision. The Minister is to give the consent authority and the applicant for the grant or modification of the development consent written notice of the suspension as soon as practicable after it is imposed.
 (3) If a decision of a consent is to give the consent is authority and the applicant for the grant or modification of the development consent is imposed.
- (4) A suspension imposed by the Minister may be lifted by the Minister at any time and is taken to be lifted if the proceedings concerned are not instituted within 6 months after the suspension is imposed.
 37
 38
 39
- (5) The Land and Environment Court may, in proceedings to which this section applies, suspend the decision of a consent authority to grant or modify a development consent pending the determination of the proceedings. The Court may lift a suspension imposed by the Minister under this section.
 40
 41
 42
 43
- (6) The Land and Environment Court may, in proceedings to which this section applies, revoke the decision of a consent authority to grant or modify a development consent if:
 44
 45
 46
 - (a) the decision is tainted by corrupt conduct, and

		(b)	the Court is satisfied that the revocation of the decision will not significantly disadvantage any person affected by the decision who was not a party to the corrupt conduct.	1 2 3
			Court retains its discretion in proceedings to which this section applies as to her to revoke a decision that is tainted by corrupt conduct.	4 5
	(7)	the de be su autho	velopment consent for the erection of a building, the carrying out of a work or emolition of a building or work (or a modification of any such consent) is not to aspended or revoked under this section if the building, work or demolition prised by the development consent (or by the modification) has been eantially commenced.	6 7 8 9 10
	(8)		pensation is not payable by the Minister or the State for any loss suffered by a on because:	11 12
		(a)	a decision is suspended under this section (whether or not the Land and Environment Court decides to revoke the decision), or	13 14
		(b)	a decision is revoked under this section.	15
	(9)	This	section applies:	16
		(a)	to decisions made by a consent authority before or after the commencement of this section, and	17 18
		(b)	to serious corrupt conduct, and to criminal proceedings instituted or admissions made in respect of serious corrupt conduct, before or after that commencement.	19 20 21
	(10)	In thi	is section:	22
			us corrupt conduct means corrupt conduct (within the meaning of the	23
			<i>pendent Commission Against Corruption Act 1988</i>) that may constitute a serious table offence.	24 25
Div	ision ²	indic		
Div 10.14		indict 10.4	table offence.	25
		indict 10.4 mum p If <i>Tie</i> of pro- contr	table offence. Criminal offences and proceedings	25 26
	Maxi	indict 10.4 mum p If <i>Tie</i> of pro- contr	table offence. Criminal offences and proceedings penalty—Tier 1 <i>er 1</i> is specified as the maximum penalty at the end of a provision (or a number ovisions) of this Act or the <i>Planning Administration Act 2013</i> , a person who avenes or fails to comply with that provision (or those provisions) is guilty of fence and (subject to subsection (2)) liable to a penalty not exceeding: in the case of a corporation:	25 26 27 28 29 30
	Maxi	indict 10.4 mum µ If <i>Tie</i> of pro- contr an of	 table offence. Criminal offences and proceedings penalty—Tier 1 er I is specified as the maximum penalty at the end of a provision (or a number ovisions) of this Act or the <i>Planning Administration Act 2013</i>, a person who avenes or fails to comply with that provision (or those provisions) is guilty of fence and (subject to subsection (2)) liable to a penalty not exceeding: in the case of a corporation: (i) \$5 million, and 	25 26 27 28 29 30 31 32 33
	Maxi	indict 10.4 mum p If <i>Tie</i> of pro- contr an of (a)	 table offence. Criminal offences and proceedings penalty—Tier 1 er 1 is specified as the maximum penalty at the end of a provision (or a number ovisions) of this Act or the <i>Planning Administration Act 2013</i>, a person who avenes or fails to comply with that provision (or those provisions) is guilty of fence and (subject to subsection (2)) liable to a penalty not exceeding: in the case of a corporation: (i) \$5 million, and (ii) for a continuing offence—a further \$50,000 for each day the offence continues, or 	25 26 27 28 29 30 31 32 33 34 35
	Maxi	indict 10.4 mum µ If <i>Tie</i> of pro- contr an of	 table offence. Criminal offences and proceedings penalty—Tier 1 er I is specified as the maximum penalty at the end of a provision (or a number ovisions) of this Act or the <i>Planning Administration Act 2013</i>, a person who ravenes or fails to comply with that provision (or those provisions) is guilty of fence and (subject to subsection (2)) liable to a penalty not exceeding: in the case of a corporation: (i) \$5 million, and (ii) for a continuing offence—a further \$50,000 for each day the offence continues, or in the case of an individual: 	25 26 27 28 29 30 31 32 33 34 35 36
	Maxi	indict 10.4 mum p If <i>Tie</i> of pro- contr an of (a)	 table offence. Criminal offences and proceedings penalty—Tier 1 per 1 is specified as the maximum penalty at the end of a provision (or a number ovisions) of this Act or the <i>Planning Administration Act 2013</i>, a person who avenes or fails to comply with that provision (or those provisions) is guilty of fence and (subject to subsection (2)) liable to a penalty not exceeding: in the case of a corporation: (i) \$5 million, and (ii) for a continuing offence—a further \$50,000 for each day the offence continues, or in the case of an individual: (i) \$1 million, and 	25 26 27 28 29 30 31 32 33 34 35 36 37
	Maxii	indict 10.4 mum p If <i>Tie</i> of pro- contr an of (a) (b)	 table offence. Criminal offences and proceedings penalty—Tier 1 per 1 is specified as the maximum penalty at the end of a provision (or a number ovisions) of this Act or the <i>Planning Administration Act 2013</i>, a person who ravenes or fails to comply with that provision (or those provisions) is guilty of fence and (subject to subsection (2)) liable to a penalty not exceeding: in the case of a corporation: (i) \$5 million, and (ii) for a continuing offence—a further \$50,000 for each day the offence continues, or in the case of an individual: (i) \$1 million, and (ii) for a continuing offence—a further \$10,000 for each day the offence continues. 	25 26 27 28 29 30 31 32 33 34 35 36
	Maxi	indict 10.4 mum p If <i>Tie</i> of pro- contr an of (a) (b) A Tie	 table offence. Criminal offences and proceedings penalty—Tier 1 er 1 is specified as the maximum penalty at the end of a provision (or a number ovisions) of this Act or the <i>Planning Administration Act 2013</i>, a person who avenes or fails to comply with that provision (or those provisions) is guilty of fence and (subject to subsection (2)) liable to a penalty not exceeding: in the case of a corporation: (i) \$5 million, and (ii) for a continuing offence—a further \$50,000 for each day the offence continues, or in the case of an individual: (i) \$1 million, and (ii) for a continuing offence—a further \$10,000 for each day the offence continues. 	25 26 27 28 29 30 31 32 33 34 35 36 37 38
	Maxii	indict 10.4 mum p If <i>Tie</i> of pro- contr an of (a) (b) A Tie stand (a)	 table offence. Criminal offences and proceedings penalty—Tier 1 per 1 is specified as the maximum penalty at the end of a provision (or a number ovisions) of this Act or the <i>Planning Administration Act 2013</i>, a person who avenes or fails to comply with that provision (or those provisions) is guilty of fence and (subject to subsection (2)) liable to a penalty not exceeding: in the case of a corporation: (i) \$5 million, and (ii) for a continuing offence—a further \$50,000 for each day the offence continues, or in the case of an individual: (i) \$1 million, and (ii) for a continuing offence—a further \$10,000 for each day the offence continues. er 1 maximum penalty applies only if the prosecution establishes (to the criminal lard of proof): that the offence was committed intentionally, and 	25 26 27 28 29 30 31 32 33 34 35 36 37 38 39 40
	Maxii	indict 10.4 mum p If <i>Tie</i> of pro- contr an of (a) (b) A Tie stand	 table offence. Criminal offences and proceedings penalty—Tier 1 er 1 is specified as the maximum penalty at the end of a provision (or a number ovisions) of this Act or the <i>Planning Administration Act 2013</i>, a person who avenes or fails to comply with that provision (or those provisions) is guilty of fence and (subject to subsection (2)) liable to a penalty not exceeding: in the case of a corporation: (i) \$5 million, and (ii) for a continuing offence—a further \$50,000 for each day the offence continues, or in the case of an individual: (i) \$1 million, and (ii) for a continuing offence—a further \$10,000 for each day the offence continues. 	25 26 27 28 29 30 31 32 33 34 35 36 37 38 39 40 41

		(ii) caused the death of or serious injury or illness to a person.	1
		For the Tier 1 maximum penalty to apply, the court attendance notice or application	2
		commencing the proceedings must allege that those factors apply to the commission of the offence.	3 4
	(3)	If a Tier 1 maximum penalty is specified in this Act or the <i>Planning Administration</i>	5
		Act 2013 but does not apply because of subsection (2), then a Tier 2 maximum penalty applies instead.	6 7
10.15	Maxi	imum penalty—Tier 2	8
		If <i>Tier 2</i> is specified as the maximum penalty at the end of a provision (or a number	9
		of provisions) of this Act or the <i>Planning Administration Act 2013</i> , a person who contravenes or fails to comply with that provision (or those provisions) is guilty of	10 11
		an offence and liable to a penalty not exceeding:	12
		(a) in the case of a corporation:	13
		(i) \$2 million, and	14
		(ii) for a continuing offence—a further \$20,000 for each day the offence continues, or	15 16
		(b) in the case of an individual:	17
		(i) \$500,000, and	18
		(ii) for a continuing offence—a further \$5,000 for each day the offence continues.	19 20
10.16	Maxi	imum penalty—Tier 3	21
		If <i>Tier 3</i> is specified as the maximum penalty at the end of a provision (or a number of provisions) of this Act or the <i>Planning Administration Act 2013</i> , a person who contravenes or fails to comply with that provision (or those provisions) is guilty of an offence and liable to a penalty not exceeding:	22 23 24 25
		(a) in the case of a corporation:	26
		(i) \$1 million, and	27
		(ii) for a continuing offence—a further \$10,000 for each day the offence continues, or	28 29
		(b) in the case of an individual:	30
		(i) \$250,000, and	31
		(ii) for a continuing offence—a further \$2,500 for each day the offence continues.	32 33
		Note. Section 11.1 provides that the regulations may create offences and impose a penalty for an offence against the regulations not exceeding \$110,000.	34 35
10.17	Proc	seedings for offences	36
-	(1)	Proceedings for an offence against the planning legislation may be dealt with	37
	(1)	summarily before the Local Court or before the Land and Environment Court in its summary jurisdiction.	38 39
	(2)	If proceedings for an offence are brought in the Local Court, the maximum monetary	40
		penalty that the Local Court may impose for the offence is, despite any other provision of the planning legislation, \$110,000 or the maximum monetary penalty provided for the offence, whichever is the lesser.	41 42 43

10.18 Criminal and civil enforcement proceedings in respect of same matter

- (1) A person is not to be convicted of an offence against the planning legislation where the matter constituting the offence, at the date on which the conviction would but for this subsection be made:
 - (a) is the subject of proceedings for an order under Division 10.3, which proceedings have not been concluded, or

1

2

3

4

5

6

7

8

9

10

11

25

26

27

28

29

30

31

32

33

34

35

36

37

43

- (b) is the subject of an order made under Division 10.3.
- (2) Nothing in subsection (1) (a) precludes a conviction being made where those proceedings are concluded otherwise than by the making of an order under Division 10.3.

10.19 Time within which proceedings may be commenced

- Proceedings for an offence against the planning legislation may be commenced not later than 2 years after the date on which the offence is alleged to have been committed.
- Proceedings for an offence against the planning legislation may also be commenced within, but not later than, 2 years after the date on which evidence of the alleged offence first came to the attention of any relevant investigation officer.
- (3) If subsection (2) is relied on for the purpose of commencing proceedings for an offence, the court attendance notice or application must contain particulars of the date on which evidence of the offence first came to the attention of any relevant investigation officer and need not contain particulars of the date on which the offence was committed. The date on which evidence first came to the attention of any relevant investigation officer is the date specified in the court attendance notice or application, unless the contrary is established.
- (4) This section applies despite anything in the *Criminal Procedure Act 1986* or any other Act.
- (5) In this section:

evidence of an offence means evidence of any act or omission constituting the offence.

investigation officer means an investigation officer within the meaning of Part 8 of the *Planning Administration Act 2013*, whether or not the person has the functions of an investigation officer in connection with the offence concerned.

relevant investigation officer means:

- (a) in relation to proceedings for an offence instituted by or with the consent of the NSW Planning Director-General or other member of the staff of the Department of Planning and Infrastructure—any investigation officer who is a member of the staff of the Department, or
- (b) in relation to proceedings for an offence instituted by or with the consent of a council or a member of staff of a council—any investigation officer who is a member of staff of that council, or
 38
 39
 40
- (c) in relation to proceedings for an offence instituted by any other person—any 41 investigation officer. 42

10.20 Penalty notices for certain offences

 An authorised person may serve a penalty notice on a person if it appears to the authorised person that the person has committed an offence under the planning legislation, being an offence prescribed by the regulations.

(2)	have perso	nalty notice is a notice to the effect that, if the person served does not wish to the matter determined by a court, the person may pay, within the time and to the on specified in the notice, the amount of penalty prescribed by the regulations for ffence if dealt with under this section.	1 2 3 4
(3)	A pe	nalty notice:	5
	(a)	may be served personally or by post, or	6
	(b)	if it relates to an offence involving the use of a vehicle, may be addressed to the owner (without naming the owner or stating the owner's address) and may be served by leaving it on or attaching it to the vehicle.	7 8 9
(4)		e amount of penalty prescribed for an alleged offence is paid under this section, erson is liable to any further proceedings for the alleged offence.	10 11
(5)	of, ar	nent under this section is not regarded as an admission of liability for the purpose and does not in any way affect or prejudice, any civil claim, action or proceeding ag out of the same occurrence.	12 13 14
(6)	The 1	regulations may:	15
	(a)	prescribe an offence for the purposes of this section by specifying the offence or by referring to the provision creating the offence, and	16 17
	(b)	prescribe the amount of penalty payable for the offence if dealt with under this section, and	18 19
	(c)	prescribe different amounts of penalties for different offences or classes of offences, and	20 21
	(d)	prescribe different amounts of penalties for the same offence, including, in the case of a continuing offence, different amounts of penalties for different periods during which the offence continues.	22 23 24
(7)		amount of a penalty prescribed under this section for an offence must not exceed naximum amount of penalty which could be imposed for the offence by a court.	25 26
(8)		section does not limit the operation of any other provision of, or made under, or any other Act relating to proceedings which may be taken in respect of ices.	27 28 29
(9)	to be	is section, <i>authorised person</i> means a person who is declared by the regulations an authorised person for the purposes of this section or who belongs to a class rsons so declared.	30 31 32
Anci	llary o	ffences	33
	A pe	rson who:	34
	(a)	aids, abets, counsels or procures another person to commit, or	35
	(b)	conspires to commit,	36
	agair	fence under another provision of the planning legislation is guilty of an offence ast that other provision and is liable, on conviction, to the same penalty cable to an offence against that other provision.	37 38 39
Offei	nce—f	alse or misleading information	40
(1)	perso partic	rson who provides information in connection with a planning matter that the on knows, or ought reasonably to know, to be false or misleading in a material cular is guilty of an offence. mum penalty: Tier 3.	41 42 43 44

- (2) For the purposes of this section, a person provides information in connection with a planning matter if:
 - (a) the person is an applicant for a planning approval (or for the modification of a planning approval) and the information is provided by the applicant in or in connection with the application, or

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21 22

23

24

25

26

27

28

29

30

31

32

44

- (b) the person is engaged by any such applicant and the information is provided by that person for the purposes of the application, or
- (c) the person is a proponent of proposed development and the information is provided in or in connection with a formal request to the Minister, a council, the NSW Planning Director-General or other planning authority for the making of provisions of a strategic or other plan or Ministerial planning order under this Act in relation to the proposed development, or
- (d) the person provides information in connection with any other matter or thing under the planning legislation that the regulations declare to be the provision of information in connection with a planning matter for the purposes of this section.
- (3) An environmental impact statement or other document is part of information provided in connection with a matter if it forms part of or accompanies the matter or is subsequently submitted in support of the matter.

Note. The *Crimes Act 1900* contains other offences relating to false and misleading information: section 192G (Intention to defraud by false or misleading statement—maximum penalty 5 years imprisonment); sections 307A, 307B and 307C (False or misleading applications/information/documents—maximum penalty 2 years imprisonment or \$22,000, or both).

10.23 Continuing obligations under orders etc

- (1) A requirement of an order, notice or other instrument under the planning legislation to do or cease to do something before a particular time or within a specified period continues, after the time has expired or the period has ended, until it is complied with or the instrument is revoked.
- (2) This section does not apply if, in the context, the planning legislation does not require continuing compliance.

10.24 Miscellaneous provisions

- Part 8.3 of the *Protection of the Environment Operations Act 1997* (Court orders in connection with offences) applies to an offence against the planning legislation in the same way as it applies to an offence against that Act or the regulations under that Act, but only in relation to proceedings before the Land and Environment Court and subject to any modifications prescribed by the regulations under this Act.
 Note. An offence under section 251 of that Act in relation to an order will become an offence against this Act.
 Will and the form offence in the section 251 of that Act in relation to an order will become an offence against this Act.
- Where a person is guilty of an offence against the planning legislation involving the destruction of or damage to a tree, the court dealing with the offence may, in addition to or in substitution for any pecuniary penalty imposed or liable to be imposed, direct that person:
 - (a) to plant new trees and maintain those trees to a mature growth, and
 - (b) to provide security for the performance of any obligation imposed under 45 paragraph (a). 46

(3) In determining the sentence for a person who has previously been found guilty of an offence that arises from a failure to comply with a brothel closure order or the unlawful use of premises for the purposes of a brothel, a court must take into account the fact of the previous offence as an aggravating factor and is, accordingly, to impose a higher sentence than it would otherwise impose.

Par	t 11	Miscellaneous	1
Divi	sion ²	11.1 Regulations	2
11.1	Regu	lations generally	3
	(1)	The Governor may make regulations, not inconsistent with this Act, for or with respect to any matter that by this Act is required or permitted to be prescribed or that is necessary or convenient to be prescribed for carrying out or giving effect to this Act.	4 5 6 7
	(2)	Regulations that may be made in connection with a particular Part of this Act may be made separately or by amending the corresponding Schedule to this Act relating to ancillary provisions for that Part (whether by addition to, or by omission or amendment of, the ancillary provisions of that Schedule).	8 9 10 11
	(3)	The regulations may create offences punishable by a penalty not exceeding \$110,000.	12 13
	(4)	The regulations may make provision for or with respect to any matter that, by a provision of this Act, is to be or may be determined by the Minister or NSW Planning Director-General. Any such determination of the Minister or NSW Planning Director-General is subject to the provisions of those regulations.	14 15 16 17
	(5)	A regulation may apply, adopt or incorporate any publication as in force from time to time.	18 19
11.2	Spec	ific regulation-making powers	20
		The regulations may, in particular, make provision for or with respect to the matters set out in Part 1 of Schedule 11. Other regulation-making powers are contained in other provisions of this Act.	21 22 23
11.3	Savir	ngs and transitional regulation-making powers	24
	(1)	The regulations may contain provisions of a savings or transitional nature consequent on the enactment of this Act, the <i>Planning Administration Act 2013</i> or any Act that amends this Act or that Act, or consequent on the making or amendment of provisions or instruments under this Act or that Act. Note. See Schedule 12 for specific savings and transitional provisions consequent on the enactment of this Act and other matters.	25 26 27 28 29 30
	(2)	Any such provision of the regulations has effect despite anything to the contrary in Schedule 12. The regulations may make separate savings and transitional provisions or amend Schedule 12 to consolidate the savings and transitional provisions.	31 32 33
	(3)	Any such provision of the regulations may, if the regulations so provide, take effect from the date of assent to the Act (or making of the provision or instrument) concerned or a later date.	34 35 36
	(4)	To the extent to which any such provision of the regulations takes effect from a date that is earlier than the date of its publication on the NSW legislation website, the provision does not operate so as:	37 38 39
		(a) to affect, in a manner prejudicial to any person (other than the State or an authority of the State), the rights of that person existing before the date of its publication, or	40 41 42
		(b) to impose liabilities on any person (other than the State or an authority of the State) in respect of anything done or omitted to be done before the date of its publication.	43 44 45

Division 11.2 Fees and charges

11.4	Fees	and charges to which this Division applies	2
	(1)	This Division applies to the fees payable under the planning legisla who makes an application for a planning approval or information c seeks the exercise of any other function under the planning legislation	ertificate or who 4
	(2)	This Division also applies to fees for services provided in com administration of the planning legislation by the Minister, the Plan Corporation, the NSW Planning Director-General or other planning	ning Ministerial 7
	(3)	Those fees include, but are not limited to, the administrative of provided under the planning legislation and the costs associated with the Minister, the Planning Ministerial Corporation, the Director-General or other planning bodies.	n the functions of 10
11.5	Pres	cription of fees and charges and their payment	13
		The regulations may make provision for or with respect to the determination of the amount) of fees and charges to which this Divitime the fees and charges are due for payment and the payment and refees and charges.	ision applies, the 15
Divi	sion	11.3 Planning and building information certificates	18
11.6	Infor	mation certificates to which this Division applies	19
		This Division applies to the following certificates (an <i>information</i> a	<i>certificate</i>): 20
		(a) planning information certificates,	21
		(b) building information certificates.	22
11.7	Who	may apply for information certificates	23
	(1)	Any person may apply for a planning information certificate in relati parcel of land.	on to a particular 24 25
	(2)	The following persons may apply for a building information certific a building:	cate in relation to 26 27
		(a) the owner of the land on which the building is erected,	28
		(b) any other person with the consent of the owner of that land,	29
		(c) the purchaser under a contract for the sale of property th includes the building, or the purchaser's Australian legal practice.	titioner or agent, 31
		(d) a public authority that has notified the owner of that land o apply for the certificate.	f its intention to 32 33
11.8	Maki	ng of applications for information certificates	34
	(1)	Applications for information certificates are to be made to the cound which the land to which the application relates is situated.	cil for the area in 35 36
	(2)	The regulations may provide for the procedure for making an applications for information certificates. Note. Division 11.2 enables the regulations to prescribe the fee for an certificate.	38
	(3)	The regulations may assign an area that is outside a local governme of a specified adjoining local government area in relation to informa	

For the purposes of this Division, the assigned area is taken to be a part of the local government area concerned.

1

2

3

4

5

6

7

8

9

10

11

12

13

20

22

23

24

25

29

30

31

32

33

34

35

36

37

38

39

43

44

45

11.9 Issue of information certificates

- A council is (subject to this Division) required to issue an information certificate as (1)soon as practicable after an application for the certificate is made to the council.
- The regulations may prescribe the form and manner in which an information (2)certificate is issued.

11.10 Nature and effect of planning information certificates

- A planning information certificate is a certificate that specifies such matters relating (1)to the land to which the certificate relates as are prescribed by the regulations (whether arising under this or any other Act or otherwise).
- (2)A council may, in a planning information certificate, include advice on such other relevant matters affecting the land of which it may be aware.
- (3) For the purpose of any proceedings for an offence against the planning legislation 14 which may be taken against a person who has obtained a planning information 15 certificate or who might reasonably be expected to rely on that certificate, that 16 certificate is, in favour of that person, conclusively presumed to be true and correct. 17 Note. A planning information certificate does not preclude civil enforcement proceedings 18 under this Act to remedy or restrain a breach of this Act if the certificate is not in fact correct. 19
- (4)A council does not incur any liability in respect of any advice provided in good faith under subsection (2). However, this subsection does not apply to advice provided in 21 relation to contaminated land (including the likelihood of land being contaminated land) or to the nature or extent of contamination of land within the meaning of Division 11.5.

11.11 Issue, nature and effect of building information certificate

- A building information certificate is to be issued by a council only if it appears that: (1)26
 - there is no matter discernible by the exercise of reasonable care and skill that (a) 27 would entitle the council, under this Act or the Local Government Act 1993: 28
 - (i) to order the building to be repaired, demolished, altered, added to or rebuilt, or
 - to take proceedings for an order or injunction requiring the building to (ii) be demolished, altered, added to or rebuilt, or
 - (iii) to take proceedings in relation to any encroachment by the building onto land vested in or under the control of the council, or
 - there is such a matter but, in the circumstances, the council does not propose (b) to make any such order or take any such proceedings.
- A building information certificate is a certificate that states that the council will not (2)make an order or take proceedings referred to in subsections (3) and (4).
- (3) A building information certificate operates to prevent the council:
 - from making an order (or taking proceedings for the making of an order or (a) 40 injunction) under this Act or the Local Government Act 1993 requiring the 41 building to be repaired, demolished, altered, added to or rebuilt, and 42
 - (b) from taking civil proceedings in relation to any encroachment by the building onto land vested in or under the control of the council,

in relation to matters existing or occurring before the date of issue of the certificate.

(4)	A building information certificate operates to prevent the council, for a period of 7 years from the date of issue of the certificate:	1 2
	(a) from making an order (or taking proceedings for the making of an order or injunction) under this Act or the <i>Local Government Act 1993</i> requiring the building to be repaired, demolished, altered, added to or rebuilt, and	3 4 5
	(b) from taking civil proceedings in relation to any encroachment by the building onto land vested in or under the control of the council,	6 7
	in relation to matters arising only from the deterioration of the building as a result solely of fair wear and tear.	8 9
(5)	However, a building information certificate does not operate to prevent a council from making a development control order that is a fire safety order.	10 11
(6)	An order or proceeding that is made or taken in contravention of this section is of no effect.	12 13
Misc	ellaneous provisions relating to building information certificates	14
(1)	A building information certificate may apply to the whole or to part only of a building.	15 16
(2)	On receipt of an application for a building information certificate, the council may, by notice in writing served on the applicant, require the applicant to supply it with such information (including building plans, specifications, survey reports and certificates) as may reasonably be necessary to enable the proper determination of the application.	17 18 19 20 21
(3)	If the applicant is able to provide evidence that no material change has occurred in relation to the building since the date of a survey certificate which, or a copy of which, is supplied to the council by the applicant, the council is not entitled to require the applicant to supply a more recent survey certificate.	22 23 24 25
(4)	If the council refuses to issue a building information certificate, it must inform the applicant, by notice, of its decision and of the reasons for it.	26 27
(5)	The reasons must be sufficiently detailed to inform the applicant of the work that needs to be done to enable the council to issue a building information certificate.	28 29
(6)	The council must not refuse to issue or delay the issue of a building information certificate by virtue of the existence of a matter that would not entitle the council to make any order or take any proceedings of the kind referred to in section $11.11(1)$ (a).	30 31 32 33
(7)	Nothing in this section prevents the council from informing the applicant of the work that would need to be done before the council could issue a building information certificate or from deferring its determination of the application until the applicant has had an opportunity to do that work.	34 35 36 37
(8)	The council must keep a record of building information certificates issued.	38
(9)	A person may inspect the record at any time during the ordinary office hours of the council.	39 40
(10)	A person may obtain a copy of a building information certificate from the record with the consent of the owner of the building.	41 42

Divi	sion	11.4	Existing uses	1
11.13	Defir	nition o	of "existing use" etc	2
	(1)	In thi	s Division, <i>existing use</i> means:	3
		(a)	the use of a building, work or land for a lawful purpose immediately before the coming into force of a planning instrument which would have the effect of prohibiting that use, and	4 5 6
		(b)	the use of a building, work or land:	7
			(i) for which development consent was granted before the commencement of a provision of a planning instrument having the effect of prohibiting the use, and	8 9 10
			(ii) for which development was carried out, within 1 year after the date on which that provision commenced, in accordance with the terms of the development consent and to such an extent as to ensure that the development consent would not lapse.	11 12 13 14
		regar	he purposes of this definition, a use prohibited by a planning instrument is to be ded as prohibited despite provisions of this Act that enable development bited by planning instruments to be carried out.	15 16 17
	(2)		is Division, <i>planning instrument</i> means the planning control provisions of a plan, including a transitional planning instrument.	18 19
11.14	Cont	inuano	ce of and limitations on existing use	20
	(1)		ing in this Act, the regulations or a planning instrument prevents the nuance of an existing use.	21 22
	(2)	Subse contr	ection (1) is subject to any express provision of this Act or the regulations to the ary.	23 24
	(3)	Noth	ing in subsection (1) authorises:	25
		(a)	any alteration or extension to or rebuilding of a building or work, or	26
		(b)	any increase in the area of the use made of a building, work or land from the area actually physically and lawfully used immediately before the coming into operation of the planning instrument concerned, or	27 28 29
		(c)	without affecting paragraph (a) or (b), any enlargement, expansion or intensification of an existing use, or	30 31
		(d)	the continuance of the use concerned in breach of any planning approval in force under this Act in relation to that use or any condition imposed or applicable to that planning approval (including a condition relating to the modification or surrender of the existing use), or	32 33 34 35
		(e)	the continuance of the use concerned if it is abandoned.	36
	(4)		out limitation, a use is presumed to be abandoned (unless the contrary is lished) if it ceases to be actually used for a continuous period of 12 months.	37 38
11.15	Regu	lation	s respecting existing use	39
	(1)		regulations may make provision for or with respect to existing uses and, in cular, for or with respect to:	40 41
		(a)	the carrying out of alterations or extensions to or the rebuilding of a building or work being used for an existing use, and	42 43
		(b)	the change of an existing use to another use, and	44
		(c)	the enlargement or expansion or intensification of an existing use.	45

	(2)	part	provisions of any such regulations that are in force have effect as if they were of the planning control provisions of each local plan (<i>the incorporated isions</i>).	1 2 3
	(3)	supp] subse	blanning instrument may contain provisions extending, expanding or lementing the incorporated provisions, but any such provisions that, but for this ection, would derogate from the incorporated provisions have no force or effect the incorporated provisions remain in force.	4 5 6 7
	(4)	planr provi to a devel	right or authority granted by the incorporated provisions or any provisions of a ning instrument extending, expanding or supplementing the incorporated isions do not apply to or in respect of an existing use which commenced pursuant consent of the Minister under section 4.44 (Determination of Crown lopment applications) to an application for consent to carry out prohibited lopment.	8 9 10 11 12 13
11.16	Cont	inuan	ce of and limitations on other lawful uses	14
	(1)	obtai lawfu of th	ing in a planning instrument operates so as to require development consent to be ned under this Act for the continuance of a use of a building, work or land for a al purpose for which it was being used immediately before the coming into force the instrument or so as to prevent the continuance of that use except with lopment consent under this Act being obtained.	15 16 17 18 19
	(2)	Noth	ing in subsection (1) authorises:	20
		(a)	any alteration or extension to or rebuilding of a building or work, or	21
		(b)	any increase in the area of the use made of a building, work or land from the area actually physically and lawfully used immediately before the coming into operation of the planning instrument concerned, or	22 23 24
		(c)	without affecting paragraph (a) or (b), any enlargement or expansion or intensification of the use concerned, or	25 26
		(d)	the continuance of the use concerned in breach of any planning approval in force under this Act in relation to that use or any condition imposed or applicable to that approval (including a condition relating to the modification or surrender of the existing use), or	27 28 29 30
		(e)	the continuance of the use concerned if it is abandoned.	31
	(3)		out limitation, a use is presumed to be abandoned (unless the contrary is lished) if it ceases to be actually used for a continuous period of 12 months.	32 33
11.17	Uses	unlav	wfully commenced	34
			use of a building, work or land which was unlawfully commenced is not ered lawful by the occurrence of any subsequent event except:	35 36
		(a)	the commencement of a planning instrument which permits the use without the need to obtain development consent, or	37 38
		(b)	the granting of development consent to that use.	39
11.18	Savi	ng of e	effect of existing development consents	40
	(1)	conse	ing in a planning instrument prohibits, or requires a further development ent to authorise, the carrying out of development in accordance with a lopment consent that has been granted and is in force.	41 42 43
	(2)	This	section:	44
		(a)	applies to development consents lawfully granted before or after the commencement of this Act, and	45 46

11.20

		(b)	does not prevent the lapsing or modification, in accordance with this Act, of a development consent, and	1 2
		(c)	has effect despite anything to the contrary in this Division.	3
Divi	sion	11.5	Contaminated land liability	4
1.19	Defin	itions		5
		In thi	s Division:	6
		conce on or	<i>uminated land</i> means land in, on or under which any substance is present at a entration above the concentration at which the substance is normally present in, or under (respectively) land in the same locality, being a presence that presents a of harm to human health or any other aspect of the environment.	7 8 9 10
			<i>ning authority</i> means the public authority or other person or body responsible xercising a planning function.	11 12
			ning function means:	13
		(a)	the making, amendment or repeal of any provisions of a local plan or of any other strategic plan or the preparation of a draft plan or proposal under Part 3 for that purpose, or	14 15 16
		(b)	determining an application for a planning approval or a modification of a planning approval, or	17 18
		(c)	furnishing advice or information in an information certificate, or	19
		(d)	any function under the planning legislation prescribed by the regulations, or	20
		(e)	anything incidental or ancillary to the carrying out of any of the above functions.	21 22
.20	Exen	nption	from liability—contaminated land	23
	(1)	omitt funct likeli	anning authority does not incur any liability in respect of anything done or ted to be done in good faith by the authority in duly exercising any planning ion of the authority in so far as it relates to contaminated land (including the hood of land being contaminated land) or to the nature or extent of mination of land.	24 25 26 27 28
	(2)	acted have accor	out limiting any other circumstance in which a planning authority may have in good faith, a planning authority is (unless the contrary is proved) taken to acted in good faith if the thing was done or omitted to be done substantially in rdance with the contaminated land planning guidelines in force under this sion at the time the thing was done or omitted to be done.	29 30 31 32 33
	(3)		protection conferred on a planning authority by this section extends to its oyees and agents and to the members of any governing body of the planning ority.	34 35 36
.21	Cont	amina	ted land planning guidelines	37
	(1)	guide	the purposes of this Division, the Minister may, from time to time, issue planning elines relating to contaminated land (the <i>contaminated land planning</i> <i>elines</i>).	38 39 40
	(2)		egulations may make provision with respect to the making of contaminated land ning guidelines, including the public exhibition of draft guidelines.	41 42
	(3)		contaminated land planning guidelines (and any amendment or replacement of guidelines) are to be published on the NSW planning portal.	43 44

Division 11.6 Bush fire prone land

11.22	Bush	fire p	prone land	2
	(1)		bush fire risk management plan applies to land within the area of a council, the cil must, every 5 years:	3 4
		(a)	request the Commissioner of the NSW Rural Fire Service to designate land (if any) within the area that the Commissioner considers, having regard to the bush fire risk management plan, to be bush fire prone land, and	5 6 7
		(b)	must record any land so designated on a map.	8
	(2)	desig	Commissioner of the NSW Rural Fire Service must, if satisfied that the land mated by the Commissioner has been recorded by the council on a map, certify map as a bush fire prone land map for the area of the council.	9 10 11
	(3)	regul	Commissioner of the NSW Rural Fire Service may, in accordance with the ations, review the designation of land on a bush fire prone land map for an area y time after the map is certified and revise the map accordingly. The revised	12 13 14 15
		(a)	becomes the bush fire prone land map for the area on being certified by the Commissioner, and	16 17
		(b)	is to be provided to the council by the Commissioner.	18
	(4)		recorded for the time being as bush fire prone land on a bush fire prone land for an area is bush fire prone land for the area for the purposes of this or any Act.	19 20 21
	(5)		bush fire prone land map for an area is to be available for public inspection ag normal office hours for the council.	22 23
	(6)	In thi	is section:	24
		bush Act 1	<i>fire risk management plan</i> has the same meaning as it has in the <i>Rural Fires</i> 997.	25 26
		-	includes spatial dataset.	27
			Division 8 of Part 4 of the <i>Rural Fires Act 1997</i> contains provisions relating to the ng out of development and bush fire hazard reduction work on bush fire prone land.	28 29
Divis	sion '	11.7	General miscellaneous	30
			additional miscellaneous provisions are contained in Schedule 11:	31
Paper	subdiv	isions	(Part 2 of Schedule 11).	32
11.23	Act to	o bind	l Crown	33
		powe	Act binds the Crown in right of New South Wales and, in so far as the legislative er of the Parliament of New South Wales permits, the Crown in all its other cities.	34 35 36
11.24	Discl	osure	of political donations	37
	(1)	or gif	object of this section is to require the disclosure of relevant political donations fts when planning applications are made to minimise any perception of undue ence by:	38 39 40
		(a)	requiring public disclosure of the political donations or gifts at the time planning applications (or public submissions relating to them) are made, and	41 42

(b) providing the opportunity for appropriate decisions to be made about the persons who will determine or advise on the determination of the planning applications.

Political donations or gifts are not relevant to the determination of any such planning application, and the making of political donations or gifts does not provide grounds for challenging the determination of any such planning application.

Note. This Act makes provision for planning applications to be referred to various bodies for advice or determination. Section 10.13 makes special provision where development consent is tainted by corruption. The *Local Government Act 1993* makes provision with respect to voting by local councillors with a conflict of interest in any matter before the council.

(2) In this section:

gift means a gift within the meaning of Part 6 of the Election Funding, Expenditure and Disclosures Act 1981.

Note. A gift includes a gift of money or the provision of any other valuable thing or service for no consideration or inadequate consideration.

local councillor means a councillor (including the mayor) of the council of a local government area.

relevant planning application means:

- (a) a formal request to the Minister, a council or the NSW Planning Director-General to initiate the making of provisions of a local plan in relation to development on a particular site, or
- (b) a formal request to the Minister or the NSW Planning Director-General for development on a particular site to be made State significant development or State infrastructure development, or
- (c) an application for approval of State infrastructure development (or for the modification of the approval for any such development), or
- (d) an application for development consent under Part 4 (or for the modification of a development consent), or
- (e) any other application or request under or for the purposes of the planning legislation that is prescribed by the regulations as a relevant planning application,

but does not include:

- (f) an application for (or for the modification of) a complying development certificate, or
- (g) an application or request made by a public authority on its own behalf or made on behalf of a public authority, or
- (h) any other application or request that is excluded from this definition by the regulations.

relevant public submission means a written submission made by a person objecting to or supporting a relevant planning application or any development that would be authorised by the granting of the application.

reportable political donation means a reportable political donation within the meaning of Part 6 of the *Election Funding, Expenditure and Disclosures Act 1981* that is required to be disclosed under that Part.

Note. Reportable political donations include those of or above \$1,000.

- (3) A person:
 - (a) who makes a relevant planning application to the Minister or the NSW
 Planning Director-General is required to disclose all reportable political
 donations (if any) made within the relevant period to anyone by any person
 with a financial interest in the application, or

(b) who makes a relevant public submission to the Minister or the NSW Planning Director-General in relation to the application is required to disclose all reportable political donations (if any) made within the relevant period to anyone by the person making the submission or any associate of that person.

1

2

3

4

5

6

7

8

9

10

11

12

13

22

23

27

28

The relevant period is the period commencing 2 years before the application or submission is made and ending when the application is determined.

- (4) A person who makes a relevant planning application to a council is required to disclose the following reportable political donations and gifts (if any) made by any person with a financial interest in the application within the period commencing 2 years before the application is made and ending when the application is determined:
 - (a) all reportable political donations made to any local councillor of that council,
 - (b) all gifts made to any local councillor or employee of that council.

A reference in this subsection to a reportable political donation made to a local councillor includes a reference to a donation made at the time the person was a candidate for election to the council.

- (5) A person who makes a relevant public submission to a council in relation to a relevant planning application made to the council is required to disclose the following reportable political donations and gifts (if any) made by the person making the submission or any associate of that person within the period commencing 2 years before the submission is made and ending when the application is determined: 21
 - (a) all reportable political donations made to any local councillor of that council,
 - (b) all gifts made to any local councillor or employee of that council.

A reference in this subsection to a reportable political donation made to a local councillor includes a reference to a donation made at the time the person was a candidate for election to the council. 26

- (6) The disclosure of a reportable political donation or gift under this section is to be made:
 - (a) in, or in a statement accompanying, the relevant planning application or submission if the donation or gift is made before the application or submission is made, or
 29
 30
 31
 - (b) if the donation or gift is made afterwards, in a statement to the person to whom
 the relevant planning application or submission was made within 7 days after
 the donation or gift is made.
 34
- (7) For the purposes of this section, a person has a financial interest in a relevant planning application if:
 35
 - (a) the person is the applicant or the person on whose behalf the application is 37 made, or 38
 - (b) the person is an owner of the site to which the application relates or has entered 39 into an agreement to acquire the site or any part of it, or 40
 - (c) the person is associated with a person referred to in paragraph (a) or (b) and is
 likely to obtain a financial gain if development that would be authorised by the
 application is authorised or carried out (other than a gain merely as a
 shareholder in a company listed on a stock exchange), or
 - (d) the person has any other interest relating to the application, the site or the owner of the site that is prescribed by the regulations. 46

(8)	For t	he purposes of this section, persons are associated with each other if:	1
	(a)	they carry on a business together in connection with the relevant planning application (in the case of the making of any such application) or they carry on a business together that may be affected by the granting of the application (in the case of a relevant planning submission), or	2 3 4 5
	(b)	they are related bodies corporate under the <i>Corporations Act 2001</i> of the Commonwealth, or	6 7
	(c)	they are directors of the same body corporate, or they are directors of different bodies corporate that are related bodies corporate under the <i>Corporations Act 2001</i> of the Commonwealth, or	8 9 10
	(d)	one is a director of a body corporate and the other is a related body corporate under the <i>Corporations Act 2001</i> , or	11 12
	(e)	they have any other relationship prescribed by the regulations.	13
(9)	discl	disclosure of reportable political donations under this section is to include osure of the following details of each such donation made during the relevant osure period:	14 15 16
	(a)	the name of the party or person for whose benefit the donation was made,	17
	(b)	the date on which the donation was made,	18
	(c)	the name of the donor,	19
	(d)	the residential address of the donor (in the case of an individual) or the address of the registered or other official office of the donor (in the case of an entity),	20 21
	(e)	the amount (or value) of the donation.	22
	Note Part (The above details are the details required to be disclosed of political donations under 6 of the <i>Election Funding, Expenditure and Disclosures Act 1981</i> .	23 24
(10)		disclosure of gifts under this section is to include disclosure of the following ils of each such gift made during the relevant disclosure period:	25 26
	(a)	the name of the person to whom the gift was made,	27
	(b)	the date on which the gift was made,	28
	(c)	the name of the person who made the gift,	29
	(d)	the residential address of the person who made the gift (in the case of an individual) or the address of the registered or other official office of the person who made the gift (in the case of an entity),	30 31 32
	(e)	the amount (or value) of the gift.	33
(11)	acco	section only requires the disclosure of a political donation or gift by a person in rdance with this section that the person knows, or ought reasonably to know, was e and is required to be disclosed under this section.	34 35 36
(12)		losures of reportable political donations and gifts under this section are to be e available to the public on, or in accordance with arrangements notified on:	37 38
	(a)	a website maintained by the NSW Planning Director-General (in the case of planning applications or submissions made to the Minister or the NSW Planning Director-General), or	39 40 41

		(b)		bsite maintained by the council (in the case of planning applications or hissions made to that council).	1 2
				ures are to be made so available within 14 days after the disclosures are this section.	3 4
		Expe	nditur	penalty: The maximum penalty under Part 6 of the <i>Election Funding</i> , <i>e and Disclosures Act 1981</i> for making a false statement in a declaration res lodged under that Part.	5 6 7
11.25	Proc	of of ov	wnersł	hip of land	8
				al proceedings under the planning legislation, in addition to any other proof available:	9 10
		(a)	any r lesse	ence that the person proceeded against is rated in respect of any land to ate under the <i>Local Government Act 1993</i> , otherwise than as a rate paying e, is, until the contrary is proved, evidence that the person is the owner of and, or	11 12 13 14
		(b)	any l is pro	tificate issued by the Registrar-General as to the proprietor or owner of and at a particular time or during a particular period is, until the contrary oved, evidence that the person described in the certificate as the proprietor wher of the land was the owner of that land at the relevant time or period.	15 16 17 18
11.26	Righ	t to be	e heard	d—legal representation	19
	(1)	entitl		ning legislation confers a right on a person to be heard, the person is be heard personally or to be represented by an Australian legal practitioner.	20 21 22
	(2)	That	entitle	ment is subject to the regulations.	23
11.27	Serv	ice or	giving	g of documents	24
	(1)			It that is authorised or required by or under the planning legislation to be served on any person may be given or served by:	25 26
		(a)	in the	e case of a natural person:	27
			(i)	delivering it to the person personally, or	28
			(ii)	sending it by post to the address specified by the person for the giving or service of documents or, if no such address is specified, the residential or business address of the person last known to the person giving or serving the document, or	29 30 31 32
			(iii)	sending it by facsimile transmission to the facsimile number of the person, or	33 34
			(iv)	sending it by electronic transmission (including for example the Internet) to the person in accordance with arrangements indicated by the person as appropriate for transmitting documents to the person, or	35 36 37
		(b)	in the	e case of a body corporate:	38
			(i)	leaving it with a person apparently of or above the age of 16 years at, or sending it by post to, the head office, a registered office or a principal office of the body corporate or to an address specified by the body corporate for the giving or service of documents, or	39 40 41 42
			(ii)	sending it by facsimile transmission to the facsimile number of the body corporate, or	43 44
			(iii)	sending it by electronic transmission (including for example the Internet) to the person in accordance with arrangements indicated by the person as appropriate for transmitting documents to the person.	45 46 47

(2)	Nothing in this section affects the operation of any provision of a law or of the rules of a court authorising or requiring a document to be given or served on a person in any other manner.	1 2 3
Revi	ew of legislation	4
(1)	The Minister is to review this Act and the <i>Planning Administration Act 2013</i> to determine whether the policy objectives of the Acts remain valid and whether the terms of the Acts remain appropriate for securing those objectives.	5 6 7
(2)	The review is to be undertaken as soon as possible after the period of 5 years from the commencement of this Act.	8 9
(3)	A report on the outcome of the review is to be tabled in each House of Parliament within 12 months after the end of the period of 5 years.	10 11

Sch	nedule	e 1 General—ancillary provisions	1
		(Section 1.4)	2
1.1	Dictior	nary	3
	1	In this Act:	4
		<i>affordable housing</i> means (subject to the regulations) housing for very low income households, low income households or moderate income households.	5 6
	i	amend includes alter, vary or substitute (and amend provisions or a document includes amend a map or spatial dataset adopted by or under the provisions or document).	7 8 9
		<i>area</i> , in relation to a council, has the same meaning as it has in the <i>Local Government</i> Act 1993.	10 11
	l	biodiversity offset contribution—see section 1.8.	12
	l	biodiversity offsets—see section 1.12.	13
	(brothel means a brothel within the meaning of the <i>Restricted Premises Act 1943</i> , other than premises used or likely to be used for the purposes of prostitution by no more than one prostitute.	14 15 16
	l	building includes:	17
		(a) part of a building, and	18
		(b) any temporary or permanent structure or part of any such structure,	19
		but does not include a (or part of a) manufactured home, moveable dwelling or associated structure.	20 21
	ł	building certifier means a council or the holder of a certificate of accreditation as a building certifier under the <i>Building Professionals Act 2005</i> acting in relation to matters to which the accreditation applies.	22 23 24
		<i>Building Code of Australia</i> means the official publication known as the <i>Building Code of Australia</i> , as in force in New South Wales from time to time.	25 26
		<i>building information certificate</i> means a building information certificate under Division 11.3 of Part 11.	27 28
		Building Professionals Board means the Building Professionals Board constituted under the Building Professionals Act 2005.	29 30
	l	building work—see section 1.9.	31
	(bush fire prone land means land recorded for the time being as bush fire prone land on a bush fire prone land map for the area concerned that is certified under Division 11.6 of Part 11.	32 33 34
	Ċ	certifier means a building certifier or a subdivision certifier.	35
		change of building use—see section 8.1.	36
		<i>community participation</i> —see section 2.1 (2).	37
		<i>complying development</i> means development requiring development consent declared by a local plan to be complying development.	38 39
		<i>complying development application</i> means an application for a complying development certificate.	40 41
	(2	<i>complying development certificate</i> means a consent under Part 4 to carry out complying development that is granted following a complying development application.	42 43 44
	C	consent authority—see section 4.4.	45

proh	<i>rolling</i> development or other thing includes permitting, regulating, restricting or ibiting the development or other thing, or conferring functions on a person or to do so.	
critic	<i>cal habitat</i> means critical habitat within the meaning of the threatened species lation.	
demo	olition of a building or work includes enclosing a public place in connection with lemolition of a building or work.	
deve	<i>lopment</i> —see section 1.5.	
deve	<i>lopment assessment code</i> —see section 4.16.	
	<i>lopment consent</i> means a consent under Part 4 to carry out development, and ides a complying development certificate.	
deve	<i>lopment control order</i> means an order under Division 10.2 of Part 10.	
deve the d are	<i>lopment standards</i> means provisions that relate to the carrying out of lopment and that specify requirements or standards in respect of any aspect of levelopment. The regulations may designate any requirements or standards that or are not development standards for the purposes of all or any specified isions of the planning legislation.	
decla	assessed development means development requiring development consent ared by the planning control provisions of a local plan to be EIS assessed lopment.	
	<i>ronment</i> includes all aspects of the surroundings of humans, whether affecting humans as individuals or in their social groupings.	
erect	tion of a building includes:	
(a)	the rebuilding of, the making of alterations to, or the enlargement or extension of, a building, or	
(b)	the placing or relocating of a building on land, or	
(c)	enclosing a public place in connection with the construction of a building, or	
(d)	erecting an advertising structure over a public road, or	
(e)	extending a balcony, awning, sunshade or similar structure or an essential service pipe beyond the alignment of a public road,	
gene	loes not include any act, matter or thing excluded by the regulations (either rally for the purposes of this Act or only for the purposes of specified provisions is Act).	
exen	<i>upt development</i> —see section 1.14.	
-	er Act means the Environmental Planning and Assessment Act 1979.	
	<i>tion</i> includes a power, authority or duty, and <i>exercise</i> a function includes orm a duty.	
•	way determination—see section 3.17.	
secti	<i>th infrastructure plan</i> means a growth infrastructure plan made under on 7.19.	
asses	pendent hearing and assessment panel means an independent hearing and assessment panel of a council established under Part 7 of the <i>Planning inistration Act 2013</i> .	
<i>infor</i> infor	<i>mation certificate</i> means a planning information certificate or a building mation certificate.	
infra	<i>ustructure contribution</i> —see section 1.8.	
:	ustructure plan—see section 1.8.	

land includes.

land	includes:	1
(a)	the sea and any other body of water, and	2
(b)	a building on the land.	3
local	<i>infrastructure plan</i> means a local infrastructure plan made under section 7.10.	4
	<i>sterial planning order</i> means an order made by the Minister and published on ISW planning portal.	5 6
	<i>Planning Director-General</i> means the Director-General of the Department of ning and Infrastructure.	7 8
NSW	<i>planning policy</i> means a NSW planning policy made under section 3.6.	9
www Direc	<i>planning portal</i> means the website with the URL of planningportal.nsw.gov.au, or any other website, used by the NSW Planning etor-General to provide public access to documents or other information in the planning database.	10 11 12 13
objec	<i>tor</i> —see section 9.8 (2).	14
owne	r has the same meaning it has in the Local Government Act 1993.	15
	<i>5 environmental impact assessment development</i> means relevant development s subject to environmental impact assessment under Division 5.1 of Part 5.	16 17
	<i>of shared accommodation</i> includes a boarding house, a house let in lodgings backpackers' hostel.	18 19
-	ning approval—see section 1.7.	20
	<i>ning Assessment Commission</i> means the Planning Assessment Commission ituted under Part 4 of the <i>Planning Administration Act 2013</i> .	21 22
plann	ning body—see section 1.10.	23
plann	ning control provisions—see section 1.6 (2).	24
	<i>ning Growth Fund</i> means a Planning Growth Fund established under dule 7.	25 26
	<i>ning information certificate</i> means a planning information certificate under sion 11.3 of Part 11.	27 28
plann	ning legislation means any of the following:	29
(a)	this Act and the instruments under this Act,	30
(b)	the Planning Administration Act 2013 and the instruments under that Act.	31
	<i>ning Ministerial Corporation</i> means the corporation constituted under Part 3 of <i>lanning Administration Act 2013</i> .	32 33
prem	ises includes:	34
(a)	a building, or	35
(b)	a (or part of a) manufactured home, moveable dwelling or associated structure, or	36 37
(c)	land or place (whether enclosed or built on or not), or	38
(d)	a vehicle, vessel or aircraft.	39
devel	<i>onent</i> of development means the person proposing to carry out all or any of the lopment, and includes any person determined by the NSW Planning ctor-General (or taken under this Act) to be the proponent.	40 41 42
publi	<i>c authority</i> means:	43
(a)	a public or local authority constituted by or under an Act, or	44
(b)	a NSW government agency or statutory body representing the Crown, or	45
(c)	a Public Service agency, or	46

(d)	the head of a Public Service agency (including the NSW Planning Director-General), or	1 2
(e)	a statutory State owned corporation (and its subsidiaries) within the meaning of the <i>State Owned Corporations Act 1989</i> or its chief executive officer, or	3 4
(f)	a person or body declared by the regulations to be a public authority for the purposes of all or any specified provisions of the planning legislation.	5 6
publ	<i>ic priority infrastructure</i> —see Division 5.3 of Part 5.	7
publ	ic road means a public road within the meaning of the Roads Act 1993.	8
regia	on means an area declared to be a region under clause 1.2.	9
regia	onal growth plan means a regional growth plan made under section 3.6.	10
	<i>ponal planning panel</i> means a regional planning panel constituted under Part 5 of <i>Planning Administration Act 2013</i> .	11 12
cons	<i>conally significant development</i> means development requiring development ent declared by the planning control provisions of a local plan to be regionally ificant development.	13 14 15
relev	<i>pant planning authority</i> for the preparation of a strategic plan—see section 3.1.	16
	<i>ies impact statement</i> means a species impact statement prepared in accordance the threatened species legislation.	17 18
	<i>infrastructure approval</i> means an approval for carrying out State infrastructure clopment under Division 5.2 of Part 5.	19 20
contr	<i>e infrastructure development</i> means development declared by the planning rol provisions of a local plan (or by the Minister under section 5.10) to be State astructure development—see section 5.10.	21 22 23
decla	<i>e significant development</i> means development requiring development consent ared by the planning control provisions of a local plan (or by the Minister under on 4.28) to be State significant development.	24 25 26
	<i>egic compatibility certificate</i> means a strategic compatibility certificate in force or Division 4.7 of Part 4.	27 28
strat	<i>egic plan</i> —see section 1.6 (1).	29
subd	<i>livision</i> of land—see clause 1.4.	30
subd	<i>livision certifier</i> means:	31
(a)	a council, or	32
(b)	in relation to a subdivision that the planning control provisions of a local plan authorise an accredited subdivision certifier under the <i>Building Professionals</i> <i>Act 2005</i> to act as the certifier—the holder of a certificate of accreditation as a subdivision certifier under that Act acting in relation to matters to which the accreditation applies, or	33 34 35 36 37
(c)	in relation to a subdivision carried out by the Crown or a person prescribed by the regulations—a person acting on behalf of the Crown or a person so prescribed or acting on behalf of a person so prescribed.	38 39 40
subd	livision work—see section 1.9.	41
subr	egion means an area declared to be a subregion under clause 1.2.	42
	egional delivery plan means a subregional delivery plan made under section 3.6.	43
	<i>egional planning board</i> means a subregional planning board constituted under 6 of the <i>Planning Administration Act 2013</i> .	44 45
not p	borary structure includes a booth, tent or other temporary enclosure (whether or part of the booth, tent or enclosure is permanent), and also includes a mobile eture.	46 47 48

threatened species legislation means the Threatened Species Conservation Act 1995 1 or, in the case of fish or marine vegetation, Part 7A of the Fisheries Management Act 2 1994. 3 threatened species, populations and/or ecological communities has the meaning 4 given by the threatened species legislation, but does not include a vulnerable 5 ecological community (except so much of the community as comprises a threatened 6 species or endangered population). 7 *Tier 1, Tier 2, Tier 3* in relation to an offence, indicates the maximum penalty that a 8 court may impose for the offence-see sections 10.14-10.16 for the relevant 9 maximum amounts 10 transitional planning instrument means an environmental planning instrument, or 11 deemed environmental planning instrument, under the former Act that is continued 12 in force on the commencement of this Act under Schedule 12. 13 use of land includes a change of building use. 14 wilderness area means a wilderness area within the meaning of the Wilderness Act 15 1987. 16 *work* includes any physical activity in relation to land that is specified by a regulation 17 to be a work for the purposes of this Act, but does not include a reference to any 18 activity that is specified by a regulation not to be a work for the purposes of this Act. 19 The *carrying out* of a work includes: 20 the rebuilding of, the making of alterations to, or the enlargement or extension 21 (a) of, a work, or 22 (b) enclosing a public place in connection with the carrying out of a work. 23 **Regions and subregions** 24 The Minister may, by Ministerial planning order, declare any area of the State (or 25 parts of areas of the State) to be a region or subregion for the purposes of the planning 26 legislation. 27 Interpretation Act 1987 provisions 28 The Interpretation Act 1987 contains provisions that affect the interpretation and 29 operation of this Act. 30 Note. 1. Section 21 and other provisions of Part 2 of that Act define common words and 31 expressions used in Acts and instruments eg person includes an individual, a corporation and 32 a body corporate or politic. 33 2. The provisions of the Interpretation Act 1987 that apply to instruments apply to instruments 34 under this Act and various sections applying to statutory rules also apply to instruments 35 making or amending planning control provisions-see section 5 (6) of that Act. 36 3. Section 49 of that Act contains provisions relating to the delegation of functions under the 37 planning legislation. 38 Meaning of "subdivision of land" 39 (1)For the purposes of this Act, *subdivision* of land means the division of land into two 40 or more parts that, after the division, would be obviously adapted for separate 41 occupation, use or disposition. The division may (but need not) be effected: 42 (a) by conveyance, transfer or partition, or 43 by any agreement, dealing, plan or instrument rendering different parts of the (b) 44 land available for separate occupation, use or disposition. 45 Without limiting subclause (1), *subdivision* of land includes the procuring of the (2)46 registration in the office of the Registrar-General of: 47 a plan of subdivision within the meaning of section 195 of the *Conveyancing* (a) 48 Act 1919, or 49

1.2

1.3

		(b)	Sche	ata plan or a strata plan of subdivision within the meaning of the Strata mes (Freehold Development) Act 1973 or the Strata Schemes (Leasehold elopment) Act 1986.	1 2 3
		exten	ds to p	efinition of <i>plan of subdivision</i> in section 195 of the <i>Conveyancing Act</i> 1919 lans of subdivision for lease purposes (within the meaning of section 23H of that various kinds of plan under the <i>Community Land Development Act</i> 1989.	4 5 6
	(3)	How	ever, s	ubdivision of land does not include:	7
		(a)	a lea	se (of any duration) of a building or part of a building, or	8
		(b)		ppening of a public road, or the dedication of land as a public road, by the vn, a statutory body representing the Crown or a council, or	9 10
		(c)	of an	cquisition of land, by agreement or compulsory process, under a provision a Act (including a Commonwealth Act) that authorises the acquisition of by compulsory process, or	11 12 13
		(d)		ision of land effected by means of a transaction referred to in section 23G e <i>Conveyancing Act 1919</i> , or	14 15
		(e)	the p	procuring of the registration in the office of the Registrar-General of:	16
			(i)	a plan of consolidation, a plan of identification or a miscellaneous plan within the meaning of section 195 of the <i>Conveyancing Act 1919</i> , or	17 18
			(ii)	a strata plan of consolidation or a building alteration plan within the meaning of the <i>Strata Schemes (Freehold Development) Act 1973</i> or the <i>Strata Schemes (Leasehold Development) Act 1986</i> .	19 20 21
1.5		rminat atened		f whether proposed development likely to significantly affect ies	22 23
	(1)	Act w (unle	vhethe ss this	ing is to be taken into account for the purposes of determining under this or proposed development is likely to significantly affect threatened species of clause or another provision of this Act provides that it is not likely to do s it is carried out in critical habitat):	24 25 26 27
		(a)	to ha	e case of a threatened species, whether the proposed development is likely we an adverse effect on the life cycle of the species such that a viable local lation of the species is likely to be placed at risk of extinction,	28 29 30
		(b)	likely the e	e case of an endangered population, whether the proposed development is y to have an adverse effect on the life cycle of the species that constitutes ndangered population such that a viable local population of the species is y to be placed at risk of extinction,	31 32 33 34
		(c)		e case of an endangered ecological community or critically endangered ogical community, whether the proposed development:	35 36
			(i)	is likely to have an adverse effect on the extent of the ecological community such that its local occurrence is likely to be placed at risk of extinction, or	37 38 39
			(ii)	is likely to substantially and adversely modify the composition of the ecological community such that its local occurrence is likely to be placed at risk of extinction,	40 41 42
		(d)		elation to the habitat of a threatened species, population or ecological munity:	43 44
			(i)	the extent to which habitat is likely to be removed or modified as a result of the proposed development, and	45 46
			(ii)	whether an area of habitat is likely to become fragmented or isolated from other areas of habitat as a result of the proposed development, and	47 48

		(iii)	the importance of the habitat to be removed, modified, fragmented or isolated to the long-term survival of the species, population or ecological community in the locality,	1 2 3		
	(e)		her the proposed development is likely to have an adverse effect on eal habitat (either directly or indirectly),	4 5		
	(f)	whether the proposed development is consistent with the objectives or action of a recovery plan or threat abatement plan,				
	(g)	proce	her the proposed development constitutes or is part of a key threatening ess or is likely to result in the operation of, or increase the impact of, a key itening process.	8 9 10		
	Cons of the	<i>ervati</i> e Fish	sment guidelines under section 94A of the <i>Threatened Species</i> on Act 1995 or, in the case of fish and marine vegetation, section 220ZZA <i>eries Management Act 1994</i> apply to the determination of whether any sed development is likely to significantly affect threatened species.	11 12 13 14		
(2)	Proposed development is not likely to significantly affect threatened species for the purposes of this Act if:					
	(a)	asses claus threa	roposed development is not likely to do so as determined by a biodiversity ssment procedure adopted by the regulations for the purposes of this se (being a procedure approved by the Ministers administering the tened species legislation and set out in the regulations or published in the ette), and	17 18 19 20 21		
	(b)		proposed development is of a kind that the planning control provisions of ocal plan declare may be assessed in accordance with that procedure.	22 23		
	Note. of this		anning control provisions are to apply to development subject to either Part 4 or 5	24 25		
(3)	If a determination under subclause (2) that proposed development is not likely to significantly affect threatened species is based on measures to avoid or mitigate the impacts of the proposed development or on measures for biodiversity offsets, the terms and conditions of any planning approval for the development are to reflect those measures or measures having substantially the same effect.					
	Note.	e. Development is taken not to significantly affect threatened species if:		31		
	(a)	speci	evelopment is to be carried out on biodiversity certified land under the threatened es legislation, or	32 33		
	(b)	a biol of the	banking statement has been issued in respect of the development under Part 7A a Threatened Species Conservation Act 1995.	34 35		
Misc	ellane	ous ir	nterpretation provisions	36		
(1)	When coun		ctions are conferred or imposed under the planning legislation on a	37 38		
	(a)	those area,	e functions may be exercised in respect of an area by the council of that or	39 40		
	(b)		e functions are conferred or imposed in respect of part of an area, those tions may be exercised in respect of that part by the council of that area.	41 42		
(2)	Where functions are conferred or imposed under the planning legislation on a public authority that is a Public Service agency or other unincorporated group of persons, those functions may be exercised by a person who is authorised to exercise those functions on behalf of the public authority.					
(3)	A reference in this Act (or in an instrument made under this Act) to an order or determination is (if the order or determination is not required to be published on the NSW planning portal) a reference to an order or determination published on the NSW planning portal or (if not so published) in writing.					

	(4)	A power, express or implied, under the planning legislation to make an order or determination includes a power to revoke or amend the order or determination.	1 2				
	(5)	A reference in this Act to any act, matter or thing as specified in an instrument under this Act includes a reference to any act, matter or thing that is of a class as specified in such an instrument.	3 4 5				
	(6)	A reference in this Act to the granting of consent or approval includes a reference to the granting of consent or approval subject to conditions.	6 7				
	(7)	Without affecting the generality of section 8 (b) of the <i>Interpretation Act 1987</i> , a reference in this Act to the owner or lessee of land includes a reference to joint or multiple owners or lessees of land.	8 9 10				
1.7	Reso	Resolving inconsistencies in declarations of categories of development					
	(1)	Development declared by a local plan to be exempt development is not exempt development if the planning control provisions of the local plan prohibit that development or provide that the development cannot be carried out without development consent.	12 13 14 15				
	(2)	Development does not require development consent under the planning control provisions of a local plan if other planning control provisions that prevail in the event of inconsistency prohibit that development being carried out or permit that development to be carried out without development consent.	16 17 18 19				
	(3)	Development is not prohibited by the planning control provisions of a local plan if other planning control provisions that prevail in the event of inconsistency permit that development to be carried out with or without development consent.	20 21 22				
1.8	Note	Notes					
		Notes in this Act do not form part of this Act.	24				

Schedule 2		Community participation—ancillary provisions	1
		(Section 2.5)	2
Par	t 1 Ma	ndatory community participation requirements	3
Divi	sion 1	Minimum public exhibition periods for strategic and other plans	4 5
2.1	Draft strate	egic plans (other than local plans)	6
	28 da	ays.	7
2.2		structure plans	8
	28 da	ays.	9
2.3		nunity participation plans	10
	28 da	ays.	11
2.4	Draft amer	ndment to any of the above plans	12
		period (if any) determined by the planning authority in accordance with the rant community participation plan.	13 14
2.5	gateway d	proposal for planning control or other provisions of local plans subject to etermination (including any neighbourhood impact statement required to y the proposal)	15 16 17
	The	period specified in the gateway determination for the proposal.	18
2.6	gateway d	ning control, code or guide provisions of local plans not subject to etermination (including any neighbourhood impact statement required to y the draft)	19 20 21
	28 da	ays.	22
2.7	Draft strate	egic context provisions or contribution provisions of local plans	23
		period (if any) determined by the planning authority in accordance with the vant community participation plan.	24 25
Divi	sion 2	Minimum public exhibition periods for development applications and other matters	26 27
2.8		n for issue of strategic compatibility certificate and application for ent consent made in reliance on a strategic compatibility certificate	28 29
	28 da	ays in each case.	30
2.9		n for development consent for development that is subject to merit nt (other than for EIS assessed development or for State significant ent)	31 32 33
	14 da	ays.	34
2.10	Applicatio	n for development consent for EIS assessed development	35
	28 da	ays.	36

2.11		lication for development consent for State significant development that is ect to merit assessment	1 2
		28 days.	3
2.12	App exhi	lication for modification of development consent that is required to be publicly bited by the regulations	4 5
		The period (if any) determined by the consent authority in accordance with the relevant community participation plan.	6 7
2.13		ronmental impact statement (or species impact statement) obtained under sion 5.1 of Part 5	8 9
		28 days.	10
2.14		ronmental impact statement for State infrastructure development under sion 5.2 of Part 5	11 12
		28 days.	13
2.15	Re-e unde	exhibition of any amended application or matter referred to above required by or er this Schedule	14 15
		The period (if any) determined by the person or body responsible for publicly exhibiting the application or matter.	16 17
Divi	sion	3 Provisions relating to public exhibition	18
2.16		licly exhibited plans, applications etc not to be made or determined until after bition period	19 20
	(1)	If this Part requires a plan, application or other matter to be publicly exhibited, the plan or application is not to be made or determined (or the other matter finalised) until after the minimum period of public exhibition under this Part.	21 22 23
	(2)	If the plan, application or other matter is placed on public exhibition for a specified longer period, the plan or application is not to be made or determined (or the other matter finalised) until after that specified longer period.	24 25 26
2.17	Sub	missions during exhibition period	27
	(1)	Submissions with respect to a plan, application or other matter may be made during the minimum period of its public exhibition under this Part.	28 29
	(2)	If the plan, application or other matter is placed on public exhibition for a specified longer period, submissions may be made during that specified longer period.	30 31
2.18	Excl	usion of Christmas/New Year period	32
		The period between 20 December and 10 January (inclusive) is excluded from the calculation of a period of public exhibition. Note. See also section 36 (2) of the <i>Interpretation Act 1987</i> for applicable rule where an exhibition period includes a weekend or public holiday.	33 34 35
2.19	Dulo		36
2.19	Rule	where more than one exhibition period applies If a particular matter has different exhibition or notification periods that apply under	37 38
		this Part, the longer period applies.	39
2.20	Prov	rision relating to public exhibition of EIS	40
		A public authority is not required to make available for public inspection any part of an environmental impact statement whose publication would, in the opinion of the	41 42

			ic authority, be contrary to the public interest because of its confidential nature r any other reason.	1 2
Divi	ision	4	Mandatory notification requirements for applications and decisions	3 4
2.21	Deve	elopm	ent and other applications and decisions—general	5
		unde this A	mandatory notification requirements of development and other applications or this Act and of the making of decisions with respect to those applications under Act are the requirements prescribed by this Part, or the requirements prescribed he regulations, as mandatory notification requirements.	6 7 8 9
2.22	Publ	ic not	ification of certain decisions and reasons for the decisions	10
	(1)	This	clause applies to the following decisions:	11
		(a)	the making of a strategic plan (or provisions of a strategic plan) by the Minister or a council (other than in reliance on section 3.27),	12 13
			Note. Section 3.27 makes provision for the publication of reasons.	14
		(b)	the making of an infrastructure plan by the Minister,	15
		(c)	the determination by the Minister of an application for State infrastructure development,	16 17
		(d)	the determination by the Minister of a request for a modification of a State infrastructure approval (being a request that was publicly exhibited),	18 19
		(e)	the determination by a regional planning panel or by the NSW Planning Director-General of an application for a strategic compatibility certificate,	20 21
		(f)	the determination by a consent authority of an application for development consent under section 4.18 (Merit assessment),	22 23
		(g)	the determination by a consent authority of an application for the modification of a development consent (being an application that was publicly exhibited),	24 25
		(h)	the making of a community participation plan by a planning authority,	26
		(i)	the granting of an approval, or the decision to carry out development, by a determining authority where an environmental impact statement was publicly exhibited under Division 5.1 of Part 5.	27 28 29
	(2)		mandatory notification requirement in relation to a decision to which this clause less is public notification of:	30 31
		(a)	the decision, and	32
		(b)	the date of the decision, and	33
		(c)	the reasons for the decision (having regard to any statutory requirements applying to the decision), and	34 35
		(d)	how community views were taken into account in making the decision.	36
Divi	ision	5	Additional mandatory community participation requirements	37 38
2.23	Rea	uireme	ents prescribed by regulation	39
		The	regulations may prescribe additional mandatory requirements for community cipation.	40 41

Part 2		General provisions		1
2.24	Regu	ulation	ns relating to public exhibition	2
	(1)		regulations may set out the method of public exhibition under this Act, how le can make submissions and how people can obtain further information.	3 4
	(2)		regulations may specify the requirements for something to be considered a hission for the purposes of this Act.	5 6
2.25	Re-e	xhibiti	ion	7
	(1)		regulations may specify the circumstances in which a plan or other matter is ired or not required to be re-exhibited.	8 9
	(2)		xhibition is not required if the environmental impact of the development has reduced or not increased.	10 11
2.26	Сору	right	in documents used for purposes of planning legislation	12
	(1)	planr purp	levant person who is not entitled to copyright in a document that is part of a ning matter is taken to have indemnified all persons using the document for the oses of the planning legislation against any claim or action in respect of a breach pyright in the document.	13 14 15 16
	(2)	docu other purpo perso matte	regulations may require a relevant person who is entitled to copyright in a ment that is part of a planning matter to give (in the planning matter or rwise) a licence to the State or a council to use the copyright material for the oses of the planning legislation. The regulations may also require a relevant on who is not so entitled to that copyright to give a warranty (in the planning er or otherwise) that the relevant person has a licence to so use the copyright rial from the person who is entitled to copyright in any such document.	17 18 19 20 21 22 23
	(3)	For t	he purposes of this clause:	24
		(a)	an application for a planning approval (or to modify a planning approval) under this Act or the former Act is a planning matter, and the applicant is the relevant person, and	25 26 27
		(b)	an environmental impact statement or a statement of environmental effects under this Act or the former Act (including any preferred infrastructure report) is a planning matter, and the proponent of the development is the relevant person, and	28 29 30 31
		(c)	a planning proposal under Part 3 (or under the former Act) is a planning matter, and the person preparing the proposal is the relevant person, and	32 33
		(d)	a planning agreement under this Act or the former Act is a planning matter, and the developer under the agreement is the relevant person, and	34 35
		(e)	a matter or thing under the planning legislation that is declared by the regulations for the purposes of this clause is a planning matter, and the person declared by the regulations is the relevant person in respect of that matter or thing.	36 37 38 39
	(4)	of or perso with	he purposes of this clause, a document is part of a planning matter if it forms part accompanies the planning matter, or is subsequently submitted by the relevant on in support of the planning matter or is exhibited or made public in accordance a requirement made by or under the planning legislation in relation to the ning matter.	40 41 42 43 44
	(5)	The 1	regulations may limit the operation of this clause.	45
	(6)	This	clause extends to planning matters in paper or electronic form.	46

	(7)	This clause extends to a planning matter that was made or submitted before the commencement of this clause.	1 2
Par	t 3	Online delivery of planning services and information	3
2.27	Esta	blishment, content and maintenance of NSW planning database	4
	(1)	The NSW planning database is established for the purposes of the planning legislation.	5 6
	(2)	The NSW planning database is an electronic repository of:	7
		(a) documents that are required under the planning legislation to be published on the NSW planning portal, and	8 9
		(b) the planning control provisions of local plans or other documents that are required under the planning legislation to be published on the NSW legislation website, and	10 11 12
		(c) maps or spatial datasets that are adopted or incorporated by way of reference by those provisions or documents, and	13 14
		(d) other documents or information relating to the administration of the planning legislation required to be published on the NSW planning portal by the regulations or by the NSW Planning Director-General.	15 16 17
		Note. Matters required to be published on the NSW planning portal include Ministerial planning orders, instruments of delegation, community participation plans, NSW planning policies, regional growth plans, subregional delivery plans, various provisions of local plans, infrastructure plans, declarations of public priority infrastructure.	18 19 20 21
	(3)	The NSW planning database is to maintain historical as well as current versions of documents and other material required to be published on the NSW planning portal.	22 23
	(4)	The NSW planning database is to be compiled and maintained as determined by the NSW Planning Director-General.	24 25
	(5)	Any part of the NSW planning database that comprises the planning control provisions of local plans or other material published on the NSW legislation website may be compiled and maintained in the legislation database that is compiled and maintained by the Parliamentary Counsel for publication on that website.	26 27 28 29
2.28	Publ	ic access to documents and information on the NSW planning portal	30
	(1)	The NSW Planning Director-General is to make arrangements for documents or other material in the NSW planning database to be published on the NSW planning portal and such other websites as are determined by the NSW Planning Director-General.	31 32 33 34
	(2)	The NSW Planning Director-General may certify the form of such documents or other information that is correct.	35 36
	(3)	Planning control provisions of a local plan or other documents and information need not be published on the NSW planning portal if they are published on the NSW legislation website and can be readily accessed from the NSW planning portal.	37 38 39
	(4)	If the NSW planning portal is not available to publish a document or other information for technical or other reasons, the document or other information may be published on the NSW legislation website.	40 41 42
		Note. The NSW planning portal is defined by Schedule 1 to mean the website with the URL of www.planningportal.nsw.gov.au, or any other website, used by the NSW Planning Director-General to provide public access to documents or other information in the NSW planning database.	43 44 45 46

2.29 Regulations and other provisions relating to online planning services and information

- (1) The regulations may make provision for or with respect to the online delivery of planning services and information, including:
 - (a) the NSW planning portal and other specialised planning portals (including the status of services and information delivered online), and

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

23

- (b) access to information (and issue of certificates) about land use zoning and development standards or guides relating to particular land, and
- (c) the lodgment or submission of applications and other things under the planning legislation, and
- (d) the assessment of categories of development for which there are codified criteria or standards, and
- (e) the registration of certificates or other planning approvals (or other documents) and their effect on registration, and
- (f) the issue or grant of certificates or other planning approvals by means of the NSW planning portal.
- (2) For the purpose of facilitating online delivery of planning services and information:
 - (a) the NSW Planning Director-General may determine standard technical 18 requirements with respect to: 19
 - (i) the preparation of strategic and other plans, or other documents and of any maps or spatial datasets that are referred to in (or adopted under)
 20
 21
 22
 - (ii) the form of applications for planning approvals and planning approvals, and
 - (b) a council or other planning body is to provide the NSW Planning
 Director-General, when requested, with electronic files (in a specified format)
 of any such plans, other documents, maps or spatial datasets prepared or held
 by it, and
 - (c) a council or other planning body is to implement any standard technical requirements determined by the NSW Planning Director-General to facilitate access to relevant data in the electronic systems maintained by the council or other body or to transfer that data to the NSW planning database.

Schedule 3 Strategic planning—ancillary provisions

3.1 Regulations: Part 3

The regulations may make provision with respect to strategic plans, including:

(a) the appointment and functions of relevant planning authorities under Part 3 of this Act, and

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

- (b) the procedure for the preparation and making of strategic plans or provisions of those plans, and
- (c) the form, structure and content of the provisions of a local plan (including the standardisation of those provisions), and
- (d) requirements with respect to planning proposals for proposed planning control or other provisions and the submission of other related reports and documents, and
- (e) special consultation requirements with respect to planning proposals for proposed planning control or other provisions, and
- (f) requirements for referrals to or concurrences of public authorities in relation to specified provisions of a planning proposal, and
- (g) special requirements with respect to the exercise of the functions of consent authorities where proposed development may affect the Sydney drinking water catchment.
 17
 18
 19

Schedule 4 Development (other than infrastructure) assessment and consent—ancillary provisions

3 A consent authority may reduce the period of 5 years when a development consent 4 lapses as referred to in section 4.46. This subclause does not apply to development 5 consent granted to a staged development application for development that requires a 6 7 8 a development consent to erect or demolish a building or to subdivide land to 9 10

11

12

13

14

15

16

17

18

19

20

21

22

31

38

39

40

41

1

2

- (b) a development consent of a kind prescribed by the regulations to lapse within the period prescribed by the regulations in relation to the consent.
- (3)The period of 5 years (as referred to in section 4.46) when a development consent lapses is reduced to 2 years if the consent was granted in reliance on a strategic compatibility certificate.

lapse within 2 years after the date from which the consent operates, or

- (4)Development consent for:
 - the erection of a building, or (a)
 - (b) the subdivision of land, or

subsequent application and consent.

Such a reduction cannot be made so as to cause:

(c) the carrying out of a work,

does not lapse if building or subdivision work is physically commenced on the land to which the consent applies before the date on which the consent would otherwise lapse under this clause.

- (5) Development consent for development other than that referred to in subclause (4) 23 does not lapse if the use of any land, building or work the subject of that consent is 24 actually commenced before the date on which the consent would otherwise lapse. 25
- (6)Despite any other provision of this clause, a development consent that is subject to a 26 deferred commencement condition lapses if the applicant fails to satisfy the consent 27 authority as to the matter specified in the condition within 5 years from the grant of 28 the consent or, if a shorter period is specified by the consent authority, within the 29 period so specified. 30

4.2 Extension of lapsing period for 1 year

- If, in granting a development consent, the consent authority reduces the period after (1)32 which the consent lapses to less than 5 years, the applicant or any other person who 33 has the benefit of the consent may apply to the consent authority, before the period 34 expires, for an extension of 1 year. 35
- The consent authority may grant the extension if satisfied that the applicant has (2)36 shown good cause. 37
- (3)This clause does not apply to a complying development certificate.

4.3 Voluntary surrender of development consents

- (1)A development consent may be surrendered, subject to and in accordance with the regulations, by any person entitled to act on the consent.
- A development consent may be surrendered even if, on an appeal against the granting (2)42 of the consent, the consent does not have effect. 43

4.1

(1)

(2)

Lapsing of consent

(a)

4.4			t determination of applications for approval of activities under Local nt Act 1993	1 2
	(1)	on be for de	e consent authority is a council, a person (other than the Crown or a person acting ehalf of the Crown) may, in the same application for development consent, apply evelopment consent and approval for anything that requires approval under the wing provisions of the Table to section 68 of the <i>Local Government Act 1993</i> , ely:	3 4 5 6 7
		parag	graph 1 of Part A,	8
		parag	graph 1–6 of Part B,	9
		parag	graph 1–5 of Part C,	10
		parag	graph 1 of Part E,	11
		parag	graph 1–5 or 10 of Part F.	12
	(2)	any c if the relev impo	termining any such application for development consent, the council may apply of the provisions of or under the <i>Local Government Act 1993</i> that it could apply application for development consent were an application under that Act for the ant approval. In particular, if development consent is granted, the council may use a condition that is authorised under that Act to be imposed as a condition of oproval.	13 14 15 16 17 18
	(3)	have autho	velopment consent is granted on any such application, the council is taken to granted the relevant approval under the <i>Local Government Act 1993</i> that prises the activity, but that Act has no application to the approval so taken to been granted.	19 20 21 22
	(4)	limiti to ha	anting development consent on any such application, the council may, without ing any other condition it may impose, impose, in relation to the approval taken we been granted under the <i>Local Government Act 1993</i> , either or both of the wing conditions:	23 24 25 26
		(a)	a condition that the approval is granted only to the applicant and does not attach to or run with the land to which it applies,	27 28
		(b)	a condition that the approval is granted for a specified time.	29
4.5	Cons	sents r	required under Wilderness Act 1987	30
		withi	pplication for development consent cannot be made in respect of land that is n a wilderness area unless any consent to the development required under the <i>erness Act 1987</i> has been obtained.	31 32 33
4.6	Biob	anking	g—special provisions	34
	(1)	condi of the and c whet	Minister may grant consent to State significant development subject to a ition that requires the applicant to acquire and retire (in accordance with Part 7A e <i>Threatened Species Conservation Act 1995</i>) biodiversity credits of a number class (if any) specified by the Minister in the consent. This subclause applies her or not a biobanking statement under Part 7A of that Act was obtained in ect of the development.	35 36 37 38 39 40
	(2)	The I	Minister may approve an arrangement under which:	41
		(a)	the retirement of some or all of the biodiversity credits is deferred pending the completion of any rehabilitation or restoration action proposed to be taken on the site of the State significant development, after the development has been substantially completed, that will restore or improve the biodiversity values affected by the development, and	42 43 44 45 46

	(b)	the biodiversity credits the retirement of which is deferred pending the completion of those actions are required to be transferred to the Minister administering the <i>Threatened Species Conservation Act 1995</i> .	1 2 3
(3)	respe	ion 7 of Part 7A of the <i>Threatened Species Conservation Act 1995</i> applies in ct of any such arrangement as if it were a deferred retirement arrangement wed under that Division.	4 5 6
(4)	the N require Note. biodive mainta	iobanking statement was obtained in respect of State significant development, dinister may grant consent to the development subject to a condition that res the applicant to comply with any conditions of the biobanking statement. The conditions of a biobanking statement may require the proponent to retire ersity credits in respect of the State significant development in order to ensure that it ains or improves biodiversity values, or to carry out other on-site measures to minimise egative impact of the development on biodiversity values.	7 8 9 10 11 12 13
(5)		son cannot appeal to the Land and Environment Court in respect of a condition sed by the Minister under subclause (4).	14 15
Orde	rs by t	he Court of conditional validity for certain development consents	16
(1)	grante from (irres) invali	clause applies to a development consent granted, or purporting to have been ed, under this Act or the former Act. The clause extends to invalidity arising any steps preliminary to the granting of any such development consent pective of who was required to take those steps) and, in particular, extends to dity arising from non-compliance with requirements declared by this Act to be atory requirements.	17 18 19 20 21 22
(2)		Land and Environment Court may, instead of declaring or determining that a opment consent is invalid, whether in whole or in part, make an order:	23 24
	(a)	suspending the operation of the consent in whole or in part, and	25
	(b)	specifying terms compliance with which will validate the consent (whether without alterations or on being regranted with alterations).	26 27
	Any s	such terms may include (without limitation):	28
	(a)	terms requiring the carrying out again of steps already carried out, or	29
	(b)	terms requiring the carrying out of steps not already commenced or carried out, or	30 31
	(c)	terms requiring acts, matters or things to be done or omitted that are different from acts, matters or things required to be done or omitted by or under this Act or any other Act.	32 33 34
(3)	groun comp	oplication by the consent authority for an order under this subclause on the ids that the terms specified under subclause (2) have been substantially lied with and that it is not proposed that the relevant development consent be need with alterations, the Land and Environment Court may make an order:	35 36 37 38
	(a)	declaring that the terms have been substantially complied with, and	39
	(b)	declaring that the consent is valid, and	40
	(c)	revoking the order of suspension.	41

- (4) On application by the consent authority for an order under this subclause on the ground that the terms specified under subclause (2) have been substantially complied with and that the development consent has been regranted with alterations as referred to in clause 4.8, the Land and Environment Court may make an order: 45
 - (a) declaring that the terms have been complied with, and 46

(b) declaring that the development consent has been validly regranted, and

	(c)	declaring that the suspended development consent has been revoked, and	1
	(d)	revoking the order of suspension.	2
(5)	A de	velopment consent declared to be valid or validly regranted under this clause:	3
	(a)	is final and the provisions of sections 9.7 and 9.8 of this Act do not apply to or in respect of it, and	4 5
	(b)	in the case of a consent declared to be valid—is operative as from the date the development consent originally took effect or purported to take effect, unless the Court otherwise orders, and	6 7 8
	(c)	in the case of a consent declared to be validly regranted—takes effect from the date of the declaration or another date specified by the Court.	9 10
(6)	to th	ing in this clause prevents the grant of another development consent in relation ne same matter, during or after the period of suspension, pursuant to an cation for development consent duly made.	11 12 13
(7)	this c	the duty of the Land and Environment Court to consider making an order under clause instead of declaring or determining that a development consent to which clause applies is invalid, whether in whole or in part.	14 15 16
Revo	ocation	n or regrant of development consents after order of Court	17
(1)		clause applies to a development consent granted, or purporting to be granted, by asent authority, to which an order of suspension applies under clause 4.7.	18 19
(2)	appli	consent authority may revoke a development consent to which this clause es, whether or not the terms imposed by the Land and Environment Court under the 4.7 have been complied with.	20 21 22
(3)	subst conse altera regar with	ever, if the terms imposed by the Land and Environment Court have been cantially complied with, the consent authority may revoke the development ent to which this clause applies and grant a new development consent with such ations to the revoked consent as the consent authority thinks appropriate having d to the terms themselves and to any matters arising in the course of complying the terms. Such a grant of a development consent is referred to as a <i>regrant</i> of onsent.	23 24 25 26 27 28 29
(4)	conse	breliminary steps need be taken with regard to the regrant of a development ent under this clause, other than those that are required to secure compliance those terms.	30 31 32
(5)		isions of this Act as may be prescribed by the regulations apply to development ents regranted under this clause.	33 34
Regu	ulation	ns—Part 4	35
	The 1	regulations may make provision for or with respect to the following:	36
	(a)	any matter that is to be done before making an application for development consent,	37 38
	(b)	the persons who may make applications for development consent,	39
	(c)	the making, consideration and determination of applications for development consent that are made by or on behalf of the Crown, public authorities and persons prescribed by the regulations,	40 41 42
	(d)	requiring the New South Wales Aboriginal Land Council to consent to applications for the modification of development consents relating to land owned by Local Aboriginal Land Councils,	43 44 45
	(e)	the form of applications for development consent,	46

(f)	the information about applications for development consents and about development consents that is to be made publicly available,	1 2
(g)	applications for strategic compatibility certificates and the determination of those applications,	3 4
(h)	requirements concerning the issue of complying development certificates (including the notification of the issue of those certificates),	5 6
(i)	the form of complying development certificates and variation certificates,	7
(j)	the documents and information required to accompany applications for development consent, including documents that will assist the consent authority in assessing the environmental impact of development,	8 9 10
(k)	the preparation, contents, form and submission of environmental impact statements and statements of environmental effects,	11 12
(1)	the documents and information required to accompany statements of environmental effects and environmental impact statements,	13 14
(m)	additional matters of a procedural nature that are to be complied with before an application for development consent may be determined,	15 16
(n)	the modification or surrender of a development consent, or of an existing use right, in accordance with a condition of a development consent,	17 18
(0)	conditions to be imposed on development consents,	19
(p)	procedures relating to the determination of matters required to be determined or approved under a condition of a development consent,	20 21
(q)	the kinds of development that may be subject to a condition permitting the review of another condition, the matters that must be included in such a condition and the procedures for a review under such a condition,	22 23 24
(r)	procedures relating to the referral of Crown development applications to the NSW Planning Director-General and the determination of such applications,	25 26
(s)	the modification of development consents (including with respect to environmental assessment requirements),	27 28
(t)	the circumstances in which work is taken to be, or not taken to be, physically commenced under a development consent,	29 30
(u)	the periods within which specified aspects of the environmental planning control process must be completed and the variation of those periods,	31 32
(v)	the provision by councils of certificates relating to the application of development guides to particular land or in particular circumstances,	33 34
(w)	the effect of a failure to comply with any requirement of the regulations.	35

Scł	chedule 5 Infrastructure and environmental impact 1 assessment—ancillary provisions 2		
Par	rt 1	Provisions relating to environmental impact assessment under Division 5.1 of Part 5	3 4
5.1	Appl	lication	5
		This Part applies in relation to the environmental impact assessment requirements of Division 5.1 of Part 5 (referred to in this Part as <i>Part 5 environmental impact assessment</i>).	6 7 8
5.2	Nom	ination of determining authority	9
	(1)	If the approval of more than one determining authority is required in relation to any relevant development (either in respect of the carrying out of the development or the granting of an approval in respect of the development), the Minister may, by Ministerial planning order, nominate a determining authority to be the nominated determining authority for the relevant development, or class of relevant development, for the purposes of Part 5 environmental impact assessment.	10 11 12 13 14 15
	(2)	If there is a nominated determining authority for relevant development, that nominated authority may exercise the following functions on behalf of the other determining authorities in relation to that relevant development:	16 17 18
		(a) furnishing a copy of the environmental impact statement to the NSW Planning Director-General,	19 20
		(b) publicly exhibiting the environmental impact statement and receiving and forwarding submissions made in response to the public exhibition.	21 22
	(3)	A determining authority (other than the nominated determining authority) is required to forward to the nominated determining authority a copy of any submissions made to it in response to the public exhibition and provide other information to the nominated determining authority, as required by the regulations, to enable the nominated determining authority to co-ordinate the preparation and furnishing of reports in relation to the relevant development.	23 24 25 26 27 28
	(4)	If there is a nominated determining authority for any relevant development, any other determining authority for the development need not inquire into whether the nominated determining authority has complied with the functions that it exercises under this clause on behalf of that other determining authority.	29 30 31 32
5.3	Dete	rmining authorities taken to be proponents of relevant development	33
	(1)	For the purposes of Part 5 environmental impact assessment, a proponent of development is taken to include the following:	34 35
		(a) the Forestry Corporation of New South Wales in respect of forestry activities authorised by that Corporation on land under the management of that Corporation,	36 37 38
		(b) any determining authority which the Minister determines in writing to be the proponent of a particular relevant development or which the regulations declare to be the proponent of relevant development of the kind specified in the regulations.	39 40 41 42
	(2)	In any such case, a reference in Division 5.1 of Part 5 or in this Part to a determining authority carrying out development includes a reference to the Forestry Corporation or such a determining authority granting an approval for the development.	43 44 45

5.4 Particular matters to which determining authorities to have regard for purposes of Part 5 environmental impact assessment etc

- (1) For the purposes of Part 5 environmental impact assessment, a determining authority is to have regard to the effect of the relevant development on the following (to the extent that it is relevant to the assessment):
 - (a) any wilderness area,
 - (b) any critical habitat,
 - (c) any recovery plan or threat abatement plan for the purposes of assessing in a species impact statement any effect on a threatened species, population or ecological community, or its habitat,
 - (d) any conservation agreement entered into under the *National Parks and Wildlife Act 1974* that applies to any of the land concerned (and any plan of management under that Act for the land subject to the conservation agreement),
 - (e) any joint management agreement entered into under the threatened species legislation that applies to any of the land concerned,
 - (f) any biobanking agreement entered into under Part 7A of the *Threatened* 17 Species Conservation Act 1995 that applies to any of the land concerned. 18
- (2) A determining authority is not to grant an approval for any relevant development of land within a wilderness area unless any consent to the development required under the *Wilderness Act 1987* has been obtained.
 (2) A determining authority is not to grant an approval for any relevant development of the under the *Wilderness Act 1987* has been obtained.

5.5 Regulations

The regulations may make provision for or with respect to Part 5 environmental impact assessment, including for or with respect to the following:

- (a) the matters to be taken into account when consideration is being given to the likely effect of any relevant development on the environment,
- (b) the circumstances in which a determining authority is taken to have complied with its obligation under section 5.3 (General duty to consider environmental impact of relevant development),
- (c) the preparation, contents, form and submission of environmental impact statements,
- (d) the examination of environmental impact statements,
- (e) the effect of amendments to the lists of threatened species, populations and ecological communities during Part 5 environmental impact assessment.

Part 2 Provisions relating to State infrastructure development under Division 5.2 of Part 5

36 37

38

39

40

35

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

22

23

24

25

26

30

31

32

33

34

This Part applies in relation to State infrastructure development and approvals under Division 5.2 of Part 5.

5.7 Biobanking—special provisions

Application

 The Minister may approve State infrastructure development subject to a condition that requires the proponent to acquire and retire (in accordance with Part 7A of the *Threatened Species Conservation Act 1995*) biodiversity credits of a number and class (if any) specified by the Minister in the approval. This subclause applies

whether or not a biobanking statement under Part 7A of that Act was obtained in respect of the development.

2 3 4

5

6

7

8

9

10

11

12

13

14

22

23

24

33

34

35

36

37

38

39

49

1

- (2) The Minister may approve an arrangement under which:
 - (a) the retirement of some or all of the biodiversity credits is deferred pending the completion of any rehabilitation or restoration action proposed to be taken on the site of the State infrastructure development, after the development has been substantially completed, that will restore or improve the biodiversity values affected by the development, and
 - (b) the biodiversity credits the retirement of which is deferred pending the completion of those actions are required to be transferred to the Minister administering the *Threatened Species Conservation Act 1995*.
- (3) Division 7 of Part 7A of the *Threatened Species Conservation Act 1995* applies in respect of any such arrangement as if it were a deferred retirement arrangement approved under that Division.
- (4) If a biobanking statement was obtained in respect of State infrastructure development, the Minister may approve the development subject to a condition that requires the proponent to comply with any conditions of the biobanking statement.
 Note. The conditions of a biobanking statement may require the proponent to retire biodiversity credits in respect of the State infrastructure development in order to ensure that it maintains or improves biodiversity values, or to carry out other onsite measures to minimise any negative impact of the development on biodiversity values.
- (5) A person cannot appeal to the Land and Environment Court in respect of a condition imposed by the Minister under subclause (4).

5.8 Regulations for purposes of Division 5.2 of Part 5

The regulations may make provision for or with respect to State infrastructure development and approvals of State infrastructure development under Division 5.2 of Part 5, including the following: 27

- (a) the requirements and procedures for making applications for approvals under that Division, 28
- (b) requiring owners of land on which State infrastructure development is proposed to be carried out to consent to applications for approvals under that Division,
 30
 31
 32
- (c) the amendment of applications for approvals under that Division,
- (d) the preparation, notification and modification of requirements for environmental impact assessment of State infrastructure development,
- (e) the requirements for environmental impact statements under that Division,
- (f) the assessment of applications for approval and the making of reports under that Division (including the time within which any such assessment or report is to be completed),
- (g) requiring the New South Wales Aboriginal Land Council to consent to applications for approvals under that Division on land owned by Local Aboriginal Land Councils, if the consent of the Local Aboriginal Land Council concerned is required as owner of the land,
 40
 41
 42
 43
- (h) providing for public exhibition, notification and public registers of applications for approvals under that Division (or for the modification of 45 approvals) and of the determination of those applications, 46
- (i) the effect of the revocation of the declaration of development as State 47 infrastructure development, 48
- (j) the application of provisions of Part 8 to State infrastructure development.

Part 3 Provisions relating to public priority infrastructure

5.9 Application

This Part applies in relation to public priority infrastructure under Division 5.3 of Part 5.

1

2

3

4

5

6

7

8

9

10

5.10 Regulations

The regulations may make provision for or with respect to public priority infrastructure, including the following:

- (a) the declaration of public priority infrastructure,
- (b) project definition reports (including the modification of any such report at the request of the proponent),
- (c) the carrying out of development for the purposes of public priority 11 infrastructure, 12
- (d) the application (with or without modification) of provisions of Part 8 to public 13 priority infrastructure, 14
- (e) any matter for or with respect to which regulations may be made in relation to
 15
 State infrastructure development.
 16

Schedule 6 Concurrences, consultation and other legislative approvals—ancillary provisions

6.1	One	e stop referrals and decisions—heritage or bush fire approvals	3
	(1)	This clause applies to the following (a <i>heritage or bush fire approval</i>):	4
		(a) an approval under Subdivision 1 of Division 3 of Part 4 of the <i>Heritage Ac</i> 1977,	t 5 6
		(b) a bush fire safety authority under section 100B of the <i>Rural Fires Act 1997</i> .	7
	(2)	The NSW Planning Director-General is required to act in accordance with the advice of the approval body for a heritage or bush fire approval when determining under section 6.12 whether the approval body should give the heritage or bush fire approva and, if so, the general terms of any such approval.	: 9
	(3)	The NSW Planning Director-General is not required to do so if advice is not provided within the time specified in the regulations or if the advice conflicts with the advice of any other approval body under the State assessment requirements referred to in section 6.12.	9 13

Schedule 7 Infrastructure and other contributions—ancillary provisions

4

5

6

7

8

9

10

11

12

13

14

15

21

22

23

24

25

26

29

30

31

32

33

34

35

39

1

7.1 Regulations—transparency and accountability

- (1) The regulations may make provision for or with respect to requiring the collection and publication by public authorities of information concerning the provision of infrastructure and the determination, collection, application and use of infrastructure or other contributions under Part 7 of this Act.
- (2) The information required to be collected and published can include (but is not limited to):
 - (a) details of the amounts of monetary contributions paid and the purposes for which they were paid, and
 - (b) details of the purposes for which monetary contributions have been applied by a public authority, and
 - (c) details of the time frame for the provision of infrastructure for which any local infrastructure plan or growth infrastructure plan provides, and
 - (d) details of any borrowings or other arrangements made by a public authority for the provision of infrastructure, and 17
 - (e) the amount and other details of any monetary contributions that have not been applied for the purpose for which they were paid and that continue to be held by a public authority.
 18
 19
 20
- (3) The regulations can, for example, require the publication of information by a public authority by requiring inclusion of the information in any annual or other report of the public authority.
- (4) The regulations may also provide for the auditing of the administration of local infrastructure contributions by councils.

7.2 Regulations—planning agreements and infrastructure plans

- (1) The regulations may make provision for or with respect to planning agreements, 27 including the following: 28
 - (a) the form of planning agreements,
 - (b) the subject-matter of planning agreements,
 - (c) the making, amendment and revocation of planning agreements, including the giving of public notice and inspection by the public,
 - (d) requiring the provision to a public authority of a copy of a planning agreement and any amendment or notice of revocation of a planning agreement,
 - (e) the public inspection of planning agreements after they have been made.
- The regulations may make provision for or with respect to the preparation and approval of local infrastructure plans and growth infrastructure plans, including the format, structure and subject-matter of plans.
 36
 37
 38

7.3 Regulations—statutory charge etc to secure payment of contributions

- The regulations may make provision for securing the payment of infrastructure and other contributions under Part 7 by the creation of a statutory charge on the land in respect of which the contributions are imposed (including the priority of any such charge and the displacement of section 42 of the *Real Property Act 1900*).
- (2) The regulations may make provision for the issue of certificates as to whether or not any such contributions have been imposed or remain unpaid in respect of any land
 44

(and for that purpose confer functions on a public authority in relation to the issue of any such certificates).

1

2

3

4

5

6

7

8

9

10

11

12

13

14

19

25

26

27

28

34

35

36

37

38

45

(3) The regulations may apply (with or without modification) the provisions of the *Local Government Act 1993* or other Act relating to the sale of property for the recovery of unpaid rates or charges to the sale of land subject to any such statutory charge for the recovery of payments secured by the charge.

7.4 Assessment of Ioan commitments of Planning Ministerial Corporation

(1) The Planning Ministerial Corporation may, in respect of each year ending on 31 December, assess the amount required in any such year for the payment of interest on, or repayment of principal of, any loan raised by the Corporation on the councils whose areas or parts of areas are included in the development area to which the purpose for which the loan was raised relates.

(2) The regulations may make provision for or with respect to:

- (a) the assessment referred to in subclause (1), and
- (b) the notification of a council referred to in subclause (1) by the Corporation of 15 a decision to make the assessment, and 16
- (c) the provision by such a council of information necessary to determine the amount to be paid by the council in relation to the assessment, and 18
- (d) the payment by such a council of the whole or any part of an amount assessed.
- (3) A council required to pay the whole or any part of an amount assessed under 20 subclause (1) must make the payment from its consolidated fund.
 21
- (4) The Corporation may recover as a debt or liquidated demand in any court of competent jurisdiction any amount assessed upon a council and not paid on or before such day as may be prescribed in relation to the assessment.

7.5 Planning Growth Funds

- (1) There is established in the Special Deposits Account, a Planning Growth Fund in respect of each development area under clause 7.7 (a *Growth Fund*).
- (2) The Growth Fund in respect of each development area is to consist of:
 - (a) all money borrowed for the purpose of the acquisition or development of land
 within the development area and for the purpose of repaying or renewing a
 loan obtained for that purpose and the proceeds of any levy or assessment
 made by the Planning Ministerial Corporation for the purpose of repaying
 money so borrowed or renewing such a loan, and
 - (b) the proceeds of the sale or lease by the Planning Ministerial Corporation of any land situated within the development area, and
 - (c) all money and land directed by or under the planning legislation to be allocated to the Growth Fund, and
 - (d) all money received as a result of the investment of the Growth Fund, and
 - (e) such other money as the Treasurer authorises to be paid into the Growth Fund. 39
- (3) All land vested in the Planning Ministerial Corporation and situated within a development area is to form part of the assets of the Growth Fund in respect of that development area.
 40
 41
 42
- (4) The Growth Fund in respect of each development area may be applied to any of the
 43 following purposes:
 44
 - (a) the acquisition or development of any land within the development area,

	(b)	the payment of rates and charges due and payable by the Planning Ministerial Corporation in respect of land within the development area,	1 2
	(c)	transfers to any reserve for loan repayment in respect of money borrowed in respect of the development area,	3 4
	(d)	payment of principal, interest and expenses in respect of money borrowed in respect of the development area,	5 6
	(e)	any purpose authorised by or under the planning legislation for the application of the Growth Fund,	7 8
	(f)	the creation of assets and incurring and discharging liabilities not inconsistent with the purposes of the Growth Fund,	9 10
	(g)	payment of principal, interest and expenses in respect of money borrowed which is not chargeable to any fund other than the Growth Fund, or in respect of a loan or asset transferred from another fund,	11 12 13
	(h)	the investment of money for the creation of reserves for any purposes not inconsistent with the purposes of the Growth Fund,	14 15
	(i)	any costs incurred in the administration of the Growth Fund.	16
(5)	deve	Growth Fund may also be applied, with the approval of the Minister, to the lopment of land (whether vested in the Planning Ministerial Corporation or not) in the development area for the purpose of an improvement program, if:	17 18 19
	(a)	the Minister has considered likely future applications of the Growth Fund for all the purposes in subclause (4), and	20 21
	(b)	in the opinion of the Minister, implementation of the improvement program will improve public amenity by:(i) enhancing open space or the public domain, or	22 23
		(i) enhancing open space or the public domain, or(ii) providing suitable infrastructure or facilities at a regional or local level.	24 25
(6)	that a	Growth Fund in respect of each development area may be applied to purposes are necessary, incidental, subordinate or supplementary to any of the purposes ified in this clause.	26 27 28
Plan	ning T	rust Fund	29
(1)		e is established in the Special Deposits Account, the Planning Trust Fund (the <i>t</i> Fund).	30 31
(2)	The '	Trust Fund is to consist of the following assets:	32
	(a)	all money and land held by the Planning Ministerial Corporation by way of deposit or in trust for any person,	33 34
	(b)	all money and land assigned, conveyed, bequeathed or devised to the Planning Ministerial Corporation in trust for the purpose of any function which the Planning Ministerial Corporation is under the planning legislation empowered to exercise,	35 36 37 38
	(c)	all money received as a result of the investment of the Trust Fund.	39
(3)		Trust Fund is to be applied as follows:	40
	(a)	where the money or land is held by way of a deposit or in trust for any person, the money may be paid or the land may be assured to or on behalf of the person entitled thereto, but if the money has remained in the Trust Fund for 10 years, the Planning Ministerial Corporation may transfer it to such Planning Growth Fund as it may deem proper, subject to repaying it from that fund to any person entitled thereto,	41 42 43 44 45 46

	(b)	except as otherwise provided in this clause, for the purposes of and according to the trusts upon which the money or land is held by the Planning Ministerial Corporation,	1 2 3
	(c)	by investment in securities authorised under the <i>Trustee Act 1925</i> or for the purposes of and according to the trusts referred to in paragraph (b).	4 5
Deve	elopme	ent areas	6
(1)	Deve	elopment areas may be constituted in accordance with this clause.	7
(2)		NSW Planning Director-General may, by notice published on the NSW planning al, notify a proposal to constitute as a development area any area specified in the be.	8 9 10
(3)	Direc	in 14 days after the publication of the notice, the NSW Planning ctor-General is to notify the councils of the areas proposed to be included in the lopment area of the proposal and otherwise publicise the proposal.	11 12 13
(4)	Direc	person may, by notice in writing, lodge with the NSW Planning ctor-General, within 3 months after the publication of the notice, representations lation to the proposal.	14 15 16
(5)		re representations have been so lodged, the NSW Planning Director-General is fer the matter to the Minister who may either:	17 18
	(a)	confirm the proposal, or	19
	(b)	alter the proposal by excluding, from the proposed development area, any area other than an area or part in which the Planning Ministerial Corporation has acquired land.	20 21 22
(6)	Asse	e Minister has requested that a public hearing be held by the Planning assessment Commission with respect to the proposal, the Minister must not rmine the application until after:	23 24 25
	(a)	the public hearing has been held, and	26
	(b)	the Minister has considered the findings and recommendations of the Commission following the public hearing.	27 28
(7)	confi	o representations are lodged under this clause, the proposal is taken to be irmed immediately on the expiry of the period allowed for the lodgment of esentations.	29 30 31
(8)	NSW	areas specified in the proposal as confirmed or altered are, on publication on the <i>V</i> planning portal of a notice constituting them as a development area, to be tituted as a development area under the name specified in the notice.	32 33 34
(9)	porta exclu	NSW Planning Director-General may, by notice published on the NSW planning al, notify a proposal to alter a development area by including any land or by adding any land or to abolish such a development area, and the provisions of this apply to the notice as they apply to a notice referred to in subclause (2).	35 36 37 38
(10)	The time.	same land cannot be included in more than one development area at any one	39 40
(11)		ions 40 and 41 of the <i>Interpretation Act 1987</i> apply to a notice constituting, ing or abolishing a development area in the same way they apply to a statutory	41 42 43

8.2

Schedule 8 Building and subdivision—ancillary provisions

Regulations: Part 8 2				
	build	regulations may make provision for or with respect to the carrying out of ling work or subdivision work and, in particular, for or with respect to the wing:	3 4 5	
	(a)	requirements to comply with provisions of the <i>Building Code of Australia</i> or other specified standards in relation to building work or subdivision work,	6 7	
	(b)	applications for and the issue of certificates under Part 8,	8	
	(c)	the form and contents of certificates under Part 8,	9	
	(d)	conditions of certificates under Part 8,	10	
	(e)	modification of certificates under Part 8,	11	
	(f)	inspection of building work and subdivision work,	12	
	(g)	the functions of certifiers under Part 8,	13	
	(h)	the replacement of building certifiers and subdivision certifiers,	14	
	(i)	exemptions in relation to the requirement to obtain a certificate under Part 8,	15	
	(j)	the keeping of records in relation to building work or subdivision work,	16	
	(k)	notices and information to be given in relation to the carrying out of building work and subdivision work,	17 18	
	(1)	the procedure for dealing with complaints about building work or subdivision work.	19 20	
	State	Regulations authorised under Part 5 enable regulations to apply provisions of Part 8 to infrastructure development and development for the purposes of public priority structure.	21 22 23	
Smo	ke ala	rms in buildings providing sleeping accommodation	24	
(1)	The	regulations may make provision for or with respect to:	25	
	(a)	the installation of one or more smoke alarms in buildings in which persons sleep, and	26 27	
	(b)	the maintenance of smoke alarms installed in such buildings, and	28	
	(c)	prohibiting persons from removing or interfering with the operation of smoke alarms installed in such buildings.	29 30	
(2)		regulations made under this clause may (without limitation) do any one or more e following:	31 32	
	(a)	specify the kinds of buildings in which smoke alarms are to be installed,	33	
	(b)	specify the kinds of smoke alarms to be installed,	34	
	(c)	specify where a smoke alarm is to be located,	35	
	(d)	specify the maintenance that may be required in relation to a smoke alarm that has been installed,	36 37	
	(e)	specify circumstances in which development consent is not required in relation to the installation of a smoke alarm,	38 39	

(3)

(f) specify circumstances in which the consent of an owners corporation (within the meaning of the <i>Strata Schemes Management Act 1996</i>) is not required in relation to the installation of a smoke alarm.	1 2 3
In this clause:	4
<i>building</i> includes a manufactured home, a moveable dwelling or associated structure and includes a building erected before the commencement of this clause.	5 6

Schedule 9 Reviews and appeals—ancillary provisions

9.1 Regulations

The regulations may make provision for or with respect to reviews and appeals under Part 9, and in particular the procedure with respect to any such review or appeal.

1

2

Schedule 10 Enforcement—ancillary provisions

Part 1 Ministerial enforcement matters

10.1 Appointment of regional planning panel or administrator to exercise council's functions

- (1) The Minister may appoint a regional planning panel to exercise functions of a council if:
 - (a) the Minister is of the opinion that the council has failed to comply with its obligations under the planning legislation, or

1

2

3

4

5

6

7

8

9

10

11

12

13

14

20

21

22

23

24

25

- (b) the Minister is of the opinion that the performance of a council in dealing with planning and development matters (or any particular class of such matters) is unsatisfactory because of the manner in which the council has dealt with those matters, the time taken or in any other respect (having regard to criteria published by the Minister for the purposes of this clause), or
- (c) the council agrees to the appointment, or
- (d) a report referred to in section 74C of the *Independent Commission Against* (d) a report referred to in section 74C of the *Independent Commission Against* (d) a report referred to in section 74C of the *Independent Commission Against* (d) a report referred to in section 74C of the *Independent Commission Against* (e) 15
 (f) 16
 (f) 17
 (f) 16
 (f) 17
 (f) 16
 (f) 16
 (f) 17
 (f) 16
 <li(f) 16
 <li(f) 16
 (f) 16
 (f)
- (2) A regional planning panel may be appointed to exercise all or any particular function or class of functions of the council:
 - (a) as a consent authority, or
 - (b) in relation to the making or amendment of provisions of local plans or other plans under this Act, or
 - (c) as a certifier, or
 - (d) of any other kind prescribed by the regulations.
- (3) A regional planning panel may not exercise the functions of a council for a continuous period of more than 5 years. The Minister is to review the appointment of the panel after the period of 2 years after the appointment was made.
 27
 28
 29
- (4) Before appointing a regional planning panel to exercise the functions of a council, the Minister must notify the council concerned in writing of the proposed action (including the reasons for the proposed action) and request the council to show cause why the action should not be taken.
 (30) 31 32 33
 (31) 32 33
 (32) 32 33
 (33) 33 34
 (34) 34 34
 (35) 35 34
 (36) 36 34
 (37) 36 34
 (38) 36 34
 (39) 36 34
 (31) 36 34
 (32) 36 34
 (33) 36 34
 (34) 36 34
 (35) 36 34
 (36) 36 34
 (37) 36 34
 (38) 36 34
 (38) 36 34
 (39) 36 34
 (31) 36 34
 (32) 36 34
 (33) 36 34
 (34) 36 34
 (35) 36 34
 (36) 36 34
 (37) 36 34
 (38) 36 34
 (38) 36 34
 (38) 36 34
 (39) 36 34
 (31) 36 34
 (32) 36 34
 (33) 36 34
 (34) 36 34
 (35) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 (36) 36 34
 <l
- (5) The Minister must consider any written submissions made by the council within 21 days of notice being given under subclause (4) and must not take action under this 35 clause earlier than 21 days after the notice is given.
 34
 35
 36
- (6) Before appointing a regional planning panel to exercise the functions of a council,
 the Minister is to obtain the concurrence of the Minister for Local Government.
 38
- (7) The Minister may take action under this clause in the circumstances specified in subclause (1) (d) without conducting an inquiry or complying with subclauses (4)–(6) but, in that case, the Minister is to inquire into the matter as soon as practicable with a view to confirming or revoking the appointment.
 (7) The Minister may take action under this clause in the circumstances specified in subclause (1) (d) without conducting an inquiry or complying with 40 subclauses (4)–(6) but, in that case, the Minister is to inquire into the matter as soon 41 as practicable with a view to confirming or revoking the appointment.
- (8) The Minister must, as soon as reasonably practicable after appointing a regional planning panel to exercise the functions of a council, make the reasons for that appointment publicly available.
 43
 44
 45

	regional planning panel under this clause. In that case, a reference in clause 10.2 to a panel is taken to be a reference to that administrators.		Minister may appoint an administrator or administrators instead of appointing a onal planning panel under this clause. In that case, a reference in this clause and se 10.2 to a panel is taken to be a reference to that administrator or those inistrators.	1 2 3 4
	(10)	admi	regulations may make provision with respect to the exercise of functions by an inistrator or administrators, their costs and the provision of assistance and ities by the relevant council.	5 6 7
	(11)	In th	is clause:	8
		failu	re to comply with obligations under the planning legislation includes:	9
		(a)	a failure to carry into effect or enforce the provisions of the planning legislation (including a local plan) or a direction given by the Minister under a power conferred by the planning legislation in relation to the exercise of functions under the planning legislation, or	10 11 12 13
		(b)	without limiting paragraph (a), a failure to comply with a gateway determination for a planning proposal under Part 3 of this Act, or	14 15
		(c)	without limiting paragraph (a), a failure to provide access to and the use of staff and facilities to the Planning Assessment Commission or a regional planning panel as required by the planning legislation.	16 17 18
		<i>Indep</i> indic	<i>bus corrupt conduct</i> means corrupt conduct (within the meaning of the <i>pendent Commission Against Corruption Act 1988</i>) that may constitute a serious stable offence, being conduct in connection with the exercise or purported cise of the functions of a councillor.	19 20 21 22
10.2	Func	tions	of regional planning panels	23
	(1)	Duri	ng the period of appointment, the regional planning panel:	24
		(a)	is to exercise the functions of the council under the planning legislation that are specified in the order of appointment, and	25 26
		(b)	is, in the exercise of those functions, taken to be the council, and	27
		(c)	is to exercise those functions to the exclusion of the council except to the extent that the order of appointment provides otherwise, and	28 29
		(d)	is, in the exercise of those functions, to give priority to particular functions to the extent that the order of appointment so provides.	30 31
	(2)		clause has effect even if the appointment of the regional planning panel is equently found not to have been validly made.	32 33
	(3)	appli	clause applies to the Planning Assessment Commission in the same way as it les to a regional planning panel if, under the planning legislation, the mission exercises the relevant functions of the panel.	34 35 36
10.3	Prote	ection	for exercise of certain functions by Minister	37
	(1)	the N	clause applies to any function (a <i>protected function</i>) conferred or imposed on Anister (including a delegate of the Minister) relating to the appointment of a ponal planning panel to exercise the functions of a council.	38 39 40
	(2)	The e	exercise by the Minister of any protected function cannot be:	41
		(a)	challenged, reviewed, quashed or called into question before any court of law or administrative review body in any proceedings, or	42 43
		(b)	restrained, removed or otherwise affected by any proceedings.	44
	(3)		out limiting subclause (2), that subclause applies whether or not the proceedings e to any question involving compliance or non-compliance, by the Minister	45 46

(including a delegate of the Minister), with the provisions of this Part or the rules of natural justice (procedural fairness).

		uding a delegate of the Minister), with the provisions of this Part or the rules of ral justice (procedural fairness).	1 2
(4)	to co (inclu	ordingly, no court of law or administrative review body has jurisdiction or power nsider any question involving compliance or non-compliance, by the Minister uding a delegate of the Minister), with those provisions or with those rules so far ey apply to the exercise of any protected function.	3 4 5 6
(5)		clause has effect despite any provision of this Act or other legislation or any law (whether written or unwritten).	7 8
(6)		is clause:	9
	exerc	cise of functions includes:	10
	(a)	the purported exercise of functions, and	11
	(b)	the non-exercise or improper exercise of functions, and	12
	(c)	the proposed, apprehended or threatened exercise of functions.	13
	proce	eedings includes:	14
	(a)	proceedings for an order under Division 10.3 of Part 10 of this Act, and	15
	(b)	proceedings for an order in the nature of prohibition, certiorari or mandamus or for a declaration or injunction or for any other relief, and	16 17
	(c)	without limiting paragraph (b), proceedings in the exercise of the inherent jurisdiction of the Supreme Court or the jurisdiction conferred by section 23 of the <i>Supreme Court Act 1970</i> .	18 19 20
Susp	ensio	n of council's functions as a certifier	21
(1)	inves coun	e Building Professionals Board has made its final report of the results of an stigation under section 45 of the <i>Building Professionals Act 2005</i> in relation to a cil publicly available and is of the opinion that the council has not taken opriate action about a matter investigated, the Board may:	22 23 24 25
	(a)	make recommendations to the head of the Public Service agency to which the administration of the <i>Local Government Act 1993</i> is assigned as to the measures that it considers appropriate to be taken in relation to the matter, or	26 27 28
	(b)	recommend to the Minister that the Minister take action against the council under this clause.	29 30
		Note. Section 45 of the <i>Building Professionals Act 2005</i> enables the Building Professionals Board to investigate the work and activities of a council in its capacity as a certifier.	31 32 33
(2)	follo [.] 1993	Minister may, on the recommendation of the Board under this clause and wing consultation with the Minister administering the <i>Local Government Act</i> , make an order suspending a council's authority to exercise all or specified ions of a certifier.	34 35 36 37
(3)	A co	uncil must comply with an order under this clause that relates to the council.	38
(4)	Despite any other provision of the planning legislation, a council that is the subject of an order must not exercise any function of a certifier while the council's authority to exercise that function is suspended by operation of the order.		39 40 41
(5)	An order does not operate to suspend a council's authority to exercise the functions of a certifier in relation to any matter being dealt with by the council as a certifier before the commencement of the order, unless the order provides otherwise.		42 43 44
(6)		rder may contain provisions of a savings or transitional nature consequent on the ension contained in the order.	45 46

(7)	Without limiting subclause (6), an order may contain provisions for or with respect
	to the following:

the way in which any pending matter being dealt with by the relevant council (a) as a certifier is to be completed, including, for example, enabling the council to complete any such matter or providing for the matter to be completed by another certifier.

1 2

3

4

5

6

7

8

- directing any fee paid to the council to act as a certifier in relation to any (b) pending matter to be refunded,
- (c) directing the council to pay any fees required to be paid to another certifier to complete any pending matter being dealt with by the council as a certifier. 10
- (8) The Minister must revoke an order if satisfied that the relevant council has 11 implemented measures to address the matters that led to the making of the order. 12
- (9) Nothing prevents the Minister from amending an order made under this clause by 13 another order, including amending the first order to change the functions of a certifier 14 to which the first order relates. 15
- (10)An order under this clause must be in writing and published in the Gazette and takes 16 effect on the day on which it is published in the Gazette or on a later day specified in 17 the order. 18

Part 2 Development control orders

Division 1 General orders

	Column 1	Column 2	Colu	imn 3
	To do what?	When?	To w	hom?
	Stop Use Order To stop using premises	Premises are being used:for a prohibited purpose, or	•	The owner of premises or
	or a building Not to conduct or to stop conducting an	 for a purpose for which a planning approval is required but has not been obtained, or 	•	building The person using the premises or building
	activity on the premises	• in contravention of a planning approval.		building
	-	Building is being used:		
		• inconsistently with its classification under this Act or the <i>Local Government</i> <i>Act 1993</i> , and		
		• in a manner that constitutes or is likely to constitute a life threatening hazard or a threat to public health or public safety, and		
		• in a manner that is not regulated or controlled under any other Act by a public authority.		
		Premises are being used for an activity (that would, or would be likely to require planning approval) that:		
		• constitutes or is likely to constitute a life threatening hazard or a threat to public health or public safety, and		
		• is not regulated or controlled under any other Act by a public authority.		
	Stop Work Order To stop building work	Building work or subdivision work is carried out:	•	Owner of the land Any person apparently
	or subdivision work	• in contravention of this Act, or	-	
	carried out in contravention of this Act	• in a manner that would affect the support of adjoining premises.		engaged in the work
	Demolish Works	A building:	Owne	er of building
	Order To demolish or remove	• requiring a planning approval is erected without approval, or		
	a building	• requiring approval under the <i>Local</i> <i>Government Act 1993</i> is erected without approval, or		
		• is or is likely to become a danger to the public, or		
		• is so dilapidated that it is prejudicial to persons or property in the neighbourhood.		

	Column 1	Column 2	Column 3
	To do what?	When?	To whom?
4	Stop Demolition Order To stop demolishing, or not to demolish a building	Demolition requiring a planning approval is being carried out, or would be carried out, without approval or in contravention of an approval.	 The owner of premises The person carrying out the demolition or likely to carry out the demolition
5	Repair Order To repair or make structural alterations to a building	The building is or is likely to become a danger to the public or is so dilapidated that it is prejudicial to the occupants, persons, or property in the neighbourhood.	Owner of building
6	Remove Advertising Order To modify, demolish or remove an advertisement and any associated structure	 The advertisement is: unsightly, objectionable or injurious to the amenity of any natural landscape, foreshore, public reserve or public place at or near where the advertisement is displayed, or displayed contrary to a provision made by or under this Act, or associated with a structure erected contrary to a provision made by or under this Act. 	 The owner of premises displaying the advertisement or on which the associated structure is erected The person responsible for the display of the advertisement and erection of the associated structure
7	Public Safety Order To erect or install structures or appliances necessary for public safety	 A building: is about to be erected, or is dangerous to persons or property on or in a public place, or is about to be demolished. Works are: about to be carried out, or about to be demolished. A person who has failed to comply with a Stop 	The owner or occupier of the land
	Order To stop using premises or to evacuate premises	Use Order issued because the use constitutes or is likely to constitute a life threatening hazard or a threat to public health or public safety.	Stop ¹ Use Order was given
9	Exclusion Order To leave premises or not to enter premises	A person who has failed to comply with a Stop Use Order issued because the use constitutes or is likely to constitute a life threatening hazard or a threat to public health or public safety.	Any person

	Column 1	Column 2	Column 3
	To do what?	When?	To whom?
10	Restore Works Order To restore premises to the condition in which they were before unlawful building or other works occurred	An unauthorised building has been the subject of a Demolish Works Order or unauthorised works have been carried out.	 The owner of the premises Any person entitled to act on a planning approval, or acting in contravention of a planning approval
11	Compliance Order To comply with a planning approval for the carrying out of works	A planning approval has not been complied with.	 The owner of the premises Any person entitled to act on a planning approval, or acting in contravention of a planning approval
	To do whatever is necessary so that any building or part of a building that has been unlawfully erected complies with relevant development standards	Building has been unlawfully erected and does not comply with relevant development standards.	The owner of the premises
	To carry out works associated with subdivision	Authorised subdivision works, or works agreed to by the applicant have not been carried out.	The person required to carry out the works
12	Repair or Remove Works Order To repair or remove a building in a public place	The building is unlawfully situated wholly or partly in a public place.	Owner or occupier of the building
13	Complete Works Order To complete authorised works under a planning approval within a specified time	The authorised works have commenced, but have not been completed before the planning approval has lapsed.	The owner of the relevant land

	Column 1	Column 2	Column 3
	To do what?	When?	To whom?
14	Public Priority Infrastructure Order	The measures or other requirements are not being complied with.	The proponent or other person carrying out the
	To take or refrain from taking specified actions to ensure compliance with measures and other requirements in a project definition report relating to the carrying out of development for the purposes of public priority infrastructure		development

Division 2 Fire safety orders

	Column 1	Column 2	Column 3
	To do what?	In what circumstances?	To whom?
1	To do or stop doing things for the purposes of ensuring or promoting adequate fire safety or fire safety awareness	 When provision for fire safety or fire safety awareness is inadequate to: prevent fire, or suppress fire, or prevent the spread of fire. 	The owner of the premises or, in the case of a place of shared accommodation, the owner or manager
		To ensure or promote the safety of persons in the event of fire.	
		When lack of maintenance of the premises or the use of the premises constitutes a significant fire hazard.	
2	To stop doing an activity on premises, including on premises used for the purposes of shared accommodation	 The activity is or is likely to be: a life threatening hazard, or a threat to public health or public safety, and the activity is not regulated or controlled under any other Act by a public authority. 	Any person apparently engaged in promoting, conducting or carrying out the activity
3	To stop the use of premises or to evacuate premises, or not to enter the premises	When an order under item 1 or 2 above has already been served and has not been complied with.	Any person

Division 3 Brothel closure orders

	Column 1	Column 2	Column 3
	To do what?	In what circumstances?	To whom?
1	To stop using premises as a brothel, including to specifically stop using the premises for: • sexual acts or services	When premises are being used for a purpose that is prohibited. When premises are being used for a purpose for which a planning approval is required but has not been obtained.	The owner of the premises, or the person using premises for the purpose specified in the order.
	in exchange for payment, or	When premises are being used in contravention of a planning approval.	The person entitled to act on a planning approval
	massage services (other than genuine remedial or therapeutic massage services) in exchange for payment, or		who is acting in contravention of the approval. Any person apparently in control of, or managing,
	• adult entertainment involving nudity, indecent acts or sexual activity in exchange for payment or ancillary to other goods or services.		or assisting in the control or management of, the brothel.
	To prohibit using the premises for any of the above uses if those uses are prohibited under the local plan or require planning approval and no approval has been granted.		
	To comply with the conditions of a planning approval for the use of premises as a brothel.		

1

2

3

4

5

6

7

8

18

Division 4 Provisions relating to development control orders

10.5 Order may specify standards and work that will satisfy those standards

- (1) A relevant enforcement authority may give a development control order that does the following instead of specifying in the order the things the person to whom the order is given must do or refrain from doing:
 - (a) specifies the standard that the premises concerned are required to meet,
 - (b) indicates the nature of the work that, if carried out, would satisfy that standard.
- (2) The relevant enforcement authority may, in any such development control order, require the owner or occupier to prepare and submit to the relevant enforcement authority, within the period specified in the order, particulars of the work the owner or occupier considers necessary to make provision for such matters as may be so specified.
 (2) The relevant enforcement authority may, in any such development control order, 9
 (3) The relevant enforcement authority may, in any such development control order, 9
 (4) The relevant enforcement authority, within the period specified in the order, particulars of the work the owner or occupier considers necessary to make provision for such matters as may be so specified.
- (3) The relevant enforcement authority must, within 28 days after those particulars of 14 work are submitted to the authority: 15
 - (a) accept the particulars without modification or with such modifications as the authority thinks fit, or 17
 - (b) reject the particulars.

	(4)) If the relevant enforcement authority accepts the particulars of work without modification, the authority must as soon as possible order the owner to carry out that work.							
	(5)	If the relevant enforcement authority accepts the particulars of work with modifications or rejects the particulars, or if an owner fails to submit particulars of work as required under this clause, the authority must:							
		(a) prepare, within 3 months after the acceptance, rejection or failure, particulars of the work that the authority considers necessary to make provision for the matters specified in the order given to the owner, and	7 8 9						
		(b) order the owner to carry out that work.	10						
	(6)	enforcement authority who gave the order to accept or reject any particulars of work	11 12 13						
	(7)	A relevant enforcement authority may recover from an owner as a debt the authority's expenses of preparing particulars of work under this clause.							
	(8)	An order under this clause forms part of the development control order to which it relates.							
10.6	Orders that make or are likely to make residents homeless								
	(1)	If a development control order will or is likely to have the effect of making a resident homeless, the relevant enforcement authority proposing to give the order must consider whether the resident is able to arrange satisfactory alternative accommodation in the locality.							
	(2)	If the resident is not able to arrange satisfactory alternative accommodation in the locality, the relevant enforcement authority must provide the resident with:							
			25 26						
		(b) any other assistance that the person considers appropriate.	27						
10.7	Orders affecting heritage items 28								
	(1)	This clause applies to an item of the environmental heritage:							
		under the Heritage Act 1977 applies or to which an order under section 136 of	30 31 32						
			33 34						
	(2)	A relevant enforcement authority must not give a development control order respect of an item of the environmental heritage until after the authority h considered the impact of the order on the heritage significance of the item.							
	(3)	A relevant enforcement authority must not give a development control order in respect of an item of the environmental heritage to which subclause (1) (a) applies until after the authority has given notice of the proposed order to the Heritage Council and has considered any submissions duly made by the Heritage Council.							
	(4)	authority from the requirements of subclause (3), either unconditionally or subject to conditions. Any such exemption may be varied or revoked by the Heritage Council	42 43 44 45						

	(5)	The l	Heritage Council may make a submission about a proposed order:	1					
		(a)	within 28 days after it is given notice by the relevant enforcement authority, or	2					
		(b)	if, within 28 days after it is given notice by the relevant enforcement authority, the Heritage Council requests that a joint inspection of the item be made, within 28 days after the joint inspection is made.	3 4 5					
	(6)	This clause does not apply to:							
		(a)	a general order not to demolish or cease demolishing a building if given in an emergency, or	7 8					
		(b)	a general order of a kind prescribed by the regulations, or	9					
		(c)	a brothel closure order.	10					
10.8	8 Giving and taking effect of orders								
	 A development control order is given by serving a copy of the order on the person t whom it is addressed and takes effect from the time of service or a later time specifie in the order. 								
	(2)	The o	copy of the development control order is to be accompanied by a notice stating:	15					
		(a)	that the person to whom the order is addressed may appeal to the Land and Environment Court against the order, and	16 17					
		(b)	the period within which an appeal may be made.	18					
10.9	Reasons for orders to be given								
	(1)	A relevant enforcement authority that gives a development control order must give the person to whom the order is addressed the reasons for the order.							
	(2)	The reasons may be given in the development control order or in a separ instrument.							
	(3)		reasons must be given when the development control order is given, except in an gency. In an emergency, the reasons may be given the next working day.	24 25					
Divi	ision	5	Process for giving orders	26					
10.10	Natu	ral jus	stice requirements	27					
	(1)		re giving a development control order, a relevant authority must comply with ses 10.6, 10.12 and 10.13 and Division 7.	28 29					
	(2)	Subc	lause (1) does not apply to the following development control orders:	30					
		(a)	a general order (under item 2, Division 1),	31					
		(b)	a fire safety order (under item 2, Division 2),	32					
		(c)	an order given, and expressed to be given, in an emergency,	33					
		(d)	an order given by the Minister or the NSW Planning Director-General in connection with public priority infrastructure or State infrastructure development.	34 35 36					
			Division 8 has special provisions relating to fire safety orders and Division 9 has special sions relating to brothel closure orders.	37 38					
10.11	1 Effect of compliance								
			evant authority that complies with clauses 10.6, 10.12 and 10.13 and Division 7 ken to have observed the rules of procedural fairness.	40 41					

Division 6			Notices to be given	1					
10.12	Notic	e to b	e given of proposed order to person who will be subject to order	2					
	(1)	Before giving a development control order, a relevant enforcement authority must give notice to the person to whom the proposed order is directed of the following:							
		(a)	the intention to give the order,	5					
		(b)	the terms of the proposed order,	6					
		(c)	the period proposed to be specified as the period within which the order is to be complied with,	7 8					
		(d)	that the person to whom the order is proposed to be given may make representations to the relevant enforcement authority as to why the order should not be given or as to the terms of or period for compliance with the order.	9 10 11 12					
	(2)	autho in the comr	notice may provide that the representations are to be made to the relevant ority or a nominated person on a nominated date, being a date that is reasonable e circumstances of the case. In the case of a council this may be to a specified nittee of the council on a specified meeting date or to a specified employee of ouncil on or before a specified date.	13 14 15 16 17					
10.13	Notice to be given to other persons and bodies of proposed order								
	(1)	Notic	ce to other consent authorities	19					
		for w	council proposes to give a development control order in relation to development which another person is the consent authority, the council must give the other on notice of its intention to give the order.	20 21 22					
	(2)	Notic	ce to certifier	23					
		or su	ouncil proposes to give a development control order in relation to building work bdivision work for which the council is not the certifier, the council must give ertifier notice of its intention to give the order.	24 25 26					
10.14	Notice of fire safety orders to be given to Commissioner of Fire and Rescue NSW								
	A relevant authority must immediately give notice to the Commissioner of Fire and Rescue NSW after giving a fire safety order.								
10.15	Notice of giving of complete works order								
			levant authority must, on or as soon as practicable after the day on which the ant authority gives a complete works order, send a copy of the order to:	31 32					
		(a)	such persons (if any) as are, in the opinion of the relevant authority, likely to be disadvantaged by the giving of the order, and	33 34					
		(b)	such persons (if any) as are referred to in the regulations for the purposes of this clause.	35 36					
10.16	Details of c		orders and notices to be given to councils	37					
	(1)	1) A relevant authority (other than a council) who gives a notice or an order under the Part must immediately give a copy of the notice or order to the council.							
	(2)	coun	relevant authority, if requested by the council, must immediately inform the cil whether or not the notice is outstanding or the order is in force and of any n proposed to be taken by the relevant authority in relation to the notice or order.	40 41 42					

Division 7		7	Representations concerning proposed orders			
10.17	Maki	ing of	representations	2		
	(1)	deve	erson who is given notice under clause 10.12 of the intention to give a clopment control order may make representations concerning the proposed order ecordance with the notice.	3 4 5		
	(2)		the purpose of making the representations, the person may be represented by an tralian legal practitioner or agent.	6 7		
10.18	Hear	ring ar	nd consideration of representations	8		
		nom	relevant authority who intends to give the development control order or the inated person is required to hear and to consider any representations made under Division.	9 10 11		
10.19	Proc	edure	e after hearing and consideration of representations	12		
	(1)	deve	r hearing and considering any representations made concerning the proposed elopment control order, the relevant authority or the nominated person may rmine:	13 14 15		
		(a)	to give an order in accordance with the proposed order, or	16		
		(b)	to give an order in accordance with modifications made to the proposed order, or	17 18		
		(c)	not to give an order.	19		
	(2)	mod	the determination is to give a development control order in accordance with ifications made to the proposed order, the relevant authority is not required to notice under this Part of the proposed order as so modified.	20 21 22		
Divi	ision	8	Special provisions relating to fire safety orders	23		
10.20	Pow	ers of	fire brigades	24		
	(1)		authorised fire officer who inspects a building in accordance with section 8.20 e brigades inspection powers) of the <i>Planning Administration Act 2013</i> may give:	25 26		
		(a)	a fire safety order (under item 1) if the order does not require the carrying out of any structural work to the premises concerned, or	27 28		
		(b)	a fire safety order (under item 2) if the premises concerned are a place of shared accommodation, or	29 30		
		(c)	a fire safety order (under item 3) if a person to whom an order under paragraph (a) or (b) is given has failed to comply with the order.	31 32		
	(2)	deve	uses 10.6, 10.10, 10.12, 10.13 and 10.35 and Division 7 do not apply to a elopment control order given in accordance with this clause in circumstances on the authorised fire officer believes constitute an emergency or a serious risk to ty.	33 34 35 36		
	(3)	may	he purpose of giving such a development control order, an authorised fire officer exercise such of the powers of a relevant authority under this Part as are ified in the fire officer's authorisation under this clause.	37 38 39		
	(4)		cercising a power under this Part, an authorised fire officer may be accompanied assisted by a police officer.	40 41		
	(5)		authorised fire officer must forward a copy of a development control order given accordance with this clause to the relevant council.	42 43		

10.21 Inspection reports by fire brigades 1 If the Commissioner of Fire and Rescue NSW carries out an inspection of a building 2 (1)under section 8.20 (Fire brigades inspection powers) of the *Planning Administration* 3 Act 2013, the Commissioner must furnish to the council of the area in which the 4 building is located: 5 a report of the inspection, and (a) 6 (b) if of the opinion that adequate provision for fire safety has not been made 7 concerning the building, such recommendations as to the carrying out of work 8 or the provision of fire safety and fire-fighting equipment as the Commissioner 9 considers appropriate. 10 (2)A council must: 11 (a) table any report and recommendations it receives under this clause at the next 12 meeting of the council, and 13 at any meeting of the council held within 28 days after receiving the report and (b) 14 recommendations or at the next meeting of the council held after the tabling of 15 the report and recommendations, whichever is the later, determine whether it 16 will exercise its powers to give a fire safety order. 17 A reference in subclause (2) to a meeting of a council does not include a reference to (3)18 a special meeting of the council unless the special meeting is called for the purpose 19 of tabling any report and recommendations or making any determination referred to 20 in that subclause. 21 (4)A council must give notice of a determination under this clause to the Commissioner 22 of Fire and Rescue NSW. 23 **Division 9** Special provisions relating to brothel closure orders 24 10.22 Interpretation 25 (1)In this Division: 26 brothel closure order means a brothel closure order under Division 3. 27 (2)This Division has effect despite any other provision of this Part. 28 Note. Failure to comply with a brothel closure order is an offence (see section 10.7). 29 10.23 Procedure relating to making of brothel closure orders 30 Natural justice requirements not applicable (1)31 A person who gives a brothel closure order is not required to comply with 32 clauses 10.6, 10.12 and 10.13 and Division 7. 33 (2)Additional prohibitions may be included 34 A brothel closure order may also prohibit the use of the premises for specified related 35 sex uses, if the use of the premises for the specified uses is a prohibited development 36 or a development for which planning approval is required but has not been obtained. 37 (3) Additional persons to whom order may be given 38 In addition to any other person to whom a brothel closure order may be given, a 39 brothel closure order may be given to any person apparently in control of or 40 managing, or assisting in the control or management of, the brothel. 41

10.24	Com	plianc	ce with brothel closure orders	1		
	(1)	Peric	od for compliance	2		
			othel closure order must specify a period of not less than 5 working days within the order must be complied with.	3 4		
	(2)	Defences				
			a sufficient defence to a prosecution for an offence that arises from a failure to ply with a brothel closure order if the defendant satisfies the court that:	6 7		
		(a)	in a case where the defendant is the owner of the premises, the defendant has taken all reasonable steps to evict the persons operating the brothel or using the premises for the specified related sex uses, or	8 9 10		
		(b)	in all cases, the defendant has taken all reasonable steps to prevent the use of the premises as a brothel or for the specified related sex uses.	11 12		
10.25	Арр	eals		13		
		Regu	lations may be made for or with respect to the following matters:	14		
		(a)	the conferral of jurisdiction on the Local Court with respect to appeals against brothel closure orders,	15 16		
		(b)	removing the right to appeal under Part 9 if an appeal is made to the Local Court against a brothel closure order under the regulations,	17 18		
		(c)	the conferral of jurisdiction on the Land and Environment Court with respect to appeals from decisions of the Local Court on appeals against brothel closure orders,	19 20 21		
		(d)	the modification of provisions of the <i>Crimes (Appeal and Review)</i> Act 2001 for the purposes of appeals referred to in paragraph (c).	22 23		
Divi	sion	10	Modification and revocation of orders	24		
10.26	Mod	ificatio	on of orders	25		
	(1)		levant authority who gives a development control order may, at any time, modify order (including a modification of the period specified for compliance with the r).	26 27 28		
	(2)	Plan	ept in the case of a development control order given by the Minister or the NSW ning Director-General, a modification may only be made if the person to whom order is given agrees to that modification.	29 30 31		
10.27	Reve	ocatio	n of orders	32		
	(1)	at an	evelopment control order given by the Minister may be revoked by the Minister by time, and an order given by the NSW Planning Director-General may be ked by the Minister or the NSW Planning Director-General at any time.	33 34 35		
	(2)		evelopment control order given by a consent authority may be revoked by the ent authority at any time.	36 37		
	(3)	A de any t	evelopment control order given by a council may be revoked by the council at time.	38 39		
	(4)		evelopment control order given by an authorised fire officer may be revoked by athorised fire officer at any time.	40 41		
10.28	Mini	ster m	ay revoke or modify a council's order	42		
	(1)	The	Minister may revoke or modify a development control order given by a council.	43		

	(2)		ce of the revocation or modification must be given to the council and the person nom the development control order was given.	1 2
	(3)		revocation or modification takes effect from the date specified in the Minister's e. The date may be the date on which the order was given by the council or a date.	3 4 5
	(4)	that i	Minister may prohibit a council from re-making a development control order s revoked or modified under this clause, totally or within such period or except cordance with such terms and conditions (if any) as the Minister may specify.	6 7 8
	(5)		ce of a prohibition may be given in the same notice as notice of the revocation odification of a development control order or in a separate notice.	9 10
10.29	Limit	ation	on Minister's orders	11
		has t	Minister must not take any action under clause 10.28 that is inconsistent with, or he effect of revoking or modifying, a development control order given by the cil unless the Minister is of the opinion that:	12 13 14
		(a)	it is necessary because of an emergency, or	15
		(b)	it is necessary because of the existence or reasonable likelihood of a serious risk to health or safety, or	16 17
		(c)	the order relates to a matter of State or regional significance, or	18
		(d)	the order relates to a matter in which the intervention of the Minister is necessary in the public interest.	19 20
Divi	sion	11	Effect of orders and compliance with orders	21
10.30	Effec	t of o	rder on successors in title	22
10.30	Effec	A de unde	rder on successors in title velopment control order given to a person binds any person claiming through or r or in trust for or in succession to the person or who is a subsequent owner or pier to the person, as if the order had been given to that person.	22 23 24 25
10.30 10.31		A de unde occuj	velopment control order given to a person binds any person claiming through or r or in trust for or in succession to the person or who is a subsequent owner or	23 24
		A der unde occup od for A de	velopment control order given to a person binds any person claiming through or r or in trust for or in succession to the person or who is a subsequent owner or pier to the person, as if the order had been given to that person.	23 24 25
	Perio	A de unde occup od for A de terms How terms	velopment control order given to a person binds any person claiming through or r or in trust for or in succession to the person or who is a subsequent owner or pier to the person, as if the order had been given to that person. compliance with order evelopment control order must specify a reasonable period within which the	23 24 25 26 27
	Peric (1) (2)	A der unde occuj od for A de terms How terms serio	velopment control order given to a person binds any person claiming through or r or in trust for or in succession to the person or who is a subsequent owner or pier to the person, as if the order had been given to that person. compliance with order evelopment control order must specify a reasonable period within which the s of the order are to be complied with. ever, a development control order may require immediate compliance with its s in circumstances which the person who gives the order believes constitute a	23 24 25 26 27 28 29 30
10.31	Peric (1) (2)	A de unde occup od for A de terms serio inuing A de the o	 velopment control order given to a person binds any person claiming through or r or in trust for or in succession to the person or who is a subsequent owner or pier to the person, as if the order had been given to that person. compliance with order evelopment control order must specify a reasonable period within which the s of the order are to be complied with. ever, a development control order may require immediate compliance with its s in circumstances which the person who gives the order believes constitute a us risk to health or safety or an emergency. 	23 24 25 26 27 28 29 30 31
10.31	Peric (1) (2) Cont	A de unde occup od for A de terms serio inuing A de the o with This	 welopment control order given to a person binds any person claiming through or r or in trust for or in succession to the person or who is a subsequent owner or pier to the person, as if the order had been given to that person. compliance with order evelopment control order must specify a reasonable period within which the s of the order are to be complied with. ever, a development control order may require immediate compliance with its s in circumstances which the person who gives the order believes constitute a us risk to health or safety or an emergency. g effect of orders velopment control order that specifies a time by which, or period within which, rder must be complied with continues to have effect until the order is complied 	23 24 25 26 27 28 29 30 31 31 32 33 34
10.31	Peric (1) (2) Cont (1) (2)	A de unde occup od for A de terms serio inuing A de the o with This contr	 welopment control order given to a person binds any person claiming through or r or in trust for or in succession to the person or who is a subsequent owner or pier to the person, as if the order had been given to that person. compliance with order evelopment control order must specify a reasonable period within which the s of the order are to be complied with. ever, a development control order may require immediate compliance with its s in circumstances which the person who gives the order believes constitute a us risk to health or safety or an emergency. geffect of orders velopment control order that specifies a time by which, or period within which, rder must be complied with continues to have effect until the order is complied even though the time has passed or the period has expired. clause does not apply to the extent that any requirement under a development 	23 24 25 26 27 28 29 30 31 31 32 33 34 35 36

10.34 Compliance with order under Division 4

A person complies with a requirement of an order under clause 10.5 (2) by submitting to the relevant authority who gives the order such matters as the person would be required to submit if applying to a consent authority for development consent to carry out the work.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

20

21

22

23

24

25

26

27

28

29

10.35 Compliance with orders by occupiers or managers

If an occupier or manager complies with a development control order, the occupier or manager may (unless the occupier or manager has otherwise agreed):

- (a) deduct the cost of so complying (together with interest at the rate currently prescribed by the Supreme Court rules in respect of unpaid judgment debts) from any rent payable to the owner, or
- (b) recover the cost (and that interest) from the owner as a debt in any court of competent jurisdiction.

10.36 Occupier of land may be required to permit owner to carry out work

- A relevant authority who gives a development control order may order the occupier of any land to permit the owner of the land to carry out specified work on the land, being work that is, in the relevant authority's opinion, necessary to enable the requirements of this Act or the regulations or of any development control order to be complied with.
- (2) An occupier of land on whom such an order is served must, within 2 days after the order is served, permit the owner to carry out the work specified in the order.
- (3) If an order under this clause is in force, the owner of the land concerned is not guilty of an offence arising from his or her failure to comply with the requirements of this Act or the regulations, or of any development control order, that is caused by the occupier of the land refusing to permit the owner to carry out the work specified in the order.
- (4) Subclause (3) applies only if the owner of the land satisfies the Court that the owner has, in good faith, tried to comply with the requirements concerned.

10.37 Failure to comply with order—carrying out of work by consent authority

- A relevant authority who gives a development control order may do all such things as are necessary or convenient to give effect to the terms of the order (including the carrying out of any work required by the order) if the person to whom the order was given fails to comply with the terms of the order.
 30 31 32 33
- (2) The relevant authority may exercise the relevant authority's functions under this clause irrespective of whether the person required to comply with the order has been prosecuted for an offence against this Act.
 34
 35
 36
- (3) In any proceedings before the Land and Environment Court that are brought by a relevant authority who gave a development control order to a person as a result of the person's failure to comply with the order, the Court may, at any stage of the proceedings, order the relevant authority to exercise the relevant authority's functions under this clause. Having made such an order, the Court may continue to hear and determine the proceedings or may dismiss the proceedings.
- (4) If the Minister or the NSW Planning Director-General gave the development control order, the Minister's or NSW Planning Director-General's functions under this clause may be exercised by the Planning Ministerial Corporation.
 43
 44
 45

10.38 Recovery of expenses by relevant authority for carrying out work

(1)	If a relevant authority takes action under clause 10.37 to give effect to a development control order by demolishing a building, the relevant authority may remove any materials concerned.

5

6

7

8

9

10

11

12

13

14

15

16

17

21

22

23

31

32

37

40

41

42

- (2) The relevant authority may sell those materials but only if the relevant authority's expenses in giving effect to the terms of the development control order are not paid to the relevant authority within 14 days after removal of the materials.
- (3) If the proceeds of such a sale exceed the expenses incurred by the relevant authority in relation to the demolition and the sale, the relevant authority:
 - (a) may deduct out of the proceeds of the sale an amount equal to those expenses, and
 - (b) must pay the surplus to the owner on demand.

(4) If the proceeds of sale do not exceed those expenses, the relevant authority:

- (a) may retain the proceeds, and
- (b) may recover the deficiency (if any) together with the relevant authority's costs of recovery from the owner as a debt.
- (5) Materials removed that are not saleable may be destroyed or otherwise disposed of.
- (6) A relevant authority who carries out work under clause 10.37 in relation to development for which an amount of security has been provided to the relevant authority:
 - (a) may be recompensed for the work from the security if the security is more than the costs of carrying out the work, and
 - (b) must pay any surplus remaining to the person entitled to it on demand.
- (7) Any expenses incurred under this clause by a relevant authority who gave a development control order, together with all associated costs, may be recovered by the relevant authority in any court of competent jurisdiction as a debt due to the relevant authority by the person required to comply with the order.
 (7) Any expenses incurred under this clause by a relevant authority who gave a development control order, together with all associated costs, may be recovered by the relevant authority in any court of competent jurisdiction as a debt due to the relevant authority by the person required to comply with the order.
- (8) The expenses are to be reduced by the amount of any proceeds of any sale under this clause or the amount of any security provided in respect of development to which the order relates.
 28
 29
 30
- (9) Nothing in this clause affects the owner's right to recover any amount from any lessee or other person liable for the expenses concerned.
- (10) The recovery of costs and expenses by a relevant authority under this clause does not include the costs and expenses of court proceedings, but nothing in this clause prevents the relevant authority from receiving costs as between party and party in respect of those proceedings.
 33
 34
 35
 36

10.39 Enforcement of orders by cessation of utilities

- (1) This clause applies in relation to a failure to comply with any of the following 38 development control orders: 39
 - (a) a brothel closure order,
 - (b) a stop use order in respect of such classes of residential, tourist or other development as are prescribed by the regulations.
- (2) In this clause, the *Court* means the Land and Environment Court and, in relation to
 43 a brothel closure order, includes the Local Court.
 44

	(3)	appli order	berson fails to comply with a development control order to which this clause es, the Court may, on the application of the person who gave the order, make an (a <i>utilities order</i>) directing that a provider of water, electricity or gas to the ises concerned cease to provide those services.	1 2 3 4		
	(4)	contr	lities order is not to be made in respect of a failure to comply with a development ol order that is a stop use order unless the Court is satisfied that the failure has ed or is likely to cause a significant adverse impact on health, safety or public hity.	5 6 7 8		
	(5)	A uti	lities order may apply to the whole or part of the premises.	9		
	(6)	A utilities order ceases to have effect on the date specified in the utilities order, or 3 months after the order is made, whichever occurs first.				
	(7)		pplication for a utilities order must not be made unless not less than 7 days notice e proposed application is given to the following persons:	12 13		
		(a)	any person to whom the development control order was given,	14		
		(b)	any provider of water, electricity or gas to the premises who is affected by the application,	15 16		
		(c)	any owner or occupier of the premises.	17		
	(8)	prem	owner or occupier of premises, or a provider of water, electricity or gas to ises, who is affected by an application for a utilities order is entitled to be heard epresented in proceedings for the order.	18 19 20		
	(9)		etermining whether to make a utilities order, the Court is to take into deration the following matters:	21 22		
		(a)	the effects of the failure to comply with the development control order,	23		
		(b)	the uses of the premises,	24		
		(c)	the impact of the order on the owner, occupier or other users of the premises,	25		
		(d)	whether health, safety or public amenity will be adversely affected by the order,	26 27		
		(e)	any other matter the Court thinks appropriate.	28		
	(10)		lities order must not be made for premises, or any part of premises, used for ential purposes unless the regulations authorise the making of a utilities order.	29 30		
	(11)	other	by or agreement or arrangement applying to the provision of water, electricity s to the premises, or part of the premises, concerned.	31 32 33		
	(12)		ompensation is payable to any person for any damage or other loss suffered by person because of the making or operation of a utilities order or this clause.	34 35		
	(13)	is in provi	by order of water, electricity or gas must not, during a period that a utilities order force in relation to premises, or part of premises, require payment for the asion of water, electricity or gas services to the premises or part of the premises or than services related to the implementation of the order).	36 37 38 39		
	(14)		Court may make a utilities order when it determines an appeal against a lopment control order, if subclauses (7) and (8) have been complied with.	40 41		
Div	ision	12	Compliance cost notices	42		
).40	Com	plianc	e cost notices	43		
	(1)		evant authority who gives a development control order to a person may also a compliance cost notice on the person.	44 45		

10.40

	(2)	serve	<i>impliance cost notice</i> is a notice in writing requiring the person on whom it is ed to pay all or any reasonable costs and expenses incurred by the relevant ority in connection with:	1 2 3
		(a)	monitoring action under the development control order, and	4
		(b)	ensuring that the development control order is complied with, and	5
		(c)	any costs or expenses relating to an investigation that leads to the giving of the development control order, and	6 7
		(d)	any costs or expenses relating to the preparation or serving of the notice of the intention to give the development control order, and	8 9
		(e)	any other matters associated with the development control order.	10
	(3)	reasc	ompliance cost notice is to specify the amount required to be paid and a onable period within which the amount is to be paid or, if the regulations cribe the period to be allowed for payment, that period.	11 12 13
	(4)		relevant authority may recover any unpaid amounts specified in a compliance notice as a debt in a court of competent jurisdiction.	14 15
	(5)	but w of the who	e person on whom a compliance cost notice is served complies with the notice was not the person who was responsible for the situation giving rise to the issue e notice, the cost of complying with the notice may be recovered by the person complied with the notice as a debt in a court of competent jurisdiction from the on who was responsible.	16 17 18 19 20
	(6)	The 1	regulations may make provision for or with respect to the following:	21
		(a)	the issue of compliance cost notices,	22
		(b)	the form of compliance cost notices,	23
		(c)	limiting the amounts that may be required to be paid under compliance cost notices or the matters in respect of which costs and expenses may be required to be paid under those notices.	24 25 26
Divi	sion	13	Miscellaneous	27
10.41	Com	bined	orders	28
			rson who gives a development control order may include two or more orders in ame instrument.	29 30
10.42	Orde	ers ma	y be given to two or more persons jointly	31
			propriate in the circumstances of the case, a development control order may at two or more people to do the thing specified in the order jointly.	32 33
10.43	Notio	ce in r	espect of land or building owned or occupied by more than one person	34
	(1)		nd, including land on which a building is erected, is owned or occupied by more one person:	35 36
		(a)	a development control order in respect of the land or building is not invalid merely because it was not given to all of those owners or occupiers, and	37 38
		(b)	any of those owners or occupiers may comply with such a development control order without affecting the liability of the other owners or occupiers to pay for or contribute towards the cost of complying with the order.	39 40 41
	(2)	other	ing in this Part affects the right of an owner or occupier to recover from any r person all or any of the expenses incurred by the owner or occupier in plying with such a development control order.	42 43 44

Part 3 Civil enforcement—ancillary evidentiary provisions

10.44 Evidence of use of premises as backpackers' hostel

(1) This clause applies to proceedings before the Land and Environment Court for an order under Division 10.3 of Part 10 to remedy or restrain a breach of this Act in relation to the use of premises as a backpackers' hostel.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

21

22

23

24

25

26

27

28

35

39

42

43

(2) In any proceedings to which this clause applies, the Land and Environment Court may rely on circumstantial evidence to find that particular premises are used as a backpackers' hostel.

Note. Examples of circumstantial evidence include (but are not limited to) the following:

- (a) evidence relating to persons entering and leaving the premises (including the depositing of luggage) that is consistent with the use of the premises for a backpackers' hostel,
- (b) evidence of the premises being advertised expressly or implicitly for the purposes of a backpackers' hostel (including advertisements on or in the premises, newspapers, directories or the Internet),
- (c) evidence relating to internal and external signs and notices at the premises (including price lists, notices to occupants and offers of services) that is consistent with the use of the premises for a backpackers' hostel,
- (d) evidence of the layout of rooms, and the number and arrangement of beds, at the premises that is consistent with the use of the premises for a backpackers' hostel. 20

10.45 Proceedings relating to use of premises as brothel

(1) Application

This clause applies to proceedings before the Land and Environment Court for an order under Division 10.3 of Part 10 to remedy or restrain a breach of this Act in relation to the use of premises as a brothel. Subclauses (5) and (6) extend to any such proceedings in relation to all brothels within the meaning of the *Restricted Premises Act 1943*.

(2) Adjournments to obtain consent only in exceptional circumstances

The Land and Environment Court may not adjourn the proceedings to which this clause applies to enable an application for a planning approval to be made unless it is of the opinion that the adjournment is justified because of the exceptional circumstances of the case. The fact that it is intended to make an application for approval, or that an application for approval has been made, is not by itself an exceptional circumstance. 34

(3) Time for making application for approval limited to 10 days

If the Land and Environment Court adjourns the proceedings to enable an application36for a planning approval to be made, the proceedings must be brought back before the37Court if the application is not made within 10 working days of the adjournment.38

(4) **Only one adjournment**

The Land and Environment Court may make only one adjournment of particular 40 proceedings to enable an application for a planning approval to be made. 41

(5) Finding may be made on circumstantial evidence

In any proceedings to which this clause applies, the Land and Environment Court:

- (a) may rely on circumstantial evidence to find that particular premises are used 44 as a brothel, and 45
- (b) may make such a finding without any direct evidence that the particular 46 premises are used as a brothel. 47

(6)

ence	wever, the presence in any premises of articles or equipment that facilitate or ourage safe sex practices does not of itself constitute evidence of any kind that the nises are used as a brothel.	1 2 3
Note	e. Examples of circumstantial evidence include (but are not limited to) the following:	4
(a)	evidence relating to persons entering and leaving the premises (including number, gender and frequency) that is consistent with the use of the premises for prostitution,	5 6
(b)	evidence of appointments with persons at the premises for the purposes of prostitution that are made through the use of telephone numbers or other contact details that are publicly advertised,	7 8 9
(c)	evidence of information in books and accounts that is consistent with the use of the premises for prostitution,	10 11
(d)	evidence of the arrangement of, or other matters relating to, the premises, or the furniture, equipment or articles in the premises, that is consistent with the use of the premises for prostitution.	12 13 14

Schedule 11		le 11	Miscellaneous—ancillary provisions			
Par	't 1	Mis	cellaneous regulation-making powers	2		
11.1	Reg	ulation	s—miscellaneous	3		
		Regu	lations may be made, in particular, for or with respect to the following:	4		
		(a)	exempting specified or classes of persons, premises or other matters from any specified provision of the planning legislation (either generally or subject to conditions),	5 6 7		
		(b)	any function conferred by the planning legislation on any person,	8		
		(c)	requiring information, particulars, returns and statistics to be furnished to the NSW Planning Director-General by councils and the time and mode of furnishing and the manner of verifying them,	9 10 11		
		(d)	the form, time, manner and mode of giving notices under the planning legislation,	12 13		
		(e)	the content, form, erection, maintenance and removal of signs relating to the carrying out of development or persons involved with the carrying out of development,	14 15 16		
		(f)	obligations on persons regarding fire and building safety,	17		
		(g)	temporary structures,	18		
		(h)	entertainment venues (including in connection with the existing use of premises),	19 20		
		(i)	the purposes, objectives, provision and maintenance of affordable housing,	21		
		(j)	the documents to be provided to, and the matters to be notified to, a consent authority, council or certifier under the planning legislation.	22 23		
Par	rt 2	Рар	er subdivisions	24		
11.2	Defi	nitions		25		
		In this	s Part:	26		
			opment plan—see clause 11.7.	27		
		devel	opment plan costs means the following:	28		
		(a)	the costs of obtaining or preparing any reports,	29		
		(b)	the amount of any contributions, fees or other charges applicable to the proposed subdivision or subdivision works,	30 31		
		(c)	administrative costs of the relevant authority relating to the development plan,	32		
		(d)	any other costs prescribed by the regulations for the purposes of this definition.	33		
		-	<i>ing purpose</i> —see clause 11.4 (1) (c).	34		
			<i>unt authority</i> for subdivision land means the authority designated by a vision order as the relevant authority for the land.	35 36		
			vision land means land subject to a subdivision order.	37		
			vision order means an order under clause 11.4.	38		
		subdi	vision works means works for the following purposes:	39		
		(a)	roads,	40		
		(b)	water supply, sewerage services and drainage,	41		
		(c)	telecommunications,	42		

		(d)	electricity supply,	1
		(e)	any other purpose prescribed by the regulations for the purposes of this definition.	2 3
11.3	Subd	livisio	n authorities	4
			of the following authorities may be designated in a subdivision order as the ant authority for the subdivision land:	5 6
		(a)	the Planning Ministerial Corporation,	7
		(b)	a council,	8
		(c)	Urban Growth NSW,	9
		(d)	a development corporation established under the Growth Centres (Development Corporations) Act 1974,	10 11
		(e)	any other body prescribed by the regulations.	12
11.4	Subd	livisio	n orders	13
	(1)	The M	Minister may, by Ministerial planning order:	14
		(a)	declare specified land to be subdivision land, and	15
		(b)	specify the relevant authority for the subdivision land, and	16
		(c)	specify the purpose for which the order is made (the <i>planning purpose</i>), and	17
		(d)	specify the functions (if any) under this Part conferred on the relevant authority, and	18 19
		(e)	specify the conditions (if any) to which the exercise of those functions are subject, and	20 21
		(f)	specify the subdivision works (if any) to be undertaken by the relevant authority in respect of the subdivision land.	22 23
	(2)	The M	Minister may make a subdivision order only if:	24
		(a)	the Minister is of the opinion that it is desirable to do so to promote and co-ordinate the orderly and economic use and development of the land affected by the order, and	25 26 27
		(b)	the land has been subdivided and is held by more than one owner and the Minister is satisfied that the land is land for which no provision or inadequate provision has been made for subdivision works, and	28 29 30
		(c)	that land is subject to a planning control provision of a local plan, or a planning proposal, that will facilitate the proposed planning purpose, and	31 32
		(d)	the Minister has consulted with the proposed relevant authority, any other Minister responsible for that authority and the council of the area in which that land is situated, and	33 34 35
		(e)	the Minister is satisfied that a development plan for that land has been prepared by the relevant authority in accordance with this Part, and	36 37
		(f)	the Minister has considered any provisions of the development plan that modify or disapply the provisions of Division 4 of Part 3 of the <i>Land</i> <i>Acquisition (Just Terms Compensation) Act 1991</i> , and	38 39 40
		(g)	at least 60% of the total number of owners of that land, and the owners of at least 60% of the total area of that land, have consented to the proposed development plan.	41 42 43
	(3)		he purposes of subclause (2) (b) and (g), two or more owners of the same lot are treated as one owner.	44 45

	(4)	The	Minister may repeal a subdivision order only if the Minister:	1
		(a)	has consulted with the relevant authority for the subdivision land and the council of the area in which the land is situated, and	2 3
		(b)	is satisfied that notice of the proposed repeal has been given to the owners of the land subject to the order in the manner prescribed by the regulations.	4 5
	(5)	Subc	clause (2) (g) does not apply to an order amending a subdivision order.	6
11.5	Fund	ctions	of relevant authority	7
	(1)	A re	levant authority has the functions conferred on it by a subdivision order.	8
	(2)	orde	levant authority may only exercise functions conferred on it under a subdivision r for the purposes of, or purposes ancillary to, the planning purpose specified in ubdivision order.	9 10 11
	(3)		ctions conferred on a relevant authority by a subdivision order are in addition to other functions conferred on the authority under any other law.	12 13
	(4)	autho	uses 11.8–11.14 set out the functions that may be conferred on a relevant ority under a subdivision order but do not otherwise confer those functions on a vant authority.	14 15 16
	(5)		levant authority may not exercise functions under clause 11.8 or 11.10 unless e is a development plan in force in relation to the subdivision land.	17 18
11.6	Oblig	gation	s of relevant authority	19
		deve purp	elevant authority must, in accordance with the subdivision order and any elopment plan applicable to the subdivision land, give effect to the planning ose specified in the order and must undertake or arrange for the undertaking of subdivision works specified in the order.	20 21 22 23
11.7	Deve	elopm	ent plans	24
	(1)		are a development plan for subdivision land or proposed subdivision land.	25 26
	(2)	A de	evelopment plan is to contain the following matters:	27
		(a)		~ ~ ~
			a proposed plan of subdivision for the land,	28
		(b)	details of subdivision works to be undertaken for the land,	28
		(b) (c)	details of subdivision works to be undertaken for the land, details of the costs of the subdivision works and of the proposed means of funding those works,	
		. ,	details of subdivision works to be undertaken for the land, details of the costs of the subdivision works and of the proposed means of funding those works, details of the development plan costs,	29 30
		(c)	details of subdivision works to be undertaken for the land, details of the costs of the subdivision works and of the proposed means of funding those works,	29 30 31
		(c) (d)	details of subdivision works to be undertaken for the land, details of the costs of the subdivision works and of the proposed means of funding those works, details of the development plan costs, details of the proportion of the costs referred to in paragraphs (c) and (d) to be borne by the owners of the land and of the manner in which the owners may meet those costs (including details of any proposed voluntary land trading scheme or voluntary contributions or, if voluntary measures are not agreed to	29 30 31 32 33 34 35 36
		(c) (d) (e)	details of subdivision works to be undertaken for the land, details of the costs of the subdivision works and of the proposed means of funding those works, details of the development plan costs, details of the proportion of the costs referred to in paragraphs (c) and (d) to be borne by the owners of the land and of the manner in which the owners may meet those costs (including details of any proposed voluntary land trading scheme or voluntary contributions or, if voluntary measures are not agreed to by the owners, of compulsory land acquisition or compulsory contributions), rules as to the form of compensation for land that is compulsorily acquired and	29 30 31 32 33 34 35 36 37 38
		(c) (d) (e) (f)	details of subdivision works to be undertaken for the land, details of the costs of the subdivision works and of the proposed means of funding those works, details of the development plan costs, details of the proportion of the costs referred to in paragraphs (c) and (d) to be borne by the owners of the land and of the manner in which the owners may meet those costs (including details of any proposed voluntary land trading scheme or voluntary contributions or, if voluntary measures are not agreed to by the owners, of compulsory land acquisition or compulsory contributions), rules as to the form of compensation for land that is compulsorily acquired and how entitlement to compensation is to be calculated, rules as to the distribution of any surplus funds after the completion of	29 30 31 32 33 34 35 36 37 38 39 40

(4)) Without limiting subclause (3), the regulations may require the consent of the owners							
	of subdivision land to be obtained to proposed amendments to the applicable development plan in the circumstances, and in the manner, specified by the regulations.							

(5) The validity of a development plan must not be questioned in any legal proceedings except those commenced in the Court by any person within 3 months of the date of its publication in the Gazette.

11.8 Land acquisition powers

8

5

6

7

9

10

11

- (1) A relevant authority may, for a planning purpose specified in a subdivision order, acquire subdivision land by agreement or by compulsory process in accordance with the *Land Acquisition (Just Terms Compensation) Act 1991*.
- (2) A relevant authority may not give a proposed acquisition notice under the Land
 Acquisition (Just Terms Compensation) Act 1991 without the approval of the
 Minister.
- (3) If compensation provided for that acquisition is in accordance with the rules set out in a development plan in force in relation to the land: 15
 - (a) sections 44 (2), 45 (3), 49–51, 64, 66 (4) and 68 (2) of the *Land Acquisition* (*Just Terms Compensation*) *Act 1991* do not apply in relation to compensation other than monetary compensation, and
 - (b) all or any provisions of Division 4 of Part 3 of that Act do not apply, or apply 20 with modifications, if the development plan so provides. 21
- (4) The rules set out in a development plan may provide that all or any of the provisions of Division 4 of Part 3 of the *Land Acquisition (Just Terms Compensation) Act 1991* 23 do not apply to the determination of compensation under that plan, or apply with such modifications as are set out in that plan.
- (5) If the rules set out in a development plan make provision as referred to in subclause (4), the Valuer-General must determine compensation to be offered to a person under the *Land Acquisition (Just Terms Compensation) Act 1991* in respect of land acquired under this clause in accordance with the rules set out in any applicable development plan adopted by a relevant authority for the land.
- (6) For the purposes of this clause, a reference in the *Land Acquisition (Just Terms* 31 *Compensation) Act 1991* to an amount of compensation includes a reference to 32 compensation other than monetary compensation and a reference to payment of 33 compensation includes a reference to the provision of such compensation.
- (7) Subclauses (3)–(6) have effect despite any provision of the *Land Acquisition (Just Terms Compensation) Act 1991*.

11.9 Other powers to acquire and dispose of land

37 38

39

40

41

42

43

44

35

36

A relevant authority may sell, lease, exchange, mortgage or otherwise deal with or dispose of subdivision land vested in the authority, or an interest in that land, and may grant easements, rights-of-way or covenants over that land.

11.10 Contribution powers

- (1) A relevant authority may, by notice in writing, require an owner of subdivision land to make a reasonable monetary contribution for the provision, extension or augmentation of subdivision works and the development costs.
- (2) A requirement under this clause must be in accordance with the development plan 45 applicable to the subdivision land. 46

- (3)The amount payable by the owner of subdivision land under this clause is to be 1 reduced by the amount or value of any voluntary contribution (whether a monetary 2 or other contribution) made by the owner for the provision, extension or 3 augmentation of subdivision works and the development costs in accordance with the 4 development plan applicable to the subdivision land or an agreement with the 5 relevant authority. 6
- (4) Compliance with a requirement for a contribution under this clause, or a voluntary contribution made in accordance with a development plan, operates to satisfy any 8 other requirement imposed by a public authority under this or any other Act (in 9 relation to or in connection with the subdivision land) for the dedication of land or 10 the payment of money in respect of the provision of the same subdivision works, to 11 the extent of the value of the land dedicated or the amount of money paid in 12 compliance with the requirement. 13

7

14

16

17

18

19

20

21

22

23

27

28

33

34

42

- The regulations may make provision for the determination of the value for the (5)purposes of this clause of the land dedicated or traded to the authority in accordance 15 with a development plan.
- A contribution required to be made under this clause may be in addition to any other (6)contribution required to be made under the planning legislation.

11.11 Use of monetary contributions and other amounts

- (1)The following are to be paid by the authority to a fund or funds approved by the Minister:
 - a monetary contribution paid to a relevant authority by the owner of (a) subdivision land for subdivision works or development costs,
 - any money paid by the relevant authority to meet contribution amounts under (b) 24 the development plan in respect of land acquired by the authority under this 25 Part, 26
 - (c) the proceeds of any disposal by the relevant authority of land acquired under this Part.
- The following may be paid from any fund to which contributions or amounts are paid (2)29 under this clause: 30
 - payments to persons or bodies with respect to the provision of subdivision (a) 31 works, 32
 - payments in connection with the exercise of functions by the relevant authority (b) for the planning purpose specified in the subdivision order,
 - payments for the whole or part of compensation payable under clause 11.8 and (c) 35 any payments required to be made under the Land Acquisition (Just Terms 36 Compensation) Act 1991, 37
 - payments for the distribution of any surplus funds after the completion of (d) 38 subdivision works and any other payments under this clause, 39
 - any money required to meet the administrative expenses of the relevant (e) 40 authority in relation to its functions under the subdivision order. 41

11.12 Powers to carry out subdivision works

- (1)The relevant authority may carry out, or arrange for the carrying out of, subdivision 43 works with respect to subdivision land. 44
- The relevant authority may enter into contracts and other arrangements for the (2)45 carrying out of subdivision works. 46

	(3)	deve	levant authority may make an application for development consent to carry out lopment on subdivision land for the purposes of subdivision works without the ent of the owner of the land.	1 2 3
	(4)		consent authority may grant consent to any such application even if the owner e land has failed to consent to the application.	4 5
	(5)		is clause, <i>subdivision works</i> includes the carrying out of any research or stigation related to the provision or augmentation of subdivision works.	6 7
11.13	Roa	ds pov	vers	8
	(1)	the n	ad within subdivision land cannot be provided, opened, dedicated, closed (within neaning of Part 4 of the <i>Roads Act 1993</i>) or realigned by the Crown, a public prity or any person except with the consent of the relevant authority.	9 10 11
	(2)	A pri	ivate road, or part of a private road, within subdivision land cannot be:	12
		(a)	provided, opened, closed or realigned, or	13
		(b)	regulated in its use, or	14
		(c)	used for a purpose other than a road,	15
		exce	pt with the consent of the relevant authority.	16
11.14	Anci	illary p	oowers	17
			levant authority has, for the purpose of any other functions conferred under this the following functions:	18 19
		(a)	the authority may enter into agreements with the owners of subdivision land for the purposes of a voluntary land trading scheme or the provision of voluntary contributions or for other purposes connected with the authority's functions under the subdivision order,	20 21 22 23
		(b)	the authority may cause surveys to be made, and plans of survey to be prepared, in relation to subdivision land or proposed subdivision land (whether or not vested in the authority),	24 25 26
		(c)	the authority may manage subdivision land vested in the authority in accordance with the development plan,	27 28
		(d)	the authority may carry out research or investigation relating to subdivision works or proposed subdivision works,	29 30
		(e)	the authority may (subject to this Act) subdivide and re-subdivide land, and consolidate subdivided or re-subdivided land vested in the authority,	31 32
		(f)	with the consent of the owner or occupier of the land, a person authorised in writing by the authority may enter subdivision land or proposed subdivision land.	33 34 35
11.15	Pow	er to iı	nvestigate land for subdivision order proposals	36
		An a	uthority specified in clause 11.3 may, before a subdivision order is made:	37
		(a)	cause surveys to be made, and plans of survey to be prepared, in relation to proposed subdivision land (whether or not vested in the authority), and	38 39
		(b)	carry out research or investigation relating to proposed subdivision works.	40

11.16 Other powers of entry

11.16	Othe	er powers of entry	1
	(1)	An authorised person may, without the consent of the owner or occupier of subdivision land or proposed subdivision land and in accordance with the regulations:	2 3 4
		(a) enter that land for a planning purpose, or	5
		(b) enter that land in connection with the carrying out of subdivision works or research or investigation relating to proposed subdivision works, or	6 7
		(c) enter that land in connection with the preparation of, or research or investigation for the purposes of, a development plan or proposed development plan.	8 9 10
	(2)	In this clause, authorised person means the following persons:	11
		(a) a person authorised in writing by a relevant authority,	12
		(b) a person authorised in writing by the Minister in connection with the exercise of the powers of an authority under clause 11.15.	13 14
11.17	Failu	ire to pay contributions	15
		clause 11.10 may be recovered by the relevant authority in any court of competent	16 17 18
11.18	Volu	ntary contributions agreements to run with land	19
	(1)	authority and a person who owns subdivision land under which the owner is required to pay a monetary contribution to be used for or applied for subdivision works or	20 21 22 23
	(2)		24 25
			26 27
			28 29
	(3)	approved by the Registrar-General, the Registrar-General is to register the voluntary	30 31 32
			33 34
		(b) by registering the agreement in the General Register of Deeds if the agreement relates to land not under the <i>Real Property Act 1900</i> .	35 36
	(4)	Registrar-General under this clause is binding on, and is enforceable against, the	37 38 39 40
	(5)		41 42
11.19	State	e taxes	43
	(1)		44 45

approval of the Treasurer, exempts the authority from payment of any or all State taxes.

1

2

3

4

5

6

7

8

9

10

11

(2) In this clause, *State tax* means duty under the *Duties Act 1997* or any other tax, duty, rate (including a local government rate), fee or other charge imposed by or under any Act or law of the State, other than payroll tax.

11.20 Regulations

Regulations may be made for or with respect to the following matters:

- (a) the manner in which consent to a development plan is to be given by owners of land,
- (b) information to be provided to the Minister by, and reports by, relevant authorities,
- (c) the effect of the repeal or amendment of a subdivision order, or of the 12 amendment of a development plan, 13
- (d) the obstruction of or interference with authorities or persons exercising 14 functions under this Part. 15

		Ie 12 Savings, transitional and other provisions onal savings and transitional provisions may be added to this Schedule by regulation.	1 2
Part 1		General provisions	3
12.1	Defir	nition	4
		In this Schedule: <i>former Act</i> means the <i>Environmental Planning and Assessment Act 1979</i> .	5 6
12.2	Refe	rences to former Act	7
	(1)	A reference in any Act or in any instrument made under an Act (other than in the planning legislation):	8 9
		(a) to the former Act is to be read as including a reference to this Act and the <i>Planning Administration Act 2013</i> , or	10 11
		(b) to a provision of the former Act is to be read as including a reference to a corresponding provision of this Act or the <i>Planning Administration Act 2013</i> .	12 13
	(2)	This clause has effect subject to this Schedule and to any contrary intention in the provisions in which the relevant reference occurs.	14 15
12.3	Gene	eral saving	16
	(1)	If anything done under the former Act before the repeal of the former Act by this Act and still having effect immediately before that repeal could have been done under this Act if this Act had been in force when the thing was done, the thing done continues to have effect as if it had been done under this Act.	17 18 19 20
	(2)	This clause has effect subject to this Schedule and to any contrary intention.	21
	(3)	A reference in this clause to this Act includes a reference to the <i>Planning</i> Administration Act 2013.	22 23
Par	t 2	Community participation provisions	24
12.4	Inter	im community participation plans of councils	25
		Until a council has a community participation plan that it is required to have under Part 2 of this Act or 30 June 2016 (whichever first occurs), the community participation plan that applies to the council is the model council community participation plan designated by Ministerial planning order.	26 27 28 29
Par	t 3	Strategic planning provisions	30
12.5	Exist	ing strategic documents taken to be strategic plans (other than local plans)	31
	(1)	A Ministerial planning order may adopt, as a strategic plan under this Act (or as a draft strategic plan under this Act), any document made or prepared in the administration of the former Act or in anticipation of the commencement of this Act.	32 33 34
	(2)	Any public exhibition or other step taken in the preparation of any such document is taken to be public exhibition or a step taken under this Act.	35 36
	(3)	This clause does not apply to a local plan.	37

12.6 **Existing SEPPs**

Existing SEPPs		
(1)	The provisions of State Environmental Planning Policies under the former Act are taken to be planning control provisions of each local plan, other than provisions that do not apply to the land to which the local plan applies or the following provisions: Note. On the enactment of this Act no provisions were listed.	2 3 4 5
(2)	The provisions of State Environmental Planning Policies under the former Act relating to exempt or complying development are taken to be development code provisions of each local plan, other than provisions that do not apply to the land to which the local plan applies or the following provisions: Note. On the enactment of this Act no provisions were listed.	6 7 8 9 10
(3)	Subclause (1) or (2) may be amended by Ministerial planning order to add or omit provisions of State Environmental Planning Policies.	11 12
(4)	A local plan does not include any provisions to which this clause applies if the planning control provisions or development code provisions (as the case requires) of the local plan declare that those provisions are not included in the local plan by the operation of this clause.	13 14 15 16
Exist	ing LEPs and deemed EPIs	17
(1)	The provisions of local environmental plans or deemed environmental planning instruments under the former Act (other than provisions declaring exempt or complying development) are taken to be planning control provisions of the relevant local plan.	18 19 20 21
(2)	The provisions of local environmental planning instruments or deemed environmental planning instruments under the former Act declaring exempt or complying development are taken to be development code provisions of the relevant local plan.	22 23 24 25
(3)	A local plan does not include any provisions to which this clause applies if the planning control provisions or development code provisions (as the case requires) of the local plan declare that those provisions are not included in the local plan by the operation of this clause.	26 27 28 29
(4)	This clause ceases to apply to deemed environmental planning instruments at the end of the period of 3 years after the commencement of this Act.	30 31
(5)	To avoid doubt, this clause extends to provisions of local environmental plans or deemed environmental planning instruments that relate to heritage items, heritage conservation areas and archaeological sites identified in those plans and instruments.	32 33 34
Existing DCPs		
(1)	The provisions of development control plans under the former Act are taken to be development guide provisions of the relevant local plan.	36 37
(2)	A local plan does not include any provisions to which this clause applies if the development guide provisions of the local plan declare that those provisions are not included in the local plan by the operation of this clause.	38 39 40
(3)	To avoid doubt, this clause extends to provisions of development control plans that relate to heritage items, heritage conservation areas and archaeological sites identified in those plans.	41 42 43
Existing draft plans and planning proposals		
(1)	The provisions of the former Act relating to the making of environmental planning instruments or development control plans continue to apply (during the period of	45 46
	 (1) (2) (3) (4) (2) (3) (4) (5) Exist (1) (2) (3) Exist (3) Exist 	 taken to be planning control provisions of each local plan, other than provisions that do not apply to the land to which the local plan applies or the following provisions: Note. On the enactment of this Act no provisions were listed. (2) The provisions of State Environmental Planning Policies under the former Act relating to exempt or complying development are taken to be development code provisions of ach local plan, other than provisions: Note. On the enactment of this Act no provisions that do not apply to the land to which the local plan applies or the following provisions: Note. On the enactment of this Act no provisions that do not apply to the land to which the local plan does not include any provisions to which this clause applies if the planning control provisions or development code provisions state case requires) of the local plan declare that those provisions are not included in the local plan by the operation of this clause. Existing LEPs and deemed EPIs (1) The provisions of local environmental planning control provisions of the relevant local plan. (2) The provisions of local environmental planning instruments or deemed environmental planning instruments under the former Act (declaring exempt or complying development) are taken to be planning control provisions of the relevant local plan. (2) The provisions of local environmental planning instruments or deemed environmental planning instruments under the former Act declaring exempt or complying development are taken to be development code provisions (as the case requires) of the local plan does not include any provisions to which this clause applies if the planning control provisions or development code provisions (as the case requires) of the local plan doetare that those provisions to which this clause applies if the planning development are taken to be development and in the local plan between the provisions are not included in the local plan by the operation of this clause.

		3 years after the commencement of this Act) to draft plans and planning proposals in existence on the commencement of this Act.	1 2
	(2)	The provisions of this Schedule relating to plans under the former Act extend to a plan that is made as a result of the operation of this clause.	3 4
Part 4		Development (other than infrastructure) assessment and consent provisions	5 6
12.10	Exis	ting development consents and applications for development consent	7
	(1)	A development consent granted under the former Act is taken to be a development consent granted under this Act.	8 9
	(2)	The former Act continues to apply to an application for development consent (or for the modification of a development consent) pending on the commencement of this Act. Subclause (1) extends to a development consent or modification granted on the determination of any such application.	10 11 12 13
Par	rt 5	Infrastructure and environmental impact assessment provisions	14 15
12.11	Exis	ting Part 5 EIS activity	16
	(1)	A determination made under Part 5 of the former Act is taken to be a determination under Division 5.1 of Part 5 of this Act.	17 18
	(2)	Part 5 of the former Act continues to apply to an activity, the environmental impact assessment for which was pending under that Part on the commencement of this Act. Subclause (1) extends to a determination made under Part 5 of the former Act on the completion of any such assessment.	19 20 21 22
12.12	Existing State significant infrastructure approvals and applications for those approvals		23 24
	(1)	An approval for State significant infrastructure granted under the former Act is taken to be a State infrastructure approval granted under Division 5.2 of Part 5 of this Act.	25 26
	(2)	The former Act continues to apply to an application for approval for State significant infrastructure (or for the modification of any such approval) pending on the commencement of this Act. Subclause (1) extends to an approval or modification granted on the determination of any such application.	27 28 29 30
Par	rt 6	Concurrences, consultation and other legislative approvals provisions	31 32
Note.	. On the	enactment of this Act this Part did not contain any provisions.	33
Par	rt 7	Infrastructure and other contribution provisions	34
12.13	Exis	ting local contribution plans and local infrastructure contributions	35
	(1)	A contribution plan of a council under Division 6 of Part 4 of the former Act is taken to be a local infrastructure plan of the council and the contribution provisions of the local plan in so far as they provide for the imposition of local infrastructure contributions.	36 37 38 39
	(\mathbf{n})	I and information contributions may continue to be imposed in the	

(2) Local infrastructure contributions may continue to be imposed in accordance with any such contribution plan (as if the provisions of any such contribution plan were 41

contribution provisions of the local plan of the council) until relevant contribution provisions have been included in the local plan of the council.

1

2

3

4

5

6

7

8

9

10

11

12

20

21

22

23

24

25

26

27

31

32

33

41

(3) Until the relevant contribution provisions are included in the local plan, contributions towards public amenities and public services of the kind authorised by the former Act and the contribution plan may be imposed even if they are not of a kind authorised to be imposed by this Act.

12.14 Existing Ministerial directions and determinations relating to special infrastructure contributions

- (1) Any directions or determinations of the Minister under Division 6 of Part 4 of the former Act relating to special infrastructure contributions are taken to be the provisions of growth infrastructure plans and the contribution provisions of local plans in so far as they require the imposition of regional infrastructure contributions.
- Regional infrastructure contributions are to continue to be imposed in accordance
 with any such directions and determinations until relevant contribution provisions
 have been included in the local plan of the council.
- (3) Until the relevant contribution provisions are included in the local plan, contributions towards infrastructure of the kind authorised by the former Act and the directions or determinations of the Minister are to be imposed even if they are not of a kind authorised to be imposed by this Act.

12.15 Existing draft contribution plans

- (1) The provisions of Division 6 of Part 4 of the former Act relating to the making of contribution plans continue to apply (during the period of 3 years after the commencement of this Act) to draft plans in existence on the commencement of this Act.
- (2) The provisions of this Schedule relating to contribution plans under the former Act extend to a plan that is made as a result of the operation of this clause.

12.16 Existing infrastructure documents taken to be infrastructure plans

- A Ministerial planning order may adopt, as an infrastructure plan under this Act (or as a draft infrastructure plan under this Act), any document made or prepared in the administration of the former Act or in anticipation of the commencement of this Act.
- (2) Any public exhibition or other step taken in the preparation of any such document is taken to be public exhibition or a step taken under this Act.

12.17 Existing planning agreements

- Any planning agreement entered into under Division 6 of Part 4 of the former Act is taken to be a planning agreement entered into in accordance with Part 7 of this Act.
- (2) The provisions of Division 6 of Part 4 of the former Act continue to apply to a draft planning agreement that was in the course of preparation and had been publicly notified before the commencement of this Act. The provisions of this Schedule relating to planning agreements under the former Act extend to a planning agreement that is entered into as a result of the operation of this subclause.
 36
 37
 38
 39
 40

12.18 Existing contribution conditions (and affordable housing provisions) not affected

 This Act does not affect any condition of a planning approval imposed under Division 6 or 6A of Part 4 of the former Act relating to local infrastructure contributions, special infrastructure contributions or contributions for affordable housing.

	(2)	Any affordable housing scheme provided under any environmental planning instrument immediately before the commencement of this Act continues in force and contributions for affordable housing are to continue to be imposed and dealt with in accordance with Division 6A of Part 4 of the former Act (including any direction under section 94G of the former Act).	1 2 3 4 5
12.19	Plan	ning Growth Funds and Trust Fund	6
	(1)	A Development Fund in respect of a development area established under section 129 of the former Act is taken to be a Planning Growth Fund in respect of that development area in the Special Deposits Account established under Schedule 7.	7 8 9
	(2)	The Trust Fund established under section 129 of the former Act is taken to be the Planning Trust Fund in the Special Deposits Account established under Schedule 7.	10 11
	(3)	Until provision is made under Schedule 7, a development area constituted under section 132 of the former Act is taken to be constituted as a development area under Schedule 7.	12 13 14
Par	t 8	Building and subdivision provisions	15
12.20	Exist	ing building and other Part 4A certificates	16
	(1)	A certificate issued under Part 4A of the former Act is taken to be a certificate issued under Part 8 of this Act.	17 18
	(2)	Part 4A of the former Act continues to apply to an application for a certificate under that Part pending on the commencement of this Act. Subclause (1) extends to a certificate issued on the determination of any such application.	19 20 21
	(3)	The regulations may provide for other circumstances in which Part 4A of the former Act continues to apply in relation to complying development or development that is the subject of a construction certificate.	22 23 24
Par	rt 9	Reviews and appeals provisions	25
12.21	Pend	ling reviews	26
		A review of any decision under the former Act by a consent authority or other person or body that is pending on the commencement of this Act is to be determined in accordance with the law in force immediately before the commencement of this Act. The provisions of this Schedule relating to things done under the former Act extend to any decision made on the review.	27 28 29 30 31
12.22	2 Pending appeals		32
		An appeal to the Land and Environment Court that is pending on the commencement of this Act is to be determined in accordance with the law in force immediately before the commencement of this Act. The provisions of this Schedule relating to things done under the former Act extend to any decision made on the appeal.	33 34 35 36
Par	t 10	Enforcement provisions	37
		enactment of this Act this Part did not contain any provisions.	38
	t 11 On the	Miscellaneous provisions enactment of this Act this Part did not contain any provisions.	39 40

Schedule 13 Repeal of Acts and instruments

The following Acts and instruments are repealed:

(a)	Environmental Planning and Assessment Act 1979 No 203	3
(b)	Miscellaneous Acts (Planning) Repeal and Amendment Act 1979 No 205	4
(c)	Local Government and Environmental Planning and Assessment Amendment (Transfer of Functions) Act 2001 No 93	5 6
(d)	Environmental Planning and Assessment Amendment Act 2008 No 36	7
(e)	Statute Law (Miscellaneous Provisions) Act (No 2) 2008 No 114	8
(f)	Statute Law (Miscellaneous Provisions) Act 2009 No 56	9
(g)	Environmental Planning and Assessment Amendment (Part 3A Repeal) Act 2011 No 22	10
(h)	Environmental Planning and Assessment Regulation 2000	11
(i)	Miscellaneous Acts (Planning) Savings and Transitional Provisions Regulation 1980	12

1

2